INVITROGEN CORP Form DEFR14A April 21, 2008

#### **SCHEDULE 14A INFORMATION**

#### PROXY STATEMENT PURSUANT TO SECTION 14(a) OF THE

#### **SECURITIES EXCHANGE ACT OF 1934**

#### **Amendment No. 1**

Filed by the Registrant b Filed by a Party other than the Registrant "

Check the appropriate box:

- " Preliminary Proxy Statement
- x Definitive Proxy Statement
- " Confidential, for Use of the Commission Only (as permitted by Rule 14a-6(e)(2))
- " Definitive Additional Materials
- " Soliciting Material Pursuant to sec. 240.14a-11(c) or sec. 240.14a-12

#### **INVITROGEN CORPORATION**

(Name of Registrant as Specified In Its Charter)

(Name of Person(s) Filing Proxy Statement, if other than the Registrant)

- þ Fee not required.
- " Fee computed on table below per Exchange Act Rules 14a-6(i)(4) and 0-11.
  - (1) Title of each class of securities to which transaction applies:
  - (2) Aggregate number of securities to which transaction applies:
  - (3) Per unit price or other underlying value of transaction computed pursuant to Exchange Act Rule 0-11 (set forth the amount on which the filing fee is calculated and state how it was determined):

(4) Proposed maximum aggregate value of transaction:

(5) Total fee paid:

" Fee paid previously with preliminary materials.

- " Check box if any part of the fee is offset as provided by Exchange Act Rule 0-11(a)(2) and identify the filing for which the offsetting fee was paid previously. Identify the previous filing by registration statement number, or the Form or Schedule and the date of its filing.
  - (1) Amount Previously Paid:

(2) Form, Schedule or Registration Statement No.:

(3) Filing Party:

(4) Date Filed:

Our Compensation and Organizational Development Committee has approved certain amendments to our 2004 Equity Incentive Plan (the 2004 Plan) which are summarized in Proposal 4 of this Amendment No. 1 to our proxy. These amendments will: (1) reduce the new number of shares we are requesting stockholders to approve from 3 million to 1.5 million; and (2) implement a no liberal share reserve feature which will limit the number of shares subject to awards which can again become available for new issuances under the 2004 Plan.

March 14, 2008

Dear Stockholder:

This year s Annual Meeting of Stockholders will be held on April 30, 2008 at 8:00 a.m. local time, at the offices of the Company, 5781 Van Allen Way, Carlsbad, California 92008. You are cordially invited to attend.

The Notice of Annual Meeting of Stockholders and the Proxy Statement, which describe the formal business to be conducted at the meeting, follow this letter.

After reading the Proxy Statement, please make sure to vote your shares by promptly voting electronically or telephonically as described on page 2 of the enclosed Proxy Statement, dating, signing and returning your proxy card, or attending the annual meeting in person. Regardless of the number of shares you own, your careful consideration of, and vote on, the matters before our stockholders are important.

A copy of our 2007 Annual Report is also enclosed.

I look forward to seeing you at the annual meeting.

Very truly yours,

Gregory T. Lucier Chairman and Chief Executive Officer

#### NOTICE OF ANNUAL MEETING OF STOCKHOLDERS

#### TO BE HELD APRIL 30, 2008

To our Stockholders:

The Annual Meeting of Stockholders of Invitrogen Corporation (the Company), will be held on April 30, 2008, at 8:00 a.m. local time, at the offices of the Company, 5781 Van Allen Way, Carlsbad, California 92008, for the following purposes:

- 1. To elect three Class III directors, each to hold office for a three-year term and until his or her respective successor is elected and qualified. The Board of Directors has nominated the following persons for election as Class III directors at the meeting: Balakrishnan S. Iyer, Ronald A. Matricaria, and W. Ann Reynolds, Ph.D.
- 2. To consider a proposal to ratify the appointment of Ernst & Young LLP as the independent auditors for the Company for the Company s fiscal year ending December 31, 2008.
- 3. To consider a proposal to amend the Company s 1998 Employee Stock Purchase Plan.
- 4. To consider a proposal to amend the Company s 2004 Equity Incentive Plan.

5. To transact such other business as may properly come before the meeting or any adjournment or postponement thereof. Stockholders of record at the close of business on February 29, 2008, are entitled to notice of, and to vote at, the Annual Meeting and any adjournments or postponements thereof. For ten days prior to the Annual Meeting, a complete list of the stockholders of record on February 29, 2008, will be available at our principal offices for examination during ordinary business hours by any stockholder for any purpose relating to the meeting.

By Order of the Board of Directors,

John A. Cottingham Senior Vice President, General Counsel & Secretary

Carlsbad, California

March 14, 2008

IMPORTANT: Please vote telephonically or electronically or promptly fill in, date, sign and return the enclosed proxy card in the accompanying pre-paid envelope to ensure that your shares are represented at the meeting. You may revoke your proxy before it is voted. If you attend the meeting, you may choose to vote in person even if you have previously sent in your proxy card.

#### **Invitrogen Corporation**

#### 5791 Van Allen Way

#### Carlsbad, California 92008

#### PROXY STATEMENT FOR ANNUAL MEETING OF STOCKHOLDERS

The accompanying proxy is being solicited by the Board of Directors of Invitrogen Corporation (Invitrogen or the Company) and contains information related to the Annual Meeting of Stockholders (the Annual Meeting) to be held April 30, 2008, at 8:00 a.m. local time, or any adjournment or postponement thereof, for the purposes described in the accompanying Notice of Annual Meeting. The Annual Meeting will be held at the offices of the Company, 5781 Van Allen Way, Carlsbad, California 92008. This Proxy Statement was filed with the Securities and Exchange Commission (the SEC) on March 5, 2008, and the approximate date on which the Proxy Statement and the accompanying form of proxy were first sent or given to stockholders was March 14, 2008.

Invitrogen will bear the cost of soliciting proxies. We may solicit stockholder proxies by mail through our regular employees, and may request banks and brokers, and other custodians, nominees and fiduciaries, to solicit their customers who have Invitrogen stock registered in their names and will reimburse them for their reasonable, out-of-pocket costs. We may use the services of our officers, directors, and others to solicit proxies, personally or by telephone, without additional compensation. In addition, Invitrogen plans to use the services of Morrow & Company and the Altman Group to solicit stockholder proxies at an aggregate cost of approximately \$15,000, plus reimbursement of reasonable out-of-pocket expenses.

#### **ABOUT THE MEETING**

#### What is the purpose of the Annual Meeting?

At the Annual Meeting, stockholders will act upon the matters outlined in the Notice of Annual Meeting that is attached to this Proxy Statement. These matters include the election of directors, the ratification of the reappointment of Ernst & Young LLP as our independent auditors, amendments to the Invitrogen 1998 Employee Stock Purchase Plan (the 1998 ESPP), including an increase to the number of shares available for issuance under the 1998 ESPP, and amendments to the Invitrogen 2004 Equity Incentive Plan (the 2004 EIP), including an increase to the number of shares of common stock (Common Stock) authorized under the 2004 EIP. In addition, management will report on Invitrogen s performance during 2007 and will respond to questions from our stockholders. The Annual Report for the fiscal year ended December 31, 2007, is available online at http://www.invitrogen.com.

#### Who is entitled to vote at the meeting?

Only stockholders of record as of the close of business on the record date, February 29, 2008, will be entitled to vote the shares of Invitrogen stock they held on the record date at the Annual Meeting. As of the close of business on the record date, there were 46,218,671 shares of Common Stock outstanding and entitled to vote.

Stockholders may vote in person or by proxy. Each holder of shares of Invitrogen Common Stock is entitled to one vote for each share of stock held on the proposals presented in this Proxy Statement. Invitrogen s bylaws provide that a majority of all of the outstanding shares of stock entitled to vote, whether present in person or represented by proxy, constitutes a quorum for the transaction of business at the Annual Meeting.

# Why did I receive a one-page notice in the mail regarding the Internet availability of proxy materials this year instead of a full set of proxy materials?

Pursuant to new rules recently adopted by the SEC, we have elected to provide access to our proxy materials over the Internet. Accordingly, we are sending a Notice of Internet Availability of Proxy Materials (the Notice) to our stockholders of record and beneficial owners. All stockholders will have the ability to access the proxy materials on a website referred to in the Notice or request to receive a printed set of the proxy materials. Instructions on how to access the proxy materials over the Internet or to request a printed copy may be found in the Notice. In addition, stockholders may request to receive proxy materials in printed form by mail or electronically by email on an ongoing basis.

#### How do I vote?

All shares represented by a proxy will be voted, and where a stockholder specifies a choice with respect to any matter to be acted upon, the shares will be voted in accordance with the specification so made. If you do not indicate a choice on the proxy card, the shares will be voted in favor of the election of the nominees for director contained in this Proxy Statement, in favor of ratifying Ernst & Young LLP as independent auditors for the Company for 2008, in favor of amending the 1998 ESPP, in favor of amending the 2004 EIP and, in the discretion of the proxy holders, on any other matter that comes before the meeting.

Once you have given your proxy, you may revoke it at any time prior to the time it is voted, by delivering to the Secretary of the Company at the Company s principal offices either a written document revoking the proxy or a duly executed proxy with a later date, or by attending the Annual Meeting and voting in person. Merely attending the Annual Meeting will not, by itself, revoke a proxy.

Abstentions and broker non-votes will be counted for purposes of determining the presence or absence of a quorum. Broker non-votes are shares held by brokers or nominees who are present in person or represented by proxy, but which are not voted on a particular matter because instructions have not been received from the beneficial owner. The effect of broker non-votes and abstentions on the specific items to be brought before the Annual Meeting is discussed under each item.

*Voting via the Internet or by telephone.* If you vote via the Internet, you should be aware that there may be incidental costs associated with electronic access, such as your usage charges from your Internet access providers and telephone companies, for which you will be responsible. You may vote your shares telephonically by calling the telephone number referenced on your proxy card. Alternatively, you may vote via the Internet by visiting www.proxyvote.com and following the instructions on your screen.

*Voting by completing the proxy card.* If you properly complete and sign the enclosed proxy card and return it as instructed on the proxy card, it will be voted as you direct. If you hold your shares in your name and you attend the meeting, you may deliver your completed proxy card in person. If you hold your shares in street name through a brokerage or other nominee, you will need to obtain a proxy card from the institution that holds your shares.

#### How do I vote my 401(k) shares?

If you participate in the Invitrogen 401(k) Savings and Investment Plan you may vote the shares of the Company s common stock (Common Stock) in your account as of the record date. If you wish to vote those shares, you must complete your proxy card and return it in the envelope provided by April 28, 2008.

If you do not complete and return your proxy card prior to April 28, 2008, Fidelity Management Trust Company, the Plan trustee, will vote the shares in your account. You may revoke instructions to the trustee by giving it written notice of revocation or a later dated written voting instruction by April 28, 2008.

#### STOCK OWNERSHIP

#### How much stock is held by the Company s directors, executive officers and largest stockholders?

The following table sets forth information as of February 22, 2008, regarding the beneficial ownership of the Company s Common Stock by (i) each person known by us to own beneficially more than five percent of our outstanding Common Stock; (ii) each director and nominee for election as a director; (iii) each executive officer named in the Executive Summary Compensation Table; and (iv) all directors and executive officers as a group. Except as otherwise specified, the named beneficial owner has sole voting and investment power over the shares listed. Except as otherwise indicated, the address for each beneficial owner is c/o Invitrogen Corporation, 5791 Van Allen Way, Carlsbad, California 92008.

#### Stock Ownership Table

	Amount and Nature of Beneficial Ownership	Percentage of	
Name and Address of Beneficial Owner	of Common Stock <sup>(1)</sup>	Common Stock	
Barclays Global Investors UK Holdings Ltd <sup>(2)</sup>	2,714,941	5.9%	
State Street Global Advisors <sup>(3)</sup>	2,355,434	5.1%	

Gregory T. Lucier <sup>(4)</sup>	472,544	1.0%
Nicolas M. Barthelemy <sup>(5)</sup>	15,226	*

Name and Address of Beneficial Owner	Amount and Nature of Beneficial Ownership of Common Stock <sup>(1)</sup>	Percentage of Common Stock
Claude D. Benchimol, Ph.D. <sup>(6)</sup>	49,456	*
David F. Hoffmeister <sup>(7)</sup>	199,142	*
Peter M. Leddy, Ph.D. <sup>(8)</sup>	53,301	*
Donald W. Grimm <sup>(9) (18)</sup>	52,614	*
Raymond V. Dittamore <sup>(10) (18)</sup>	61,220	*
Balakrishnan S. Iyer <sup>(11) (18)</sup>	40,614	*
Bradley G. Lorimier <sup>(12) (18)</sup>	72,714	*
Ronald A. Matricaria <sup>(13) (18)</sup>	38,614	*
Per A. Peterson, Ph.D. <sup>(14) (18)</sup>	3,710	*
W. Ann Reynolds, Ph.D. <sup>(15) (18)</sup>	22,957	*
Jay M. Short, Ph.D. (16) (18)	50,768	*
David C. U Prichard, Ph.D <sup>(17) (18)</sup>	28,514	*
All Directors and Section 16 Executive Officers as group Total	1,322,941	2.89

- \* Less than 1%.
- (1) Beneficial ownership is determined in accordance with the rules of the SEC, based on factors including voting and investment power with respect to shares. Percentage of beneficial ownership is based on the number of shares of Common Stock outstanding as of February 29, 2008. Shares of Common Stock issuable upon conversion of convertible notes, or the exercise of options or warrants currently exercisable, or exercisable within 60 days after February 29, 2008, are deemed outstanding for the purpose of computing the percentage ownership of the person holding such options or warrants, but are not deemed outstanding for computing the percentage ownership of any other persons.
- (2) The address for Barclays Global Investors UK Holdings Ltd is Murray House, 1 Royal Mint Court, London EC3N 4HH, United Kingdom.
- (3) The address for State Street Global Advisors is One Lincoln Place, Boston MA 02111
- (4) Consists of 58,243 shares owned directly by Mr. Lucier, and 414,301 shares Mr. Lucier may acquire upon the exercise of stock options.
- (5) Consists of 10,581 shares owned directly by Mr. Barthelemy, and 15,000 shares Mr. Barthelemy may acquire upon the exercise of stock options.
- (6) Consists of 8,706 shares owned directly by Dr. Benchimol, and 40,750 shares Dr. Benchimol may acquire upon the exercise of stock options.
- (7) Consists of 1,892 shares owned directly by Mr. Hoffmeister, and 197,250 shares Mr. Hoffmeister may acquire upon the exercise of stock options.
- (8) Consists of 1,051 shares owned directly by Dr. Leddy, and 52,250 shares Dr. Leddy may acquire upon the exercise of stock options.
- (9) Consists of 4,000 shares owed by Donald and Kathryn A. Grimm, Trustees, the Grimm Family Trust dated January 31, 1986, 4,614 shares of restricted stock, and 44,000 shares Mr. Grimm may acquire upon the exercise of stock options.

- (10) Consists of 2,000 shares owned directly by a family trust in which Mr. Dittamore has a beneficial interest, 4,614 shares of restricted stock, 606 shares owned as Deferred Stock Units, and 54,000 shares that Mr. Dittamore may acquire upon the exercise of stock options.
- (11) Consists of 2,000 shares owned directly by Mr. Iyer, 4,614 shares of restricted stock, and 34,000 shares that Mr. Iyer may acquire upon the exercise of stock options.
- (12) Consists of 4,100 shares owned directly by Mr. Lorimier, 4,614 shares of restricted stock, and 64,000 shares Mr. Lorimier may acquire upon the exercise of stock options.
- (13) Consists of 10,000 shares owned directly by Mr. Matricaria, 4,614 shares of restricted stock, and 24,000 shares that Mr. Matricaria may acquire upon the exercise of stock options.
- (14) Consists of 2,751 shares of restricted stock, and 959 shares that Dr. Peterson may acquire upon the exercise of stock options.
- (15) Consists of 2,808 shares owned directly by Dr. Reynolds, 4,806 shares of restricted stock, and 15,343 shares that Dr. Reynolds may acquire upon the exercise of stock options.
- (16) Consists of 3,500 shares owned directly by Dr. Short, 4,614 shares of restricted stock, and 42,654 shares Dr. Short may acquire upon the exercise of stock options.
- (17) Consists of 2,900 shares owned directly by Dr. U Prichard 4,614 shares of restricted stock, and 21,000 shares that Dr. U Prichard may acquire upon the exercise of stock options.
- (18) Disclosures with respect to the stock ownership guidelines for each Director are set forth in the section titled Director Compensation below.

#### Securities Authorized for Issuance Under Equity Compensation Plans

Information about Invitrogen s equity compensation plans at December 31, 2007 is as follows (shares in thousands):

Plan Category	Number of Shares to be Issued Upon Exercise of Outstanding Options	A E P Out	eighted verage xercise rice of standing options	Number of Shares Remaining Available for Future Issuance	Weighted Average Remaining Contractual Life in Years
Equity compensation plans approved by stockholders <sup>(2)</sup>	5,406	\$	65.05	1,563(4)	7.3
Equity compensation plans not approved by stockholders <sup>(3)</sup>	288		34.97		5.2
Total	5,694	\$	63.53	1,563	7.2

 All option plans as of December 31, 2007, including the 1997 Invitrogen Corporation Stock Option Plan, the Invitrogen Corporation 2001 Stock Incentive Plan and the Invitrogen Corporation 2002 Stock Incentive Plan, have been frozen. Grants will no longer be made from such plans, as they have been replaced by the Invitrogen Corporation 2004 EIP.

- (2) Consists of the 1998 ESPP and seven stock option plans: the 1995 and 1997 Invitrogen Corporation Stock Option Plans, the 1996 and 1998 NOVEX Stock Option/Stock Issuance Plans, the Life Technologies 1995 and 1997 Long-Term Incentive Plans, and the 2004 EIP.
- (3) Represents the 2000 Invitrogen Corporation Stock Option Plan, the Invitrogen Corporation 2001 and 2002 Stock Incentive Plans, and options granted to Invitrogen s Chief Executive Officer. Stock options under the Invitrogen Corporation 2001 and 2002 Stock Incentive Plans were assumed as part of the Molecular Probes acquisition in August 2003. At December 31, 2007, these two assumed plans collectively total 33,913 shares to be issued upon exercise of outstanding options at a weighted average exercise price of \$12.26, with none available for future issuance. Pursuant to an employment agreement with its Chief Executive Officer, an option to purchase 675,000 shares of Invitrogen s Common Stock is included in this amount; of which 421,324 exercised as of December 31, 2007.
- (4) Includes 320,184 shares reserved for issuance under the 1998 ESPP.

The material features of the 2000 Invitrogen Corporation Stock Plan (the 2000 Plan) are set forth below. Only employees or consultants of the Company are eligible to receive awards under the 2000 Plan. The 2000 Plan provides for the award of stock options, which typically provide for 100% vesting after four years of service. The 2000 Plan provides that no option may be granted with an exercise price less than fair market value on the date of grant, except as allowed by Section 424(a) of the Internal Revenue Code of 1986, as amended. Upon a change in control, the vesting and exercisability of all outstanding awards under the 2000 Plan is 100% accelerated only to the extent an acquiring entity does not assume such outstanding awards.

The material features of the 2001 and 2002 Stock Incentive Plans are identical to one another. Only employees, consultants or directors of the Company who were hired after the closing of the Molecular Probes acquisition in August of 2003, or any such individuals who were previously employed by Molecular Probes, were eligible to receive awards under the assumed plans. The assumed plans provide for the award of either stock options or restricted stock. These plans typically provide for 100% vesting after four years of service. The plans provide that options, other than incentive stock options, may be granted with exercise prices less than fair market value on the date of grant, although the Company has never granted any options with an exercise price lower than fair market value. Upon a change in control, the vesting and exercisability of all outstanding awards under the plans are 100% accelerated only to the extent an acquiring entity does not assume such outstanding awards.

The material terms of the CEO Option described in Footnote 3 to the table above are as follows: (i) the exercise price is \$38.01, (ii) half of the option shares vested on the two-year anniversary of the option grant and the remaining half of the shares vest on the four-year anniversary of the option grant date, (iii) upon a change in control the CEO Option fully vests, (iv) upon the CEO s death or disability the CEO Option shall become vested in an amount which would reflect an additional twelve months of service by the CEO, and (v) upon the CEO s termination without cause or termination for good reason, the CEO Option shall become vested in an amount which would reflect an additional eighteen months of service by the CEO.

#### **ELECTION OF DIRECTORS**

The Company has a classified Board of Directors currently consisting of four Class III directors (Balakrishnan S. Iyer, Ronald A. Matricaria, W. Ann Reynolds, Ph.D., and Jay M. Short, Ph.D.) who will serve until the 2008 Annual Meeting of stockholders, three Class I directors (Gregory T. Lucier, Donald W. Grimm, and Per A. Peterson, Ph.D.) who will serve until the 2009 Annual Meeting of stockholders, and three Class II directors (Bradley G. Lorimier, Raymond V. Dittamore, and David C. U Prichard, Ph.D.) who will serve until the 2010 Annual Meeting of Stockholders, and in each case until their respective successors are duly elected and qualified. Directors in a class are elected for a term of three years to succeed the directors in such class whose terms expire at such annual meeting, or a shorter term to fill a vacancy in another class of directors.

Jay M. Short, Ph.D., has announced his retirement from the Board of Directors, effective at the time of the Annual Meeting. On February 27, 2008, pursuant to Section 2.1 of the Bylaws of the Company, the Board of Directors reduced the number of directors from ten to nine, effective upon Dr. Short s resignation at the Annual Meeting. Accordingly, the nominees for election at the 2008 Annual Meeting of Stockholders to fill three Class III positions on the Board of Directors are Balakrishnan S. Iyer, Ronald A. Matricaria, and W. Ann Reynolds, Ph.D. If elected, the nominees for the Class III positions will serve as directors until the annual meeting of stockholders in 2011, and in each case until their successors are elected and qualified. If a quorum is present and voted at the meeting, the three nominees for Class III director receiving the highest number of votes will be elected Class III directors.

The following information relates to the nominees listed above and to the Company s other directors whose terms of office will extend beyond the Annual Meeting.

#### Nominees for election at the 2008 Annual Meeting of Stockholders

#### **Class III**

#### (Term Ends 2011)

Balakrishnan S. Iyer

(age 51)

Presiding Director between April 21, 2004 and April 2006. Director since July 2001. Mr. Iyer is currently a director of Conexant Systems, Inc., Skyworks Solutions, Inc., Power Integrations, Inc., IHS Inc., and Qlogic Corporation. From October 1998 to June 2003, he was Senior Vice President and Chief Financial Officer of Conexant Systems, Inc. Mr. Iyer previously served as Senior Vice President and Chief Financial Officer of VLSI Technology Inc., where he was responsible for all worldwide financial functions, information technology and strategic planning. During his career, he has held a variety of other key management positions, including Finance Director and Group Controller for a \$1 billion business at Advanced Micro Devices. Mr. Iyer received his B.S. in Mechanical

Engineering from the Indian Institute of Technology, Madras and his M.S. in Industrial Engineering from the University of California, Berkeley. Mr. Iyer also received an M.B.A. in Finance from the Wharton School.

Ronald A. Matricaria (age 65)	Director since July 2004. Mr. Matricaria is the former Chairman and Chief Executive Officer of St. Jude Medical, Inc. Mr. Matricaria spent 23 years with Eli Lilly and Company, Inc., serving in several leadership roles. His last position was Executive Vice President of the Pharmaceutical Division of Eli Lilly and Company and President of its North American operations. He also served as President of Eli Lilly International Corporation. In 2002, he was recognized by the medical device industry with a lifetime achievement award. In addition, Mr. Matricaria is a member of the board of directors of Hospira, Inc. and Volcano Therapeutics, Inc., and is Trustee Emeritus of the University of Minnesota Foundation. Mr. Matricaria holds a bachelors degree from the Massachusetts College of Pharmacy and was awarded an honorary doctorate degree in Pharmacy in recognition of his contributions to the practice of pharmacy.			
W. Ann Reynolds, Ph.D. (age 70)	Director since February 2005. Dr. Reynolds is the former President of the University of Alabama at Birmingham. She retired as Director, Center for Community Outreach and Development, The University of Alabama at Birmingham in 2003. Prior to joining The University of Alabama at Birmingham as President in 1997, Dr. Reynolds served as Chancellor of the City University of New York, where she was responsible for the 21 colleges and professional schools that comprised that system. Prior to that, Dr. Reynolds was the Chancellor of the California State University system, chief academic officer of Ohio State University and associate vice chancellor for research and dean of the graduate college of the University of Illinois Medical Center. Earlier in her career, she held appointments as professor of anatomy, research professor of obstetrics and gynecology, and acting associate dean for academic affairs at the University of Illinois College of Medicine. A native of Kansas, Dr. Reynolds holds a masters degree and a Ph.D. in Zoology from the University of Iowa, as well as a bachelor s degree in Biology from Emporia State University, Kansas. She was a National Science Foundation Predoctoral Fellow and an honorary Woodrow Wilson Fellow. Dr. Reynolds is a director of Abbott Laboratories, Humana Inc., Owens Corning and the Champaign-Urbana News Gazette.			
The Board of Directors recommends a vote For the nominee named above.				

#### Directors Continuing in Office

#### Class I

#### (Term Ends 2009)

Chairman of the Board of Directors since April 2004, Chief Executive Officer of Invitrogen Corporation, Director since May 2003. From June 2000 to May 2003, Mr. Lucier was the President and Chief Executive Officer of
General Electric (GE) Medical Systems Information Technologies. Mr. Lucier was named a corporate officer of GE in 1999 by that company s board of directors and served in a variety of leadership roles during his career at GE, including Vice President of Global Services, GE Medical Systems. Mr. Lucier is currently a Director of BIOCOM and BIO, and is actively involved at San Diego State University as a distinguished lecturer. He received his B.S. in Engineering from Pennsylvania State University and an M.B.A. from Harvard Business School.
Director since June 1998. Mr. Grimm has been a director of Hamilton BioVentures, LLC, since August 2001. Since June 1995 he has served as Chairman and President of Strategic Design LLC, a strategic planning and consulting
company. Mr. Grimm retired from Eli Lilly & Company, a research-based pharmaceutical company, in December 1993 after 23 years of service. Mr. Grimm held positions at Eli Lilly as Director of Worldwide Pharmaceutical Pricing, Director of Pharmaceutical Market Research, and Director of Sales. Following these assignments, Mr. Grimm was President and CEO of Hybritech, Inc., a wholly owned subsidiary of Lilly. In addition, he is currently a director of several private companies. Mr. Grimm received his B.S. in Pharmacy and his M.B.A. from the University of Pittsburgh.
Director since March 2007. Dr. Peterson recently retired as Chairman, Research & Development, Pharmaceuticals at Johnson & Johnson. He joined Johnson & Johnson in 1994 as Vice President, Drug Discovery, of the R.W.
Johnson Pharmaceutical Research Institute. Dr. Peterson was named Group Vice President of the Pharmaceutical Research Institute in April 1998 and its president in November 1998. In 2000, he was named Chairman, Research & Development, Pharmaceuticals Group and became a member of the Executive Committee in 2001. Prior to joining Johnson & Johnson, Dr. Peterson spent eight years at Scripps Research Institute in La Jolla, CA, where he headed the Division of Molecular Immunogenics before being appointed Chairman of the Department of Immunology in 1987. He had earlier served as Director of the Wallenberg Laboratory, as well as professor of cell biology at the university of Uppsala, Sweden. Born in Kalmar, Sweden, Dr. Peterson received his B.M. in Medicine and his Ph.D. in Medicinal Biochemistry from the University of Uppsala, Sweden.

#### Class II

#### (Term Ends 2010)

Bradley G. Lorimier (age 62)	Director since November 1998. Mr. Lorimier served as Senior Vice President, Business Development and Director of Human Genome Sciences, Inc., a biotechnology company, from March 1994 to June 1997. He was a director of Matrix Pharmaceutical, Inc., from December 1997 to March 2002, and is a Director and Chairman of the Board of Avalon Pharmaceuticals, Inc. He is also a Director for several private companies. Mr. Lorimier received his B.S. in Biology from the University of Illinois.
Raymond V. Dittamore (age 64)	Director since July 2001. Mr. Dittamore also serves as a director of Gen-Probe Incorporated, Digirad Corporation, and QUALCOMM Incorporated. In June 2001, Mr. Dittamore retired as a partner of Ernst & Young after thirty-five years of service. He brings over three decades of public accounting experience to the Board of Directors, primarily serving companies in the life sciences industry. Mr. Dittamore received his B.S. from San Diego State University.
David C. U Prichard, Ph.D. (age 59)	Director since April 2004. Dr. U Prichard currently serves as a venture partner with the private equity firm Red Abbey Venture Partners LP (Baltimore, MD), and President of Druid Consulting LLC, a consulting firm specializing in the pharmaceutical and biotechnology industries. From September 1999 to April 2003 he served as CEO of 3-Dimensional Pharmaceuticals, Inc. Dr. U Prichard served as Chairman of Research and Development at SmithKline Beecham from July 1997 to March 1999. He served in senior R&D management positions at ICI/Zeneca from July 1986 to June 1997. Dr. U Prichard has served as an Associate Professor of Pharmacology and Neurobiology at Northwestern University Medical School and has held academic appointments at The John Hopkins University, and the Universities of Maryland and Pennsylvania. He is an honorary professor at the University of Glasgow. Dr. U Prichard serves as Chairman of the Board of Oxagen Limited (Oxford, UK) and Cyclacel Pharmaceuticals Inc. (Berkeley Heights, NJ) and is a Director of Alpharma Inc. (Bridgewater, NJ), and Silence Therapeutics Ltd (London, UK). Dr. U Prichard received his B.S. in Pharmacology from the University of Glasgow and a Ph.D. in Pharmacology from the University of Kansas.

How often did the Board meet during 2007?

During the fiscal year ended December 31, 2007, the Board of Directors held ten meetings. Each director serving on the Board of Directors in fiscal year 2007 attended at least 75% of the meetings of the Board of Directors and the committees on which he or she served. The Board meets in Executive Session, without any members of management present, at each regularly scheduled meeting of the Board of Directors. The independent directors elect a Presiding Director annually. Donald Grimm has served as the Presiding Director since April 2006. The Presiding Director presided at each Executive Session in 2007.

#### What is the Company s policy regarding attendance by the Board of Directors at the Annual Meeting of Stockholders?

Members of the Board are strongly encouraged to attend the 2008 Annual Meeting of Stockholders. At the 2007 Annual Meeting of Stockholders, all ten of the incumbent directors were present.

#### What committees has the Board of Directors established?

The Board of Directors has established an Audit Committee, a Compensation and Organizational Development Committee, a Governance and Nominating Committee, and a Science and Technology Committee. Each committee operates under a written charter approved by the Board of Directors. The charters of each committee are available on the Company s website at http://www.invitrogen.com. The Audit Committee consists of Mr. Dittamore, Mr. Grimm, Mr. Iyer, and Mr. Lorimier, and Mr. Dittamore serves as the Chairman. The Compensation and Organizational Development Committee consists of Mr. Matricaria, Dr. Reynolds and Dr. U Prichard, and Mr. Matricaria serves as the Chairman. The Governance and Nominating Committee consists of Mr. Iyer, Mr. Dittamore, and Mr. Matricaria, and Mr. Iyer serves as the Chairman. The Science and Technology Committee consists of Dr. U Prichard, Mr. Lorimier, Dr. Peterson, Dr. Reynolds, and Dr. Short, and Dr. U Prichard serves as the Chairman.

*Audit Committee*. The Audit Committee s function is to review with our independent auditors and management the annual financial statements and independent auditors opinion, review and maintain direct oversight of the plan, scope and results of the audit by the independent auditors, review and approve all professional services performed and related fees charged by the independent auditors, be solely responsible for the retention or replacement of the independent auditors, and monitor the adequacy of the Company s accounting and financial policies, controls, and reporting systems. During 2007, the Audit Committee held six meetings.

The Board of Directors and the Audit Committee believe that the Audit Committee s current member composition satisfies the rule of the National Association of Securities Dealers, Inc. (NASD) that governs audit committee composition, including the requirement that audit committee members all be independent directors as that term is defined by NASD Rule 4200 (a)(15) and the definition of independent under the Sarbanes-Oxley Act of 2002. Additionally, the Company certifies that it has, and will continue to have, at least one member of the Audit Committee that is defined as an audit committee financial expert in accordance with Section 407 of the Sarbanes-Oxley Act with past employment experience in finance or accounting, requisite professional certification in accounting, or any other comparable experience or background which results in the individual s financial sophistication, including being or having been a chief executive officer, chief financial officer or other senior officer with financial oversight responsibilities. Currently, the Board has determined that Raymond V. Dittamore and Balakrishnan S. Iyer are audit committee financial experts.

*Compensation and Organizational Development Committee*. The functions of the Compensation and Organizational Development Committee in 2007 included providing guidance to management and assisting the Board of Directors in matters relating to the compensation of the Chief Executive Officer and senior executives, the organizational structure of the Company, the Company s compensation and benefits programs, the Company s succession, retention and training programs, and such other matters that have a direct impact on the success of our human resources. During 2007, the Compensation and Organizational Development Committee held six meetings.

The Board of Directors and the Compensation and Organizational Development Committee believe that the Compensation and Organizational Development Committee s current member composition satisfies the rule of the NASD that governs committee composition, including the requirement that committee members all be independent directors as that term is defined by NASD Rule 4200 (a)(15) and the definition of independent under the Sarbanes-Oxley Act of 2002.

*The Governance and Nominating Committee.* The functions of the Governance and Nominating Committee include leading any searches for new Board of Director candidates, reviewing and making recommendations to the Board regarding director compensation, and making recommendations to the Board regarding director nominees to be put forth by the Board at each annual meeting of stockholders. In addition, the area of corporate governance has taken on increasing importance in the creation and preservation of shareholder value. Therefore, the Governance and Nominating Committee focuses on core processes that the Board and its committees utilize to carry out their responsibilities, including fundamental issues such as how decisions are made. During the year ended December 31, 2007, the Governance and Nominating Committee held four meetings.

The Governance and Nominating Committee will consider for inclusion in its nominations of new directors those nominees recommended by stockholders who have held at least 1% of the outstanding voting securities of the Company for at least one year. Board candidates referred by such stockholders will be considered on the same basis as Board candidates referred from other sources. Any stockholder who wishes to recommend for the Governance and Nominating Committee s consideration a prospective nominee to serve on the Board of Directors may do so by giving the candidate s name and qualifications in writing to the Company s Secretary at the following address: 5791 Van Allen Way, Carlsbad, CA 92008.

The Board of Directors and the Governance and Nominating Committee believe that the Governance and Nominating Committee s current member composition satisfies the rule of the NASD that governs committee composition, including the requirement that committee members all be independent directors as that term is defined by NASD Rule 4200 (a)(15) and the definition of independent under the Sarbanes-Oxley Act of 2002.

*The Science and Technology Committee*. The Science and Technology Committee examines management s direction and investment in the Company s research and development and technology initiatives. The Science and Technology Committee functions as a broadly knowledgeable and objective group of scientists and non-scientists to consider and report periodically to the Board on matters relating to the investment in the Company s research and development and technology initiatives. The Science and Technology Committee s actions are generally related to high-level policy and strategy. The administration of the research and development function remains the responsibility of management. During the year ended December 31, 2007, the Science and Technology Committee held five meetings.

#### **REPORT OF THE AUDIT COMMITTEE**

The purpose of the Audit Committee is to assist the Board in its general oversight of Invitrogen s financial reporting, internal controls and audit functions. As described in the Audit Committee Charter, which is available at out website, http://www.invitrogen.com, the Audit Committee has oversight responsibilities to stockholders, potential stockholders, the investment community, and other stakeholders related to the:

financial reporting process;

systems of internal accounting and financial controls;

performance of the Company s internal audit function and independent auditors;

independent auditor s qualifications and independence; and

compliance with ethics policies and legal and regulatory requirements. The Audit Committee is composed solely of independent directors as defined by the listing standards of the National Association of Securities Dealers, Inc.

The Audit Committee has reviewed and discussed the consolidated financial statements with management and Ernst & Young LLP, the Company s independent auditors. Management is responsible for the preparation, presentation and integrity of Invitrogen s financial statements; accounting and financial reporting principles; establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rule 13a-15(e)); establishing and maintaining internal control over financial reporting (as defined in Exchange Act Rule 13a-15(f)); evaluating the effectiveness of disclosure controls and procedures; evaluating the effectiveness of internal control over financial reporting; and evaluating any change in internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, internal control over financial reporting. Ernst & Young LLP is responsible for performing an independent audit of the consolidated financial statements and expressing an opinion on the conformity of those financial statements with accounting principles generally accepted in the United States of America, as well as expressing an opinion on the effectiveness of internal control over financial reporting.

During 2007 the Audit Committee provided oversight and advice to management relating to management s assessment of the adequacy of Invitrogen s internal control over financial reporting in accordance with the requirements of the Sarbanes-Oxley Act of 2002. The Audit Committee received periodic updates from management and Ernst & Young LLP relating to such assessment. The Audit Committee held regular private sessions with Ernst & Young LLP to discuss their audit plan for the year, the results of their quarterly reviews, and the annual audit. At the conclusion of the process, the Audit Committee reviewed a report from management on the effectiveness of the Company s internal control over financial reporting. The Committee also reviewed the report of management contained in the Company s Annual Report on Form 10-K for the year ended December 31, 2007, filed with the SEC, as well as Ernst & Young LLP s Report of Independent Registered Public Accounting Firm included in the Company s Annual Report on Form 10-K related to its audit of (i) the consolidated financial statements and financial statement schedule, and (ii) the effectiveness of internal control over financial reporting.

The Audit Committee provided oversight and guidance to members of management, including the General Counsel, Director of Internal Audit (who reports to the Audit Committee), and Director of Compliance on the Company s policies and procedures relating to risk assessment and risk management and on the legal and regulatory compliance programs. The Committee received periodic reports on these matters throughout the year.

The Audit Committee met on six occasions in 2007. The Audit Committee met privately with each of Ernst & Young LLP, the internal auditor, and the Chief Financial Officer at each regular meeting.

Invitrogen has an internal audit department that reports directly to the Audit Committee. The Audit Committee reviews and approves the internal audit plan and receives regular updates on internal audit activity. Updates include discussion of results and findings by the internal audit team, follow up, staffing level of the internal audit function, and assessment of internal controls and risk of fraud.

The Audit Committee has discussed with Ernst & Young LLP the matters required to be discussed by Statement on Auditing Standards No. 61, as amended, Communication with Audit Committees and PCAOB Auditing Standard No. 5, An Audit of Internal Control Over Financial Reporting That Is Integrated with an Audit of Financial Statements. In addition, Ernst & Young LLP has provided the Audit Committee with the written disclosures and the letter required by the Independence Standards Board Standard No. 1, as amended, Independence Discussions with Audit Committee has discussed with Ernst & Young LLP their firm s independence. In addressing the quality of management s accounting judgments, the Audit Committee asked for management s representations and reviewed certifications prepared by the CEO and CFO that the unaudited quarterly and audited consolidated financial statements of the Company fairly present, in all material respects, the financial condition and results of operations of the Company.

Based on the review of the consolidated financial statements and discussions with and representations from management and Ernst & Young LLP referred to above, the Audit Committee recommended to the Board of Directors that the audited financial statements be included in Invitrogen s Annual Report on Form 10-K for the year ended December 31, 2007, for filing with the Securities and Exchange Commission.

In accordance with Audit Committee policy and the requirements of law, the Audit Committee pre-approves all non-audit services to be provided by Invitrogen s outside auditors, Ernst & Young LLP. In addition, the Audit Committee pre-approves all audit and audit related services provided by Ernst & Young LLP. The Audit Committee has delegated to its Chairman the ability to pre-approve non-audit services. Such pre-approval is later reported to the Audit Committee. A further discussion of the fees paid to Ernst & Young LLP for audit and non-audit expenses is included below under the heading **PRINCIPAL ACCOUNTING FEES & SERVICES**. Although the Audit Committee has the sole authority to appoint independent auditors, the Audit Committee is continuing its long-standing practice of recommending that the Board ask the stockholders to ratify the appointment at the annual meeting.

#### AUDIT COMMITTEE

Raymond V. Dittamore, Chairman

Donald W. Grimm

Balakrishnan S. Iyer

Bradley G. Lorimier

#### PRINCIPAL ACCOUNTING FEES AND SERVICES

In connection with the audit of the 2007 financial statements, the Company entered into an engagement agreement with Ernst & Young LLP which set forth the terms by which Ernst & Young LLP has performed audit services for the Company. That agreement is subject to alternative dispute resolution procedures.

The following table sets forth the aggregate fees agreed to by the Company for the annual and statutory audits for the fiscal years ended December 31, 2007 and 2006, and all other fees paid by the Company during 2007 and 2006 to its principal accounting firm, Ernst & Young LLP:

	For the Ended Dec	
(in thousands)	2007	2006
Audit Fees	\$ 3,412	\$ 2,969
Audit-Related Fees	933	392
Tax Fees	732	273
All Other Fees	0	0
Total	\$ 5,077	\$ 3,634

Total

The Audit Committee has determined that the rendering of all non-audit services by Ernst & Young LLP is compatible with maintaining the auditor s independence. The fees listed under Audit Fees above were incurred for service related to the annual audit of the Company s consolidated financial statements, reviews of interim consolidated financial statements and services that are normally provided in connection with statutory and regulatory filings and engagements. In 2007 and 2006, audit fees also included fees incurred for the audits of the effectiveness of internal control over financial reporting. The fees listed under Audit-Related Fees above were incurred for service related to mergers and acquisitions, dispositions, and benefit plan audits, and the fees listed under Tax Fees above were incurred for service related to federal, state and international tax compliance and planning. The Audit Committee approves non-audit services by Ernst & Young LLP on an ad hoc basis, and has vested authority with Raymond V. Dittamore, the chairman of the Audit Committee, to approve non-audit services as needed.

#### EXECUTIVE COMPENSATION

#### COMPENSATION DISCUSSION AND ANALYSIS

#### Introduction/Corporate Governance

Compensation and Organizational Development Committee Members and Compensation and Organizational Development Committee Charter

Committee Members

The Compensation and Organizational Development Committee of the Board of Directors (the Committee) is made up of the following three Board members: Ronald A. Matricaria, who serves as Chairman, W. Ann Reynolds, Ph.D. and David C. U Prichard, Ph.D.

The Board of Directors and the Committee annually determine whether the Committee s current membership satisfies the rule of the NASD that governs committee composition, including the requirement that committee members all be independent directors as defined by NASD Rule 4200 (a)(15) and the definition of independent under the Sarbanes-Oxley Act of 2002. The Board of Directors and the Committee believe that the current members of the Committee satisfy these requirements.

#### Charter and Functions of the Committee

The functions of the Committee in 2007 included providing guidance to management and assisting the Board of Directors in matters relating to:

the compensation of the Chief Executive Officer (CEO) and senior executives;

our organizational structure;

our compensation and benefits programs; and

our succession, retention and training programs. The Committee s charter states that the Committee will:

collaborate with executive management in developing a compensation philosophy;

make recommendations to the Board of Directors on the compensation of the CEO (the full Board of Directors participates in the evaluation of the CEO and makes final decisions regarding CEO compensation);

evaluate and approve compensation for the other executive officers; and

oversee the general employee benefit programs, including our 2004 EIP and our 1998 ESPP. The Committee reviews the adequacy of its charter at least annually. The Committee did not make any material changes to its charter as a result of the most recent review in July 2007. The Committee s complete charter is available at our web site at:

http://phx.corporate-ir.net/phoenix.zhtml?c=61498&p=irol-govhighlights

The Committee Chairman is responsible for setting the Committee s meeting agenda and calendar.

#### Compensation Consultant

The Compensation Group in our Human Resources Department supports the Committee in fulfilling its charter. In addition, the Committee has the authority under its charter to engage the services of outside advisors, experts and others to assist the Committee. In accordance with this authority, the Committee, beginning in September 2006, has engaged DolmatConnell & Partners, Inc. (DolmatConnell or the Consultant), as independent outside compensation consultant to advise the Committee on matters related to CEO and other executive compensation. DolmatConnell advises the Committee and our management on executive compensation benchmarking and program design and frequently attends the Committee meetings.

The Consultant:

recommends to the Committee the industry peer group (the Peer Group) for purposes of comparison and benchmarking executive compensation; and

performs compensation analyses, focused on comparisons of financial performance and compensation as requested by the Committee and management.

The Consultant s assignments are determined by the Committee Chairman and/or management as directed by the Committee.

Roles of Executives in Establishing Executive Compensation

Our CEO, Senior Vice President of Human Resources (the SVP of HR), and Chief Financial Officer (the CFO) are also involved in the executive compensation process.

Gregory T. Lucier, our CEO:

reviews with the Committee the performance of his direct reports;

recommends to the Committee base salary increases, ratings for the individual component of the Incentive Compensation Plan (annual bonus) and stock award levels for the long-term incentive plan;

recommends short-term and long-term Company financial and non-financial performance goals that are used throughout many components of the compensation plans; and

advises the Committee regarding the executive compensation program s ability to attract, retain and motivate the level of executive talent necessary to achieve our goals.

David F. Hoffmeister, SVP and Chief Financial Officer, provides:

input on the financial targets for our Incentive Compensation Plan; and

data regarding the impact of the executive compensation programs on our financials. Peter M. Leddy, Ph.D., SVP of HR, provides:

external data, including compensation benchmarking for employees below the executive level, and workforce dynamics in the marketplace;

data, including turnover ratios, 360 feedback on peers, and performance appraisal information; and

quarterly reports of all equity grants that we make. The CEO and the SVP of HR attend the meetings of the Committee, but do not participate in the Executive Sessions.

The Committee delegates to the SVP of HR and the Vice President of Global Compensation and Benefits the ability to approve long-term incentive awards to our new hires and employees in the amount of 18,000 options or less and 3,000 restricted stock units or less, within defined parameters. The Committee has approved these parameters to provide appropriate incentives to different career bands within Invitrogen. We have a written policy addressing the appropriate dating and pricing of the shares and options. We make equity grants on the 1<sup>st</sup> day of the month following the approval of the grant by the Committee or by management within the parameters described above. Stock options have an exercise price equal to the closing price for our Common Stock on the Nasdaq National Market on the date of grant. Additionally, the Committee has authorized the CEO to approve any base salary increases, bonuses, or new-hire offer packages with the exception of those for officers who are

subject to the requirements of Section 16 of the Securities and Exchange Act of 1934 (Section 16 Officers).

#### Committee Activity

Recognizing the importance of maintaining (i) sound principles for the development and administration of executive compensation and (ii) strong links between executive pay and performance, the Committee took the following steps, among others, in 2007:

held Executive Sessions (without Company management present) at every Committee meeting;

updated our industry-specific peer group for executive compensation and performance comparisons;

eliminated the Mid-term Incentive Plan (discussed below) as a component of the executive compensation program;

implemented a Performance-Based Restricted Stock Unit Plan for the CEO and the executive officers that provides a closer link between pay and pre-determined financial targets; and

implemented an annual review of detailed executive compensation and benefits tally sheets for all executive officers. Using the tally sheets provided by management and the competitive analysis provided by the Consultant, the Committee is able to review both the competitiveness and appropriateness of each element of executive compensation. The Committee considered several factors in these reviews, including:

each executive s total compensation;

all equity awards that have been granted to each executive since starting employment with Invitrogen;

the total potential value of all equity awards made to each executive officer, based on several potential share price growth scenarios; and

the degree of internal parity between executives based on performance and importance of the executive to our company. The Committee also considers the competitive market for executive compensation. The Committee seeks to maintain competitive compensation because we believe that attracting and retaining exceptional talent is a key component in building a sustainable competitive advantage in the market. The Committee considers the need and rationale for each element of compensation and the amounts that are targeted and awarded in relation to our performance vis-à-vis the performance of the Peer Group.

After these reviews, the Committee believes that both the individual elements of compensation and the compensation in total for each NEO for 2007 are appropriate given our performance, the position vis-à-vis the competitive labor market, and his/her relative performance and importance to Invitrogen.

During 2007, the Committee held six meetings, which were attended by the Committee members, the CEO, the SVP of HR, and the VP of Global Compensation and Benefits. The Committee also held six Executive Sessions during 2007, which were attended by the Committee members. The Consultant attended several of the meetings and several of the Executive Sessions as requested by the Committee.

#### **Objectives of Compensation Programs**

#### Compensation Philosophy

The primary underlying premise of our executive compensation philosophy is that pay should be performance-based, vary with the attainment of specific objectives, and be aligned with the interests of Invitrogen s stockholders. The Committee s primary objective is to employ compensation to differentiate and reward individual performance based on our overall business results, progress toward individual goals and objectives, and leadership behaviors consistent with our long-term success. The Committee employs the following core principles to guide its decisions.

*Pay competitively:* The Committee believes in positioning executive compensation at competitive levels necessary to attract and retain exceptional leadership talent. Performance can result in an individual s total compensation that is higher or lower than market position. The Consultant compiles this competitor and market data at the request of, and working with, the Committee.

*Pay-for-performance:* The Committee structures executive compensation programs to balance annual and long-term corporate objectives. These programs include the specific measures of financial performance, operational objectives, and total stockholder return, with the goal of enhancing stockholder value in the short and long term.

*Foster an ownership mentality:* The Committee believes that using compensation to help build an ownership culture effectively aligns the interest of management and our stockholders. Accordingly, the Committee requires executives to acquire and retain specific levels of Company stock. Moreover, the Committee utilizes equity based compensation for the CEO and the executive officers, including performance-contingent restricted stock units, to provide incentives for the CEO and the executive officers to enhance stockholder value.

*Have a total compensation perspective:* The Committee views all components of pay together in making compensation decisions. These components include base salary, annual incentives, long-term incentives, fringe benefits and perquisites.

The Committee reviews its compensation philosophy regularly, most recently in October 2007. The Committee believes its compensation philosophy is based on appropriate principles and did not make any changes to the overall philosophy in 2007. However, the Committee made programmatic changes to align better the executive compensation system with the Committee s philosophy. We describe these changes in detail in the Compensation Programs Design section below.

#### Benchmarking

We benchmark all elements of total direct compensation (base salary, bonus, total cash compensation, and all forms of long-term incentives) to the competitive marketplace. Working with the Consultant, the Committee considered several factors to determine the companies in the Peer Group. The Committee has in this Peer Group companies that:

provide reasonable comparisons for pay and performance purposes;

generally overlap with our labor market for talent, but may not be identical;

exhibit revenue and market capitalization size within approximately 1/3x to 3x our revenue and market capitalization (about 50% of the group is larger and 50% is smaller, with modest exceptions for industry leaders);

possess a business model, characteristics, growth potential, and human capital intensity that are similar, though they need not be identical; and

are U.S.- based public companies so that proxy statements and 10-K s can be used to provide the appropriate compensation and firm financial data.

For 2007, the Peer Group consisted of: Affymetrix, Inc., Allergan, Inc., Applera Corp., Beckman Coulter, Inc., Becton, Dickinson and Co., Biogen Idec, Inc., Bio-Rad Laboratories, Inc., Charles River Laboratories Intl, Inc., Gen-Probe, Inc., PerkinElmer, Inc., Sigma-Aldrich Corp., St. Jude Medical, Inc., Thermo Fisher Scientific, Inc., and Waters Corp.

The Committee considers the following performance metrics of Invitrogen in developing executive compensation: revenue growth, net income margin, operating margin, EBITDA margin, gross profit margin, adjusted pro forma net income margin and total stockholder return. For determining 2007 executive compensation, we compared our performance against the Peer Group for the one-year period of 2006, and the three-year period from 2004 through 2006. In October 2007, we supplemented this data to include the first two quarters of 2007, and in February 2008, we supplemented the date with information for the full year 2007.

Over the three-year period inclusive of 2004 to 2006, four financial performance measures met or exceeded the Peer Group 50th percentile, and three financial performance measures were below the Peer Group 50th percentile. For calendar year 2006, gross profit margin and adjusted pro forma net income margin exceeded the Peer Group 50th Percentile, and the remaining financial measures were below the Peer Group 50th percentile, and four financial performance measures were below the Peer Group 50th percentile, and four financial performance measures were below the Peer Group 50th percentile. For the full year 2007, five performance measures met or exceeded the Peer Group 50th percentile, and two financial performance measures were below the Peer Group 50th percentile. For the full year 2007, five performance measures met or exceeded the Peer Group 50th percentile, and two financial performance measures were below the Peer Group 50th percentile. For 2007, we had the highest total shareholder return within our Peer Group.

We target base salaries to approximate the Peer Group 55th percentile and total cash compensation to approximate the Peer Group 65th percentile. The base salaries, total cash compensation, long-term incentives, and total target direct compensation of our Named Executive Officers (NEOs) in 2007 were between the 60th percentile and the 75th percentile of individuals in comparable positions in the Peer Group.

#### **Compensation Program Design**

Elements of Compensation

Elements of In-Service Compensation

#### Compensation Mix

Our NEOs base salaries and target bonuses as a percentage of total compensation approximated the 60th percentile of the Peer Group in 2007, and the value of long-term incentives as a percentage of the total compensation were higher than the target percentile within the Peer Group. Because we did not pay ICP bonuses in 2007 (for 2006 performance), our NEOs total cash compensation was below the target for our Peer Group. Our aggressive long term incentives in 2007, in the form of stock options, time vested restricted stock units and performance vested restricted stock units, are designed to provide performance incentives and executive retention in light of the lower cash compensation in 2007.

#### Base Salary

Base salaries are the only fixed element of total compensation. They reflect each executive s responsibilities, the impact of the job, and the contributions each executive delivers to Invitrogen. Salaries are determined in part by competitive levels in the market what companies in the Peer Group and executive compensation surveys pay executives with comparable responsibilities and job scope and in part by internal parity considerations. Each year, the Committee reviews and establishes the base salary of Invitrogen s executive officers. Increases, if any, are based on individual performance, existing employment agreements and market conditions. To gauge market conditions, the Committee evaluates the competitor and market data compiled by the Consultant.

The Committee reviews officer salaries annually at the end of the year. At its December 2007 meeting, the Committee reviewed recommendations for salary adjustments for the Company s executive officers, excluding the CEO and CAO. The Committee reviews the performance and compensation of the CEO in several meetings from October to February of each year, and the Board makes adjustments to the CEO s compensation effective January <sup>st</sup>. In February 2008, the Board voted to increase the CEO s base salary to \$975,000 for 2008.

#### Non-Equity Incentive Plan Compensation

In most regions around the world, employees at Invitrogen, from the CEO down, are on the same annual bonus plan, called the Incentive Compensation Plan (ICP). The 2007 ICP provides for an annual cash bonus based on the achievement of corporate financial goals and for results relative to individual objectives.

The 2007 ICP will be funded based on operating income. The ICP for 2007 still features a performance threshold: we must achieve 95% of the operating income target before any ICP bonus is earned. If we do not achieve at least 95% of the operating income target, no incentive compensation bonuses will be paid under the ICP. However, we have the discretion to maintain a small non-incentive bonus reserve for exceptional employee recognition, reward and retention. Once we meet the threshold for operating income, we will fund an incentive bonus pool, and we will assign each eligible employee a preliminary bonus amount based on company performance.

Each employee s preliminary bonus amount for 2007 is subject to an individual multiplier based on the employee s overall performance, achievement of individual goals and objectives, and the relative contribution as determined through our performance calibration process. An employee may earn from 0% to 200% of the target bonus, depending on his/her performance as measured against the objectives set forth at the beginning of the ICP year or during the annual performance appraisal process, the employee s overall performance, and their relative contribution as determined through our performance calibration process.

Target awards are expressed as a percentage of an executive s base salary. Bonus targets for 2007 were 150% for the CEO and 60% for other NEOs. The total award to any one individual is capped at 200% of the individual s target. Based on the Company s strong financial performance in 2007, the Board of Directors awarded an ICP bonus of \$2,047,500 to the CEO, and the Committee approved bonuses for the NEOs as follows:

David F. Hoffmeister, Chief Financial Officer	\$ 332,640
Claude D. Benchimol, Ph.D., Senior Vice President, Research & Development	\$ 280,665
Peter M. Leddy, Ph.D., Senior Vice President, Human Resources	\$ 293,895
Nicolas M. Barthelemy, Senior Vice President, Cell Systems	\$ 293,580

For 2007 only, employees in the leadership paybands of the company, including but not limited to executive officers, may choose to receive the payment in one of two ways:

- 1) 100% cash, paid in mid-March 2008; or
- 2) 70% cash, paid in mid March 2008, and 30% in RSUs granted in mid-March 2008, which vest 50% after one year, with the remaining 50% vesting in two years.

As an incentive, those employees who choose to receive RSUs will receive a number of RSUs equal to 150% of the cash value that they would otherwise receive. In other words, a leadership level employee who is eligible for a \$10,000 ICP payment for 2007, and who elects to receive RSUs will receive a cash payment of \$7,000, and RSUs worth \$4,500 on the date of grant. We believe that the payout features allow the Company to conserve cash, and provide a way to retain employees at the leadership levels within the Company. The RSUs offered in lieu of cash vest over a two year period, which creates more retention through our 2007 ICP.

The 2008 ICP is substantially similar to the 2007 ICP, but features updated financial targets and an updated threshold for operating income that was adopted by the Committee in December 2007 based on our 2008 annual operating plan approved by the Board of Directors. The Committee determined not to continue RSUs as a distribution choice under the 2008 ICP in order to help conserve shares. If earned, distribution will be made 100% in cash.

#### Mid-Term Incentive Plan

The Committee oversaw our Mid-Term Incentive Plan (the Plan), which we designed to focus our executives on ambitious mid-term strategic goals, and provide a performance-based opportunity for senior management to earn shares of our stock. Awards under our Mid-Term Incentive Plan were tied to achievement of the Company s three-year strategic plan created in 2004, with target awards equal to each individual s 2006 target ICP percentage. We paid earned awards approximately 50% in cash and 50% in restricted stock units, which were subject to an additional three-year vesting requirement to aid in the retention of key executives. The Committee determined the targets for the Mid-Term Incentive Plan, with payments based 50% on three-year average revenue growth and 50% on achievement of cumulative free-cash-flow-return-on-average-capital-employed (i.e., the rate of return on invested capital).

The payout for the Plan in March 2007 was at 94% of target based on the three-year performance that was achieved from 2004 to 2006.

In October 2006, the Committee determined not to continue with the mid-term incentive program in 2007. The Committee determined that increasing the target bonuses for the ICP was more effective to drive short-term goals, and implementing performance-based equity grants would be more effective to achieve long-term results for our stockholders.

#### Long-Term Incentives

We designed the long-term incentive plan to align executives interests with those of our stockholders, promote personal ownership of Invitrogen stock and reward superior performance. We believe that the plan helps to reduce officer and employee turnover and to retain the knowledge and skills of our valued officers and employees. The plan utilizes both stock options and restricted stock units, to strike a balance between:

rewarding financial progress with the retention value of restricted stock units; and

achieving superior financial results in the long term with the grant of stock options. The Committee bases individual grants of equity-based awards on various factors, including:

Company results;

individual performance;

individual potential contribution to our success; and

competitor and market data.

In determining the size of awards, the Committee weighs the dilutive effects of annual equity-based awards against the need to provide attractive and competitive incentive compensation.

We grant non-qualified stock options at an exercise price equal to the fair market value of Invitrogen s Common Stock on the date of the grant. Options vest over four years, twenty-five percent on each anniversary of the grant date. The Committee believes that this program encourages officers and other employees to work with a long-term view of our performance and to reinforce their long-term affiliation with Invitrogen.

We typically grant annual equity awards to the CEO in March and to the other executive officers in May. At its April 2007 meeting, the Committee reviewed the analyses and recommendations for executive officer (other than the CEO) stock option grants provided by

management. In reviewing the recommended grants, the Committee considered:

each officer s performance and contribution during the fiscal year;

analyses reflecting the value delivered;

competitive practices; and

the proportion of options granted to each NEO, and the NEOs in aggregate, as a percentage of total options granted during the fiscal year.

The Summary Compensation Table includes the stock option grants to the NEOs approved by the Committee and the grant to the CEO approved by our Board of Directors. We grant time vested restricted stock and performance vested restricted stock as a way of retaining key employees. By aligning the interests of the CEO and our NEOs with the interests of the Company, the possibility of personal financial rewards can be achieved by ensuring our corporate long-term financial success. In this way the restricted stock provides additional and different incentives than the stock options granted to management as described below.

In 2007, we provided long-term incentives in the form of stock options, time-vested restricted stock units, and performance based restricted stock units. For the executive officers, we granted approximately 60% of the long-term incentive opportunity in the form of stock options, approximately 20% of the long-term incentive award in the form of time vested restricted stock units, and 20% in performance based restricted stock units. The Committee believes that this approach to long-term incentive compensation builds upon its pay-for-performance philosophy. In addition, this practice incorporates the growing prevalence in the marketplace of an incentive approach balancing short term and long term goals. This portfolio of long-term incentives provides a strong incentive to reach financial goals (via stock options and performance accelerated restricted shares) and a strong retention focus through the time-based restricted shares.

On February 28, 2007, the Board of Directors approved a grant of 400,000 performance shares to the CEO. The performance share grant, which was effective March 1, 2007, will vest on February 28, 2010, only if:

the price of our Common Stock meets certain targets in the interim; and

the CEO remains employed by the Company on February 28, 2010 (if the CEO s employment is terminated due to his death, his disability, a not-for-cause termination by the Company, or termination by the CEO for Good Reason, any tranche of performance shares associated with a targeted stock price that has been achieved will vest upon the termination).

None of the performance shares will be earned unless the average closing price of our common stock over a continuous twenty trading day period meets or exceeds \$80.00 per share, and then only five percent of the performance shares will be earned. The remaining performance shares would be earned, based on achieving targeted closing prices for our common stock between \$80.00 and \$105.00 per share, and would vest on February 28, 2010. For each target closing price between \$80.00 and \$105.00 per share, no performance shares will vest unless the average closing price of our common stock over a twenty-day period meets or exceeds the applicable target closing price per share and the CEO is still employed by the Company on February 28, 2010.

If we achieve higher targeted closing prices, then larger portions of the performance share grant will vest. The Board of Directors intends that this will be the only equity grant that it makes to the CEO for the years 2007 through 2009. The performance target for vesting of the performance shares in their entirety represents a significant increase from the Company s current stock price. We cannot assure you that we will achieve any of the remaining performance targets. However, the Board of Directors and the Committee believes that this grant and the related performance targets further align the interests of the CEO and the stockholders over the long-term, and will encourage behaviors in the CEO that will provide increases in stockholder value. The Board of Directors and Committee believe that other elements of the CEO s compensation are aligned toward near-term goals.

#### Personal Benefits and Perquisites

The Committee oversees the design, implementation and administration of all Company-wide benefit programs. We offer a limited number of additional benefit programs to our senior management. The perquisites and benefits include supplemental long-term disability insurance, a financial counseling allowance, a non-qualified deferred compensation plan, and an executive comprehensive health examination on an annual basis. The amounts relating to these perquisites are disclosed in the footnotes to the Summary Compensation Table in this Proxy Statement. The Committee, with the assistance of the Consultant, periodically reviews the cost and prevalence of these programs to ensure these programs are in line with competitive practices and are warranted, based upon the business need and contributions of the executive team.

### Employee Stock Purchase Plan

We have an Employee Stock Purchase Plan (ESPP) in which most employees globally, including executives, may choose to participate. Participants may purchase up to \$25,000 in Company stock per calendar year at a price equal to 85% of the fair market value of the stock at the lower of the beginning or end of each three-month offering period for the ESPP. The price for each offering period is carried to the next offering period for up to two years, unless 85% of the fair market value at the end of a subsequent offering period would be lower. The Committee believes that the ESPP provides another effective vehicle for enabling our executives and employees to increase their ownership position in Invitrogen, thereby promoting an ownership mentality and a tighter link between the interests of employees and stockholders.

### **Employment Agreements**

We use employment agreements for our executive officers in very select cases, generally when it is necessary to secure the services of a newly hired executive. We have entered into employment agreements with:

Gregory T. Lucier, the current Chairman and Chief Executive Officer, effective May 30, 2003; and

David F. Hoffmeister, our Chief Financial Officer, Senior Vice President, Finance, effective October 13, 2004. We have entered into letter agreements with each of our other executive officers outlining the terms of their employment and the elements of their compensation. Each of these letter agreements follows our standard employment offer template, and provides for employment at will.

The compensation resulting from the employment agreements and letter agreements are disclosed as required in the tabular and narrative disclosures below.

#### Elements of Post-Termination Compensation

#### Change-in-Control Agreements

We have change-in-control agreements in place for the CEO and executive officers who are direct reports to the CEO. The rationale for these agreements is that in the event of a change in control of Invitrogen, these individuals are the most likely to lose their jobs as a result of redundancy in executive positions.

The agreements that are in place are double trigger agreements there must be a change in ownership and a termination or constructive termination of employment of the executive after the change in control, which protects the merged company should it wish to maintain the services of the executive.

The change-in-control agreement provides for the executive to receive, in the event of the double trigger:

two times his/her existing base salary;

the higher of the last bonus paid or the target bonus;

up to twenty-four months of health insurance continuation through COBRA (which ceases should the executive accept a new position with health insurance coverage before the twenty-four month period ends);

outplacement assistance for nine months;

acceleration of vesting of all outstanding long-term incentive awards; and

a tax gross-up if IRC 280G excise tax penalty imposed for excess parachute payments. *Severance Plan* 

On February 20, 2006, we adopted a new executive officer severance plan to provide severance benefits to eligible executives whose employment is terminated involuntarily under certain circumstances. In 2007 we updated the executive officer severance plan to comply with IRS Regulation 409A. We did not make any other material changes to the executive officer severance plan in 2007

Any executive currently working for us at the executive officer level whose employment is terminated involuntarily is eligible for severance benefits, provided each of the following requirements is met:

the termination of employment is not for cause;

employment is terminated involuntarily;

the termination is not due to retirement, death or disability of the executive;

the executive is not a temporary employee or a new hire who has not yet started to work on a regular, full-time or part-time basis;

the executive is not covered under any other severance-type plan, policy, arrangement or agreement that provides severance payments and benefits more favorable than those in the plan;

the executive has not agreed in writing to waive severance benefits under the severance plan or otherwise payable from us;

the executive signs and does not revoke a Confidential Separation Agreement and General Release of All Claims in a form acceptable to us; and

the executive has returned all Invitrogen property and equipment. The following severance pay and benefits are payable under the severance plan:

twelve (12) months of base salary;

the executive s target incentive bonus under our ICP for the year in which the termination occurred, prorated to the date of termination;

nine months of outplacement assistance through a designated service provider to eligible executives; and

the monthly premiums required to continue an eligible executive s group health insurance coverage for a period of twelve (12) months.

Information regarding applicable payments under the change of control and severance arrangements for the named executive officers is provided under the heading Potential Payments Upon Termination or Change in Control below.

#### Impact of Performance on Compensation

The financial performance measure that we used in our 2007 Incentive Compensation Plan was operating income, which we believe is strongly related to the creation of total stockholder value. The Committee set our budgeted operating income figures as targets for the 2007 Incentive Compensation Plan. The Board of Directors approved the 2007 budget from which the targets were drawn.

We constructed the ICP with significant downside risk and upside potential. We needed to achieve 95% of the corporate operating income goal before there was any bonus payment at all. Because of the highly aggressive nature of the goal, we planned to pay as much as 200% of the target payout if we achieved over 115% of the operating income goal.

We set targets in 2007 that represented accelerated growth and required significant performance effort to achieve, and which represented a significant improvement over our disappointing 2006 financial performance. We exceeded our 2007 corporate operating income goal, and accordingly, we funded ICP bonuses at 130% of target. The particular ICP bonuses paid to each NEO for 2007 is included in the Summary Compensation Table, and reflects the judgment of the Committee (and in the case of the CEO, the Board of Directors) regarding the performance of each NEO in 2007.

In addition, for 2007 we did not increase base salaries for our CEO or any of the NEOs listed in our 2007 Proxy materials, which is consistent with our compensation philosophy of paying for performance in light of our 2006 financial results. In December 2007, the Committee approved

base salary increases for our NEOs listed in our 2008 Proxy materials, effective January 1, 2008. Specifically, on December 10, 2008, the Committee approved an increase of the Chief Financial Officer s base salary from \$440,000 to \$475,000, and an increase of the Senior Vice President, Research and Development s base salary from \$405,000 to \$425,000. In addition, on February 27, 2008, the Board of Directors approved an increase of the CEO s base salary from \$910,000 to \$975,000. These increases are consistent with our improved financial performance in 2007, and in line with our philosophy to pay competitively to the market.

#### Stock Ownership/Retention Guidelines

#### Stock Ownership Guidelines

The Committee has determined that each of our executive officers should own a significant amount of our Common Stock, relative to each officer s base salary in order to more closely align the interests of the executive officers with those of our stockholders. Officers are expected to attain these ownership levels within four years after their election or appointment to the specified officer position. The Committee expects the CEO to hold at least 43,000 shares of our Common Stock.

Mr. Lucier currently meets this requirement. The Committee expects our senior vice presidents to hold at least 8,000 to 16,000 shares of Common Stock, depending on their respective positions. As of February 28, 2008, all of our current executive officers were in compliance with these stock ownership guidelines. To measure compliance with the stock ownership guidelines, we include any outstanding shares of our Common Stock that the executive officer owns, as well as any restricted stock units or performance shares awarded to the executive, vested or unvested, which the officer holds. We do not include stock option awards for purposes of determining compliance with the stock ownership guidelines.

### Stock Retention Guidelines

Invitrogen does not have stock retention guidelines in addition to its stock ownership guidelines. The Committee believes that as long as executives are meeting and/or exceeding our stock ownership guidelines, there is no need for stock retention guidelines.

#### **Executive Officer Stock Ownership Summary**

Executive Officer	Shares Owned at Feb 15, 2008	Stock Ownership Guideline
Nicolas M. Barthelemy	23,417	10,000
Claude D. Benchimol, Ph.D.	20,956	10,000
Bernd Brust	10,972	10,000
John A. Cottingham	11,975	9,000
Paul Grossman, Ph.D.	15,000	8,000
David F. Hoffmeister	19,472	16,000
Peter M. Leddy, Ph.D.	28,904	8,000
Gregory T. Lucier	75,004	43,000
John L. Miller	21,375	10,000

Impact of Regulatory Requirements

#### Policy on Deductibility of Named Executive Officer Compensation

In evaluating compensation program alternatives, the Committee considers the potential impact of Section 162(m) of the Internal Revenue Code (IRC). Section 162(m) eliminates the deductibility of compensation over \$1 million paid to the NEOs, excluding performance-based compensation. Compensation programs generally will qualify as performance-based if (1) compensation is based on pre-established objective performance targets, (2) the programs material features have been approved by stockholders, and (3) there is no discretion to increase payments after the performance targets have been established for the performance period.

The Committee endeavors to maximize deductibility of compensation under Section 162(m) to the extent practicable while maintaining competitive, performance-based compensation. However, the Committee believes that it is important for it to retain maximum flexibility in designing compensation programs that meet its stated objectives and fit within the Committee s guiding principles. Further, the actual impact of the loss of deduction for compensation paid to the CEO and the other top four highly compensated executives over the limitation would be small and would have a minimal impact on our overall tax position. Therefore, the Committee will not limit compensation to those levels or types of compensation that will be deductible. The Committee will, of course, consider alternative forms of compensation that preserve deductibility, consistent with its compensation goals.

#### Internal Revenue Code Section 409A

IRC Section 409A relates to accounting treatment for deferred compensation. In 2007 the Committee reviewed all of our compensation plans and programs to ensure that they are compliant with IRC Section 409A, and has determined that they are or will be compliant by December 31, 2008. The proposed amendments to our 2004 EIP and 1998 ESPP include changes designed to comply with IRC Section 409A.

#### Impact of FAS 123R

FAS 123R requires companies to record option grants as expenses over the vesting period of the grant. Option expense is one factor that the Committee considers in the design of our long-term compensation programs. Other factors include:

the link to performance that each type of equity award provides;

the degree of upside leverage and downside risk inherent in each type of award;

the impact on dilution and overhang that the different equity awards have; and

the role that each type of equity award has in the attraction, retention, and motivation of our executive and key employee talent. The Committee monitors our FAS 123R expense to ensure that it is reasonable, although expense is not the most important factor in making decisions about our long-term incentive plans.

#### **REPORT OF THE**

#### COMPENSATION AND ORGANIZATIONAL DEVELOPMENT COMMITTEE

#### OF THE

### **BOARD OF DIRECTORS**

The Compensation and Organizational Development Committee reviewed and discussed the Compensation Discussion and Analysis required by Item 402(b) of SEC Regulation S-K with management. Based on such review and discussions, the Compensation and Organizational Development Committee recommended to the Board of Directors that the Compensation Discussion and Analysis be included in the registrant s Proxy Statement on Schedule 14A.

Ronald A. Matricaria (Chairman) W. Ann Reynolds, Ph.D. David C. U Prichard, Ph.D.

#### 2007 Summary Compensation Table

The following table sets forth information for the fiscal year ended December 31, 2007, concerning the compensation of the Chief Executive Officer and Chief Financial Officer of the Company and each of the four other most highly compensated executive officers as of December 31, 2007, whose total salary and bonus for the year ended December 31, 2007, exceeded \$100,000 for services rendered in all capacities to the Company.

(a) Name and Principal Position	(b) Year	(c) Salary (\$)	(d) Bonus (\$)	(e) Stock Awards (\$)	(f) Option Awards (\$)	(g) Non-Equity Incentive Plan Compensation (\$) <sup>(1)</sup>	(h) Change in Pension Value and Non-Qualified Deferred Comp Earnings	(i) All Other Compensation (\$) <sup>(2)</sup>	(j) Total (\$)
Gregory T. Lucier		910.000		5.037.483	()		0	45,747(3)	11,068,812
Chairman and CEO	2006	904,615	0	1,328,832	2,861,864		0	20,493(4)	5,403,478
Nicolas M. Barthelemy	2007	358,461	0	517,777	539,702	293,580	103,361	31,888(5)	1,844,769
SVP, Cell Systems	2006	320,000	0	378,813	419,474	0	147,216	31,413(6)	1,150,455
Claude D. Benchimol, Ph.D.	2007	405,000	0	613,048	932,047	280,665	333,792	34,011(7)	2,598,563
SVP, R&D	2006	403,461	0	652,976	641,108	0	0	22,007(8)	1,719,552
David F. Hoffmeister	2007	440,000	225,000	1,144,439	992,620	332,640	0	24,847(9)	3,159,546
SVP and CFO	2006	439,423	225,000	1,184,542	875,001	119,674	0	9,420(10)	2,853,060
Peter M. Leddy. Ph.D.	2007	358,846	0	768,772	695,051	293,895	0	56,352(11)	2,172,916
SVP, Human Resources	2006	330,000	100,000	637,379	350,891	0	0	65,929(12)	1,484,199

- (1) 2007 Figures consist of cash portion of the 2007 ICP payment, paid March 14, 2008. Each NEO was eligible to elect to receive a portion of the 2007 ICP in restricted stock units, as described in the section entitled Compensation Discussion & Analysis. Any restricted stock units awarded as payment pursuant to the 2007 ICP will be reflected in column (e) in future Summary Compensation Tables as the cost of the restricted stock units are amortized.
- (2) Consists of any Executive financial planning services, fringe benefits, 401(k) matching program, supplemental life insurance and any loan subsidy.
- (3) Consists of Executive financial planning services of \$10,000, 401(k) matching of \$6,750, supplemental life insurance premium payments of \$15,513, executive physical of \$2,500 and health insurance contribution of \$10,984.
- (4) Consists of Executive financial planning services of \$12,033, 401(k) matching of \$6,600 and supplemental life insurance premium payments of \$1,860.
- (5) Consists of Executive financial planning services of \$7,500, 401(k) matching of \$6,750, and supplemental life insurance premium payments of \$3,743, executive physical of \$2,500 and health insurance contribution of \$11,395.
- (6) Consists of Executive financial planning services of \$7,500, 401(k) matching of \$6,600, supplemental life insurance premium payments of \$3,656, executive physical of \$2,500 and health insurance contribution of \$11,157.
- (7) Consists of Executive financial planning services of \$7,500, 401(k) matching of \$2,761, supplemental life insurance premium payments of \$9,855, executive physical of \$2,500 and health insurance contribution of \$11,395.

- (8) Consists of Executive financial planning services of \$11,533, 401(k) matching of \$6,600 and supplemental life insurance premium payments of \$3,874.
- (9) Consists of Executive financial planning services of \$7,500, supplemental life insurance premium payments of \$7,352, executive physical of \$2,500 and health insurance contribution of \$7,495.
- (10) Consists of Executive financial planning services of \$7,133, and supplemental life insurance premium payments of \$2,287.
- (11) Consists of Executive financial planning services of \$7,500, 401(k) matching of \$5,651, supplemental life insurance premium payments of \$2,990, executive physical of \$2,500, health insurance contribution of \$10,984 and housing loan of \$26,726.
- (12) Consists of Executive financial planning services of \$7,500, 401(k) matching of \$6,600, supplemental life insurance premium payments of \$2,904, executive physical of \$2,500, health insurance contribution of \$11,157 and housing loan of \$35,268.
  - 23

#### Grants of Plan-Based Awards Table

The following table sets forth certain information with respect to stock and option awards and other plan-based awards granted to our named executive officers during the fiscal year ended December 31, 2007.

Name	Grant Date	Under No Pl	n-Equity an Awar		Under Eq	uity Ince Awards		All Other Stock Awards: Number of Shares of Stock or Units (#)	All Other Option Awards: Number of Securities Underlying Options (#)	Exercise of Base Price of Option Awards (\$/Sh)	Grant Date Fair Value of Stock and Option Awards (\$)
(a)	<b>(b</b> )	Threshold (\$) (c)	(\$) (d)	Maximum (\$) (e)	Threshold (\$) (f)	Target (\$) (g)	Maximum (\$) (h)	(i)	(j)	(k)	(1)
Gregory T. Lucier	()	(+) (-)	(+) ()	(+) (-)	(+) (-)	(+) (8)	(+) ()	(-)	<b>(</b> )/	()	(-)
Chairman and CEO	03/01/07	0	0	0	0	0	0	404,761(1)	0	0	26,013,989
Nicolas M. Barthelemy SVP, Cell Systems	03/01/07							1,326	0	0	85,222
	05/15/07	0	0	0	0	0	0	5,300	26,000	71.74	980,845
Claude D. Benchimol, Ph.D. SVP, R&D	05/15/07	0	0	0	0	0	0	5,300	24,000	71.74	934,644
David F. Hoffmeister SVP and CFO	03/01/07							1,930	0	0	124,041
	05/15/07	0	0	0	0	0	0	5,300	24,000	71.74	934,644
Peter M. Leddy, Ph.D. SVP, Human Resources	03/01/07							603	0	0	38,755
	05/15/07	0	0	0	0	0	0	5,300	26,000	71.74	980,845

(1) Consists of 400,000 performance-based restricted stock units grant to Mr. Lucier at March 1, 2007.

#### **Options Exercised and Stock Vested Table**

The following information sets forth the stock awards vested and stock options exercised by the Named Executive Officers during the fiscal year ended December 31, 2007.

	<b>Option Awards</b>		Stock Number	Awards
Name	Number of Shares Exercised	Value Realized on Exercise <sup>(1)</sup>	of Shares Vesting	Value Realized on Vesting <sup>(2)</sup>
Gregory T. Lucier				
Chairman and CEO	421,324	\$ 22,429,284	53,100	\$ 3,945,102
Nicolas M. Barthelemy				
SVP, Cell Systems	62,000	\$ 1,182,580	1,900	\$ 172,330
Claude D. Benchimol, Ph.D.				
SVP, Research & Development	100,000	\$ 3,855,261	16,900	\$ 1,335,718
David F. Hoffmeister				
SVP and CFO	N/A	N/A	50,000	\$4,151,500
Peter M. Leddy, Ph.D.				
SVP, Human Resources	N/A	N/A	N/A	N/A

(1) Represents the excess of the fair market value of the shares exercised over the aggregate price of such shares on the date of exercise.

### (2) Represents the fair market value of the shares on the date of vesting.

#### **Outstanding Equity Awards at Fiscal Year-end Table**

The following table sets forth certain information with respect to the value of all unexercised options and unvested stock awards previously awarded to our named executive officers as of December 31, 2007.

Name (a)	Number of Securities Underlying Unexercised Options (#) Exercisable (b)	Number of Securities Underlying Unexercised Options (#) Unexercisable (c)	Equity Incentive Plan Awards: Number of Securities Underlying Unexercised Unearned Options (#) (d)	Option Exercise Price (\$) (e)	Option Expiration Date (f)	Number of Shares or Units of Stock That Have Not Vested (g)	Market Value of Shares or Units of Stock That Have Not Vested (\$) (h)	Equity Incentive Plan Awards: Number of unearned Shares, Units or Other Rights Not Vested (i)	Equity Incentive Plan Awards: Market or Payout Value of Unearned Shares, Units or Other Rights That Have Not Vested (\$) (j)
Gregory T. Lucier	4,587	1,529	0	65.38	5/14/2014	0	0	0	0
Chairman and CEO	0	1,339	0	74.66	3/1/2016	0	0	0	0
	0	1,301	0	76.86	5/13/2015	0	0	0	0
	34,788	11,596	0	65.38	5/14/2014	0	0	0	0
	21,250	19,949	0	76.86	5/13/2015	6,000	461,160	0	0
	26,250	77,411	0	74.66	3/1/2016	0	0	0	0
	253,676	0	0	38.01	5/30/2013	0	0	0	0
	21,250	21,250	0	64.51	11/14/2015	6,000	387,060	0	0
	26,250	8,750	0	62.51	11/12/2014	0	0	0	0
	0	0	0	0	3/1/2017	4,761	305,989	0	0
	0	0	0	0	3/1/2010	400,000	25,708,000	0	0

	388,051	143,125	0			416,761	26,862,209	0	0
Nicolas M.	0	1,301	0	76.86	5/13/2015	0	0	0	0
Barthelemy	0	1,518	0	65.87	5/12/2016	0	0	0	0
SVP, Cell Systems	0	1,421	0	70.34	3/10/2014	0	0	0	0
	0	3,250	0	62.51	11/12/2014	0	0	0	0
	2,500	3,699	0	76.86	5/13/2015	2,500	192,150	0	0
	0	11,079	0	70.34	3/10/2014	0	0	0	0
	0	26,000	0	71.74	5/15/2017	5,300	380,222	0	0
	0	20,232	0	65.87	5/12/2016	3,600	237,132	0	0
	0	5,000	0	64.51	11/14/2015	2,500	161,275	0	0
	0	0	0	0	3/1/2017	1,326	2,750,435	0	0
	2,500	73,500	0			15,226	3,721,214	0	0

Name (a)	Number of Securities Underlying Unexercised Options (#) Exercisable (b)	Number of Securities Underlying Unexercised Options (#) Unexercisable (c)	Equity Incentive Plan Awards: Number of Securities Underlying Unexercised Unearned Options (#) (d)	Option Exercise Price (\$) (e)	Option Expiration Date (f)	Number of Shares or Units of Stock That Have Not Vested (g)	Market Value of Shares or Units of Stock That Have Not Vested (\$) (h)	Equity Incentive Plan Awards: Number of unearned Shares, Units or Other Rights Not Vested (i)	Equity Incentive Plan Awards: Market or Payout Value of Unearned Shares, Units or Other Rights That Have Not Vested (\$) (j)
Claude D. Benchimol, Ph.D.	0	1,529	0	65.38	5/14/2014	0	0	0	0
SVP, R&D	0	1,518	0	65.87	5/12/2016	0	0	0	0
	0	1,301	0	76.86	5/13/2015	0	0	0	0
	11,250	2,221	0	65.38	5/14/2014	0	0	0	0
	6,250	4,949	0	76.86	5/13/2015	3,000	230,580	0	0
	7,250	20,232	0	65.87	5/12/2016	3,600	237,132	0	0
	0	24,000	0	71.74		5,300	380,222	0	0 0
	6,250	6,250	0		11/14/2015	3,000	193,530	0	0
	9,750	3,250	0		11/12/2014	0	0		0
	2,720	3,230	0	02.01	11/12/2011	0	0	0	0
	40,750	65,250	0			14 000	1,041,464	0	0
	40,750	05,250	U			14,700	1,041,404	0	U
David F. Hoffmeister	5,454	1,818	0	55.00	10/13/2014	0	0	0	0
SVP and CFO	5,454 0	1,818	0	65.87	5/12/2014	0	0		0
SVF and CFO	0	· · · · · · · · · · · · · · · · · · ·	0			0	0	0	0
	0	1,301	0	76.86 71.74				0	0
		24,000	0	65.87	5/15/2017	5,300	380,222	0	0
	7,250	20,232			5/12/2016	4,000	263,480		
	7,500	7,500	0		11/14/2015	4,500	290,295	0	0
	7,500	6,199	0	76.86		4,500	345,870	0	0
	144,546	48,182	0		10/13/2014	0	0	0	0
	0	0	0	0	3/1/2017	1,930	124,041	0	0
	172,250	110,750	0			20,230	1,403,908	0	0
Peter M. Leddy, Ph.D.	2,354	2,354	0	84.90	7/5/2015	20,000	1.698.000	0	0
SVP, Human Resources	2,354	2,334	0		11/14/2015	20,000	1,098,000	0	0
S VI, Human Resources	0		0	65.87	5/12/2016	3.600	237,132	0	0
	22,646	22,646	0	84.90	7/5/2015	0,000	237,132	0	0
	16,250	48,750	0	63.41	9/29/2015	0	0	0	0
	10,230		0	71.74		5,300	380,222	0	0
	4,498	4,498	0		11/14/2015	1,000	64,510	0	0
	4,498	4,498	0	65.87		1,000	04,310	0	0
	6,500 0	17,982	0	05.87	5/12/2016 3/1/2017	603	38,755	0	0
	0	0	0	0	5/1/2017	003	30,733	0	0
	52,250	123,750	0			30,503	2,418,619	0	0

#### **Employment and Severance Arrangements**

#### **Employment Agreements**

The Company entered into an Employment Agreement with its current Chairman and Chief Executive Officer effective May 30, 2003. Under the terms of this agreement, upon termination of employment he could receive a payment totaling 1.5 times his annual salary plus 1.5 times an imputed bonus of 75% of his annual salary, and any Long Term Incentive Payments that he would have received within 18 months of the date of termination, which amounts to \$250,000, payable in March of each year for employment during the prior calendar year. In addition, he could receive continuing health and welfare benefits for eighteen months. The executive would be eligible for these payments and benefits upon his separation from the Company under specified circumstances other than termination for cause. The Employment Agreement was filed as Exhibit 10.57 to the Company s Quarterly Report on Form 10-Q for the period ended June 30, 2003, filed with the SEC on August 13, 2003.

The Company entered into an Employment Agreement effective on October 13, 2004, with David F. Hoffmeister for Mr. Hoffmeister to serve as the Company s Chief Financial Officer, Senior Vice President, Finance. Under the Agreement, Mr. Hoffmeister is guaranteed to receive his target bonus under the Incentive Compensation Plan for his first year of employment. Mr. Hoffmeister received a one time signing bonus of \$375,000, refundable to the Company on a pro rata basis in the event of his voluntary termination or termination for cause during the first two years of employment. The Agreement also provides for a \$225,000 employment bonus payable on or before each of the first three anniversary dates of Mr. Hoffmeister s initial employment provided he is still employed by the Company on those dates or his employment was terminated for certain reasons. The Agreement also provides Mr. Hoffmeister with severance benefits in the event of his termination for certain reasons.

We have entered into letter agreements with each of our other executive officers outlining the terms of their employment and the elements of their compensation. Each of these letter agreements follows our standard employment offer template, and provides for employment at will.

#### Compensation of Directors

Effective April 20, 2007, the Board of Directors adopted new annual compensation guidelines as follows. Each Director receives a fixed annual compensation of \$250,000 with 30% payable in cash, and 70% payable in restricted stock units. Cash payments are made in advance at the start of each calendar quarter, and the Board, at its first meeting following the Annual Meeting of stockholders, determines the amount of each cash payment for the subsequent four quarters. The Presiding Director and each Committee Chairman receive an additional \$10,000 per year.

Restricted stock units (RSUs) are granted at the first Board meeting following the Annual Meeting. The Board anticipates that members of the Board will receive RSUs with a Fair Market Value on the date of grant of \$175,000 for each year. Each RSU grant completely vests at the earlier of the anniversary of its grant date, or the date of the next Annual Meeting. The holding period for RSUs is a minimum of three years. Each Director may elect to have the company issue his or her RSUs at a specified time after three years, and if no election is made, the RSUs will be issued at termination of such Director s service. RSUs are taxed when they are issued.

Cash and equity compensation for newly appointed directors are pro-rated to the date of the next annual meeting.

#### Compensation Committee Interlocks and Insider Participation

There were no interlocks or other relationships among the Company s executive officers and directors that are required to be disclosed under applicable executive compensation disclosure requirements.

#### **Director Compensation Table**

Name (a)	Fees Earned Paid in Cash (\$) (b)	Stock Awards (\$) <sup>(1)</sup> (c)	Option Awards (\$) <sup>(1)</sup> (d)	Non-Equity Incentive Plan Compensation (\$) (e)	Change in Pension Value and Nonqualified Deferred Compensation Earnings (f)	All Other Compensation (\$) (g)	Total (\$) (h)
Raymond V. Dittamore	82,208	142,329	25,153	0	0	0	249,690
Donald W. Grimm	82,208	142,329	25,153	0	0	0	249,690
Balakrishnan S. Iyer	78,424	142,329	25,153	0	0	0	245,906
Bradley G. Lorimier	75,000	142,329	25,153	0	0	0	242,482
Ronald A. Matricaria	82,208	142,329	25,153	0	0	0	249,690
Per A. Peterson, Ph.D.	61,181	129,765	16,939	0	0	0	207,885
W. Ann Reynolds, Ph.D.	71,424	142,329	25,153	0	0	0	238,906
Jay M. Short, Ph.D.	71,424	142,329	25,153	0	0	0	238,906
David C. U Prichard, Ph.D.	82,208	142,329	25,153	0	0	0	249,690

(1) Prior to April 20, 2007, the Board of Directors received equity compensation of 7,000 stock options and 1,000 restricted stock units each year. On April 24, 2006 each director received a grant of an option to purchase 7,000 shares of our Common Stock and 1,000 restricted stock units. The per share exercise price of the options was \$63.72. These stock options and restricted stock units vested over the course of one year, and became fully vested in April 2007. In addition, the Board of Directors received 2,614 restricted stock units on April 19, 2007. The amount recorded in these columns reflects the expense of these equity grants recorded in 2007.

(2) Dr. Peterson joined the Board effective March 1, 2007.

The aggregate number of stock awards and stock option awards for each director is included in the information set forth with respect to each director in the section entitled Stock Ownership.

#### **Director Stock Ownership Guidelines Table**

In February 2008, the Board of Directors adopted stock ownership guidelines for the directors and required each director to meet the guidelines within the time period set forth below. The chart below indicates each director s progress toward compliance.

Name

(a)	Shares Owned <sup>(1)</sup> (b)	Ownership Requirement (c)	Deadline for Meeting Ownership Requirement
	. ,	.,	
Raymond V. Dittamore	7,220	10,000	2010
Donald W. Grimm	8,614	10,000	2010
Balakrishnan S. Iyer	6,614	10,000	2010
Bradley G. Lorimier	8,714	10,000	2010
Ronald A. Matricaria	14,614	10,000	2010
W. Ann Reynolds, Ph.D.	7,614	10,000	2010
Jay M. Short, Ph.D.	8,114	10,000	2010
David C. U Prichard, Ph.D.	7,714	10,000	2010
Per A. Peterson, Ph.D.	2,751	10,000	2012

(1) Consists of Direct Stock Ownership, Restricted Stock Units and Deferred Stock Units, as applicable. *Potential Payments upon Termination or Change in Control* 

We have entered into certain agreements and maintain certain plans that will require us to provide compensation to named executive officers of Invitrogen in the event of a termination of employment or a change in control of Invitrogen. The amount of compensation payable to each named executive officer in each situation is set forth in the tables below.

The following table describes the potential payments upon termination or a change in control of Invitrogen for Mr. Lucier, Invitrogen s CEO:

Executive Benefits and Payments Upon Termination <sup>(1)</sup>	Voluntary Termination For Good Reason	Involuntary mination other than for Cause <sup>(2)</sup>	Termination for Cause	Foll	<sup>°</sup> ermination owing Change n Control <sup>(3)</sup>
Compensation					
Base salary		\$ 3,412,500		\$	4,550,000
Non-equity Incentive Plan					
Long-term incentives <sup>(4)</sup>	\$ 25,435,463	\$ 25,435,463			
Option acceleration				\$	28,136,825
Restricted stock acceleration				\$	17,177,328
Benefits and Perquisites					
Health care insurance					

Executive Benefits and Payments Upon Termination <sup>(1)</sup>	Voluntary Termination For Good Reason	nvoluntary mination other than for Cause <sup>(2)</sup>	Termination for Cause	Foll	Fermination lowing Change n Control <sup>(3)</sup>
Benefit Continuation		\$ 21,008		\$	28,011
Accrued Vacation					
Outplacement Assistance		\$ 10,000		\$	25,000
280G gross-up				\$	22,473,415
Vesting of Employer 401(k) Contributions					
Total:	\$ 25,435,463	\$ 28,878,971		\$	72,390,579

- (1) For purposes of this analysis, we assumed the executive s compensation is as follows: current base salary equal to \$910,000, annual incentive opportunity equal to 150% of base salary
- (2) Assumes the executive s severance benefit under an involuntary termination other than for cause is equal to 1.5 times base salary and target annual bonus.
- (3) Based on involuntary termination or termination for good reason. Assumes the executive s severance benefit is equal to 1.5 times base salary and target annual bonus.
- (4) Assumes the executive s date of termination is December 31, 2007 (assuming a calendar fiscal year-end) and the price per share of the Company s stock on the date of termination is \$93.41 per share.

The following table describes the potential payments upon termination or a change in control of Invitrogen for Mr. Barthelemy, Invitrogen s SVP, Cell Systems:

Executive Benefits and Payments Upon Termination <sup>(1)</sup>	Tern	oluntary nination For od Reason	Involuntary Termination other than for Cause <sup>(2)</sup>		Termination for Cause	Foll	ermination owing Change 1 Control <sup>(3)</sup>
Compensation							
Base salary			\$	360,000		\$	1,152,000
Non-equity Incentive Plan							
Long-term incentives <sup>(4)</sup>	\$	453,171	\$	906,971			
Option acceleration						\$	2,036,861
Restricted stock acceleration						\$	1,835,712
Short-term incentives			\$	216,000			
Deferred compensation balance	\$	247,608	\$	247,608		\$	247,608
Benefits and Perquisites							
Health care insurance							
Benefit Continuation			\$	11,395		\$	29,190
Retirement Plan Benefits						\$	19,384
Accrued Vacation							



Executive Benefits and Payments Upon Termination <sup>(1)</sup>	Voluntary Termination For Good Reason	Involuntary Termination other than for Termination for Cause <sup>(2)</sup> Cause		Termination Following Change in Control <sup>(3)</sup>	
Outplacement Assistance		\$ 10,000		\$ 25,000	
280G gross-up				\$ 1,114,420	
Vesting of Employer 401(k) Contributions					
Total:	\$ 700,779	\$ 1,751,974		\$ 6,460,175	

- (1) For purposes of this analysis, we assumed the executive s compensation is as follows: current base salary equal to \$360,000, annual incentive opportunity equal to 50% of base salary.
- (2) Assumes the executive s severance benefit under an involuntary termination other than for cause is equal to one times base salary, target annual bonus.
- (3) Based on involuntary termination or termination for good reason. Assumes the executive s severance benefit is equal to one times base salary and target annual bonus.
- (4) Assumes the executive s date of termination is December 31, 2007 (assuming a calendar fiscal year-end) and the price per share of the Company s stock on the date of termination is \$93.41 per share.

The following table describes the potential payments upon termination or a change in control of Invitrogen for Dr. Benchimol, Invitrogen s SVP, R&D:

Executive Benefits and Payments Upon Termination <sup>(1)</sup>	Ter	Voluntary mination For ood Reason	Involuntary Termination other than for Cause <sup>(2)</sup>		Termination for Cause	Termination Following Chang in Control <sup>(3)</sup>	
Compensation							
Base salary			\$	405,000		\$	1,296,000
Non-equity Incentive Plan							
Long-term incentives <sup>(4)</sup>	\$	4,192,353	\$	4,192,353			
Option acceleration						\$	5,628,115
Restricted stock acceleration						\$	1,352,324
Short-term incentives			\$	243,000			
Deferred compensation balance	\$	287,061	\$	287,061		\$	287,061
Benefits and Perquisites							
Health care insurance							
Benefit Continuation			\$	11,395		\$	29,190
Retirement Plan Benefits						\$	12,397
Accrued Vacation							
Outplacement Assistance			\$	10,000		\$	25,000
280G gross-up							
Vesting of Employer 401(k) Contributions							
Total:	\$	4,479,414	\$	5,148,809		\$	8,630,087

(1) For purposes of this analysis, we assumed the executive s compensation is as follows: current base salary equal to \$405,000, annual incentive opportunity equal to 60% of base salary.

- (2) Assumes the executive s severance benefit under an involuntary termination other than for cause is equal to one times base salary, target annual bonus.
- (3) Based on involuntary termination or termination for good reason. Assumes the executive s severance benefit is equal to one times base salary and target annual bonus.
- (4) Assumes the executive s date of termination is December 31, 2007 (assuming a calendar fiscal year-end) and the price per share of the Company s stock on the date of termination is \$93.41 per share.

The following table describes the potential payments upon termination or a change in control of Invitrogen for Mr. Hoffmeister, Invitrogen s CFO:

Executive Benefits and Payments Upon Termination <sup>(1)</sup>	Tern	oluntary nination For od Reason	Involuntary Termination other than for Cause <sup>(2)</sup>		Termination for Cause	Foll	Termination Following Change in Control <sup>(3)</sup>	
Compensation								
Base salary			\$	440,000		\$	1,408,000	
Non-equity Incentive Plan								
Long-term incentives <sup>(4)</sup>	\$	5,845,578	\$	7,633,578				
Option acceleration						\$	8,932,540	
Restricted stock acceleration						\$	1,836,075	
Benefits and Perquisites								
Health care insurance								
Benefit Continuation			\$	14,334		\$	19,113	
Outplacement Assistance			\$	10,000		\$	25,000	
280G gross-up								
Vesting of Employer 401(k) Contributions								
Total:	\$	5,845,578	\$	8,097,912		\$	12,220,728	

- (1) For purposes of this analysis, we assumed the executive s compensation is as follows: current base salary equal to \$440,000, annual incentive opportunity equal to 60% of base salary.
- (2) Assumes the executive s severance benefit under an involuntary termination other than for cause is equal to 1.5 times base salary, target annual bonus.
- (3) Based on involuntary termination or termination for good reason. Assumes the executive s severance benefit is equal to 1.5 times base salary and target annual bonus.
- (4) Assumes the executive s date of termination is December 31, 2007 (assuming a calendar fiscal year-end) and the price per share of the Company s stock on the date of termination is \$93.41 per share.

The following table describes the potential payments upon termination or a change in control of Invitrogen for Dr. Leddy, Invitrogen s SVP, Human Resources:

Executive Benefits and Payments Upon Termination <sup>(1)</sup>	Tern	Involuntary Voluntary Termination other Fermination For than for Good Reason Cause <sup>(2)</sup>		Termination for Cause	Termination Following Change in Control <sup>(3)</sup>		
Compensation							
Base salary			\$	360,000		\$	1,152,000
Non-equity Incentive Plan							
Long-term incentives <sup>(4)</sup>	\$	870,848	\$	870,848			
Option acceleration						\$	3,448,660
Restricted stock acceleration						\$	2,768,452
Long-term incentives			\$	216,000			
Benefits and Perquisites							
Retirement Plan Benefits						\$	19,383
Benefit Continuation			\$	10,984		\$	28,011
Accrued Vacation							
Outplacement Assistance			\$	10,000		\$	25,000
280G gross-up						\$	1,272,684
Vesting of Employer 401(k) Contributions							
Total:	\$	870,848	\$	1,467,832		\$	8,714,190

- (1) For purposes of this analysis, we assumed the executive s compensation is as follows: current base salary equal to \$360,000, annual incentive opportunity equal to 50% of base salary.
- (2) Assumes the executive s severance benefit under an involuntary termination other than for cause is equal to one times base salary, target annual bonus.
- (3) Based on involuntary termination or termination for good reason. Assumes the executive s severance benefit is equal to one times base salary and target annual bonus.
- (4) Assumes the executive s date of termination is December 31, 2007 (assuming a calendar fiscal year-end) and the price per share of the Company s stock on the date of termination is \$93.41 per share.
   Nonqualified Deferred Compensation Table

Name of Executive	Type of Deferred Compensation Plan	Executive Contributions in Last Fiscal Year	Registrant Contributions in Last Fiscal Year	Aggregate Earnings in the last Fiscal Year	Aggregate Balance at Last Fiscal Year End
Gregory T. Lucier	Unfunded non-qualified	0	0	0	0
Nicolas M. Barthelemy	Unfunded non-qualified	\$ 75,553.83	0	\$ 16,939.13	\$ 241,940.06
Claude D. Benchimol, Ph.D.	Unfunded non-qualified	\$ 303,749.94	0	\$ 20,626.56	\$ 318,386.84

Name of Executive	Type of Deferred Compensation Plan	Executive Contributions in Last Fiscal Year	Registrant Contributions in Last Fiscal Year	Aggregate Earnings in the last Fiscal Year	Aggregate Balance at Last Fiscal Year End
David F. Hoffmeister	Unfunded non-qualified	0	0	0	0
Peter M. Leddy, Ph.D.	Unfunded non-qualified GOVERNANCE AND NOMINA	0 TING COMMITI	) YEE	0	0

#### Who are the nominees for election at the 2008 Annual Meeting of Stockholders?

The Governance and Nominating Committee recommended Balakrishnan S. Iyer, Ronald A. Matricaria, and W. Ann Reynolds, Ph.D., to be nominated by the Board of Directors for election to Class III of the Board of Directors at the Annual Meeting of Stockholders.

In selecting non-incumbent candidates and reviewing the qualifications of incumbent candidates for the Board of Directors, the Governance and Nominating Committee considers the Company s corporate governance principles, which include the following:

Directors should possess the highest personal and professional ethics, integrity and values, and be committed to representing the long-term interests of the stockholders. They must also have an inquisitive and objective perspective, practical wisdom and mature judgment. They must be actively engaged in the pursuit of information relevant to the Company s business and must constructively engage their fellow Board members, the CEO, and other members of management in dialogue and decision making. The Board of Directors will represent diverse experience at policy-making levels in business and technology in areas that are relevant to the Company s global activities.

Directors must be willing to devote sufficient time to carrying out their duties and responsibilities effectively, and should be committed to serve on the Board of Directors for an extended period of time. Directors should offer their resignation in the event of any significant change in their personal circumstances, including a change in their principal job responsibilities.

A supermajority of at least 2/3 of the directors will be independent directors as defined in the NASD rules for companies listed on the Nasdaq National Market. Directors who do not meet the NASD Manual s independence standards also make valuable contributions to the Board of Directors and to the Company through their experience and wisdom.

In general, to be considered independent under the NASD Manual rules, the Board must determine, among other things, that a director does not have any relationships that, in the Board s opinion, would interfere with the exercise of independent judgment in carrying out the responsibilities of a director. The Board will make an affirmative finding with respect to the independence of directors not less frequently than annually. The Board has determined that other than Mr. Lucier, the Company s CEO, each of the current members of the Board, including the nominees for Class III director, is an independent director.

In addition to the policy that a supermajority of the Board members satisfy the independence standards discussed in the section above, members of the Audit Committee must also satisfy additional NASD independence requirements. Specifically, they may not directly or indirectly receive any compensation from the Company other than their directors compensation, must not have participated in preparing the financial statements of the Company or any of its subsidiaries during the past three years, and must not be affiliated with the Company except through their membership on the Board and its committees.

#### CERTAIN RELATIONSHIPS AND RELATED PARTY TRANSACTIONS

#### **Procedures for Approval of Related Party Transactions**

Pursuant to the Invitrogen Protocol and the Audit Committee Charter, our executive officers, directors and principal stockholders, including their immediate family members and affiliates, are prohibited from entering into a related party transaction with us without the consent of our Audit Committee (or other independent committee of the Board of Directors in cases where it is inappropriate for the Audit Committee to review such transaction due to a conflict of interest). Any request for us to enter into a transaction with an executive officer, director, principal stockholder or any of such persons immediate family members or affiliates in which the amount involved exceeds \$120,000 must be presented to our Audit Committee for review, consideration and approval. In approving or rejecting the proposed transaction, our Audit Committee will consider the

relevant facts and circumstances available and deemed relevant, including but not limited to, the risks, costs, and benefits

to the Company, the terms of the transactions, the availability of other sources for comparable services or products, and, if applicable, the impact on director independence. Our Audit Committee shall only approve those transactions that, in light of known circumstances, are in or are not inconsistent with, our best interests, as determined in good faith by the Audit Committee.

#### Speaking Engagement

The Company has entered into an agreement for a speaker to address a dinner meeting during the Company s participation in BIO, an industry convention in San Diego, California in the summer of 2008. The Speaker has requested that in lieu of a speaking fee, the Company make a donation to the E.O. Wilson Biodiversity Foundation. Our CEO is a member of the Board of Trustees of the foundation, as is the speaker. Jay M. Short, Ph.D., is the Chairman of the E.O. Wilson Biodiversity Foundation and a member of the Board of Directors.

#### Indemnification Agreements

The Company has entered into indemnification agreements with each of its executive officers and directors containing provisions that may require the Company, among other things, to indemnify those officers and directors against liabilities that may arise by reasons of their status or service as officers or directors. The agreements also provide for the Company to advance to the officers and directors expenses that they expect to incur as a result of any proceeding against them as to which they could be indemnified. The Company also intends to execute such agreements with its future directors and executive officers.

### SECTION 16(a) BENEFICIAL OWNERSHIP REPORTING COMPLIANCE

Section 16(a) of the Securities Exchange Act of 1934, as amended, requires our executive officers, directors and persons who beneficially own more than 10% of the Company s Common Stock to file initial reports of ownership and reports of changes in ownership with the Securities and Exchange Commission (SEC). SEC regulations require these individuals to give us copies of all Section 16(a) forms they file.

Based solely on our review of forms that were furnished to us and written representations from reporting persons, we believe that the executive officers, directors and more than 10% stockholders complied with all filing requirements related to Section 16(a), except for the following:

Claude D. Benchimol, Ph.D. vested 12,500 shares of Restricted Stock in September 2007. Dr. Benchimol elected to have the Company withhold shares in lieu of taxes. In addition, Dr. Benchimol donated 500 shares of Invitrogen to a charity in December 2007. The Company filed a Form 4 on behalf of Dr. Benchimol, however, the filing was not within two business days.

### THE INVITROGEN PROTOCOL

The Company has adopted a code of ethics applicable to all of its employees, including the principal executive officer, principal financial officer, principal accounting officer, its controller, and all of its directors. The code of ethics is called the Invitrogen Protocol, and a copy is posted to our internet site at <u>http://www.invitrogen.com</u>.

### ITEMS FOR STOCKHOLDER CONSIDERATION

#### PROPOSAL 1

#### **Election of Directors**

The Company has a classified Board of Directors currently consisting of three Class I directors (Gregory T. Lucier, Donald W. Grimm, and Per A. Peterson, Ph.D.) who will serve until the Annual Meetings of stockholders to be held in 2009, three Class II directors (Bradley G. Lorimier, Raymond V. Dittamore, and David C. U Prichard, Ph.D.) who will serve until the Annual Meetings of stockholders to be held in 2010, and four Class III directors (Balakrishnan S. Iyer, Ronald A. Matricaria, W. Ann Reynolds, Ph.D., and Jay M. Short, Ph.D) who will serve until the Annual Meeting, and in each case until their respective successors are duly elected and qualified. Directors in a class are elected for a term of three years to succeed the directors in such class whose terms expire at such annual meeting, or a shorter term to fill a vacancy in another class of directors.

Jay M. Short, Ph.D., has announced his retirement from the Board of Directors, effective at the time of the Annual Meeting. On February 27, 2008, pursuant to Section 2.1 of the Bylaws of the Company, the Board of Directors reduced the number of directors from ten to nine, effective upon Dr. Short s resignation at the Annual Meeting. Accordingly, the nominees to fill the three Class III positions on the Board of Directors are Balakrishnan S. Iyer, Ronald A. Matricaria, and W. Ann Reynolds, Ph.D. If elected, the nominees for the Class III positions will serve as directors until the annual meeting of stockholders in 2011, and in each case until their successors are elected and qualified.

If a nominee declines to serve or becomes unavailable for any reason, or if a vacancy occurs before the election, the proxies may be voted for such substitute nominee as the proxy holders may designate.

#### Vote Required and Board of Directors Recommendation

If a quorum is present, either in person or by proxy, the three nominees for Class III who receive the greatest number of votes cast will be elected as Class III directors. Abstentions and broker non-votes will be counted for purposes of determining the presence or absence of a quorum. Neither abstentions nor broker non-votes will have any effect upon the outcome of voting with respect to the election of directors.

#### THE BOARD OF DIRECTORS RECOMMENDS A VOTE FOR

#### THE NOMINEES NAMED ABOVE.

#### PROPOSAL 2

#### **Ratification of Appointment of Independent Auditors**

The Audit Committee of the Board of Directors has selected Ernst & Young LLP as the independent auditors to audit our financial statements for the fiscal year ended December 31, 2008. Representatives of Ernst & Young LLP are expected to be present at the Annual Meeting of stockholders with the opportunity to make a statement if the representatives desire to do so, and are expected to be available to respond to appropriate questions.

#### Vote Required and Board of Directors Recommendation

The affirmative vote of the holders of a majority of the shares of Common Stock cast at the Annual Meeting is required for ratification of this selection. Abstentions and broker non-votes will be counted for purposes of determining the presence or absence of a quorum. Neither abstentions nor broker non-votes will have any effect upon the outcome of voting with respect to the ratification of independent public accountants.

#### THE BOARD OF DIRECTORS RECOMMENDS A VOTE FOR THE APPOINTMENT

### OF ERNST & YOUNG LLP AS THE COMPANY S INDEPENDENT PUBLIC

### ACCOUNTANTS FOR THE FISCAL YEAR ENDING DECEMBER 31, 2008.

### **PROPOSAL 3**

#### Amendments to the 1998 Employee Stock Purchase Plan

At the annual meeting, the stockholders will be asked to approve and ratify an amendment to the Company s 1998 Employee Stock Purchase Plan (the 1998 Plan) adopted by the Board of Directors. Among other changes, the amendment to the 1998 Plan proposes to increase by 600,000 the maximum number of shares of Common Stock that may be issued under the 1998 Plan.

The stockholders initially approved the 1998 Plan with a reserve of 250,000 shares, and subsequently approved additional amendments to increase the share reserve to 1,850,000. As of February 29, 2008, approximately 200,000 shares remained available for purchase by participants. To enable participants to continue to participate in the 1998 Plan, the Board of Directors approved an amendment to the 1998 Plan, subject to ratification by the stockholders at the Annual Meeting, to reserve an additional 600,000 shares for issuance under the 1998 Plan.

#### Summary of the 1998 Plan

The following summary of the 1998 Plan is qualified in its entirety by the specific language of the 1998 Plan, a copy of which is attached to this Proxy Statement as Appendix A and which may also be accessed from the SEC s website at *http://www.sec.gov.* In addition, a copy of the 1998 Plan may be obtained upon written request to the Company a copy of which is available to any shareholder upon request.

*General.* The 1998 Plan is intended to qualify as an employee stock purchase plan under section 423 of the Internal Revenue Code (the Code). Each participant will be granted upon entry into an offering period under the 1998 Plan the right to purchase (a Purchase Right) through accumulated payroll deductions up to a number of shares of the Common Stock of the Company determined in accordance with the 1998 Plan. A participant s Purchase Right will be automatically exercised on each successive purchase date during the offering period unless the Purchase Right has terminated prior to such date.

*Shares Subject to Plan.* As currently amended, the 1998 Plan provides for the issuance of a maximum of 1,850,000 of the Company s authorized but unissued or reacquired shares of Common Stock, subject to appropriate adjustment in the event of any stock dividend, stock split, reverse stock split, recapitalization, combination of shares, exchange of shares or other change affecting the outstanding Common Stock as a class without the receipt of consideration by the Company. If any Purchase Right expires or terminates, the shares subject to the unexercised portion of such Purchase Right will again be available for issuance under the 1998 Plan.

*Administration.* The 1998 Plan will be administered by the Board of Directors or a duly appointed committee of the Board of Directors (the Plan Administrator). Subject to the provisions of the 1998 Plan, the Plan Administrator determines the terms and conditions of Purchase Rights granted under the plan. The Company may adopt such rules, policies, procedures, limitations or guidelines as it deems advisable for proper administration of the plan, consistent with the requirements of section 423 of the Code. The Plan Administrator will interpret the 1998 Plan, and decisions of the Plan Administrator are final and binding on all parties having an interest in the plan.

*Eligibility.* All employees of the Company or of any parent or subsidiary corporation of the Company designated by the Plan Administrator for inclusion in the 1998 Plan is eligible to participate in the plan. However, no employee who owns or holds options to purchase, or who, as a result of being granted a Purchase Right under the 1998 Plan, would own or hold options to purchase, 5% or more of the total combined voting power or value of all classes of stock of the Company or of any parent or subsidiary corporation of the Company may be granted a Purchase Right under the 1998 Plan. As of February 29, 2008, approximately 4,283 employees, including 10 executive officers, are eligible to participate in the 1998 Plan.

*Offering Periods.* Shares of Common Stock are offered under the 1998 Plan through a series of successive offering periods having a duration established by the Board of Directors not exceeding 24 months. Generally, offering periods have had durations of 24 months and have been composed of a series of eight (8) quarterly purchase intervals commencing on the first business days of February, May, August and November of each year. Purchases occur on the last day of each purchase interval. Should the fair market value per share of Common Stock on any purchase date during an offering period (other than the final purchase date of any offering period) be less than the fair market value per share at the start of the offering period, then that offering period will automatically terminate immediately following the purchase of shares and a new offering period will commence on the next business day following the purchase date. A participant may participate in only one offering period under the 1998 Plan at any given time.

*Participation and Purchase of Shares.* Eligible employees may commence participation in the 1998 Plan at the beginning of an offering period. These entry dates will generally occur on the first business days of February, May, August and November of each year. To enroll in the plan, an eligible employee must authorize payroll deductions prior to the applicable entry date. Payroll deductions may not exceed 15% of a participant s base compensation (as defined by the 1998 Plan) during each purchase interval within an offering period, unless a different limit is established by the Plan Administrator. A participant s authorized payroll deductions will continue throughout the offering period, unless (i) the participant makes an election to increase or decrease the rate of or to stop his or her payroll deductions, (ii) the participant voluntarily terminates his or her Purchase Right, or (iii) the participant ceases to be eligible to participate in the 1998 Plan.

Upon termination of a participant s Purchase Right, the Company will refund without interest the participant s accumulated payroll deductions not previously applied to the purchase of shares. Once a participant s Purchase Right in an offering period has terminated, the participant may not resume participation in the same offering period and may only resume participation by enrolling in a subsequent offering period.

Subject to certain limitations, each participant in an offering period will be granted on his or her entry date a Purchase Right exercisable for the lesser of 5,000 shares of Common Stock (as determined for the entire offering period), or that number of whole shares determined by dividing the participant s payroll deductions accumulated during the purchase interval ending on the purchase date by the applicable purchase price. However, no participant may be granted a Purchase Right that would permit the participant to purchase shares of Common Stock under the 1998 Plan or any other employee stock purchase plan of the Company or of any parent or subsidiary corporation of the Company having a fair market value exceeding \$25,000 for each calendar year (measured by the fair market value of the Company s Common Stock on the date the Purchase Right would be granted) in which the Purchase Right is outstanding at any time. Purchase Rights are nontransferable and may only be exercised by the participant.

As soon as practicable after the last business day of each purchase interval during an offering period, the Company will issue to each participant in the offering period the number of shares of the Company s Common Stock determined by dividing the amount of payroll deductions accumulated for the participant during the purchase interval by the purchase price, subject to the limitations described above. The price at which shares are sold under the 1998 Plan will be established by the Plan Administrator but may not be less than 85% of the lesser of the fair market value per share of Common Stock on the participant s entry date into the offering period or on the purchase date. Fair market value means the closing price of a share of our Common Stock on any given date. On February 29, 2008, the closing price of the Company s Common Stock on the Nasdaq Global Select Market was \$84.49 per share. Any payroll deductions under the 1998 Plan not applied to the purchase of shares on any purchase date will be returned to the participant without interest, unless the amount remaining is less than the amount necessary to purchase a whole share of Common Stock, in which case the remaining amount may be applied to purchase shares on the next purchase date.

*Change in Control.* In the event of a merger or consolidation to which the Company is a party or a sale or exchange of Company voting securities by the stockholders in which securities possessing more than 50% of the total combined voting power of the Company s outstanding securities are transferred to a person or persons different from the persons holding such securities immediately prior to the transaction or the sale, exchange or transfer of all or substantially all of the assets of the Company, each outstanding Purchase Right may be assumed by the surviving, continuing or purchasing corporation or parent corporation thereof. However, if such corporation elects not to assume the outstanding Purchase Rights, they will be exercised on a date prior to the effective date of such transaction specified by the Plan Administrator.

*Termination or Amendment.* The 1998 Plan will continue until terminated by the Plan Administrator or until all of the shares reserved for issuance under the plan have been issued. The Plan Administrator may at any time amend or terminate the 1998 Plan, except that the approval of the Company s stockholders is required within twelve months of the adoption of any amendment increasing the number of shares authorized for issuance under the 1998 Plan or changing the definition of the corporations which may be designated by the Plan Administrator as corporations the employees of which may participate in the 1998 Plan.

#### Summary of United States Federal Income Tax Consequences

The following summary is intended only as a general guide as to the United States federal income tax consequences under current law of participation in the 1998 Plan and does not attempt to describe all possible federal or other tax consequences of such participation or tax consequences based on particular circumstances.

The 1998 Plan is intended to qualify as an employee stock purchase plan within the meaning of section 423 of the Code. Provided that the 1998 Plan so qualifies, there are generally no tax consequences to an employee of either being granted a Purchase Right or purchasing shares.

The tax consequences of a disposition of shares acquired under the 1998 Plan vary depending on the period such stock is held before its disposition. If a participant disposes of shares within two years after his or her entry date into the offering period in which the shares are acquired or within one year after the purchase date on which the shares are acquired (a disqualifying disposition), the participant recognizes ordinary income in the year of disposition in an amount equal to the difference between the fair market value of the shares on the purchase date and the purchase price. Such income is not currently subject to income tax withholding, and the proposed regulations discussed above would continue this treatment. Any additional gain or resulting loss recognized by the participant from the disposition of the shares is a capital gain or loss.

If the participant disposes of shares more than two years after his or her entry date into the offering period in which the shares are acquired and more than one year after the purchase date on which the shares are acquired, the participant recognizes ordinary income in the year of disposition in an amount equal to the lesser of (i) the difference between the fair market value of the shares on the date of disposition and the purchase price or (ii) the difference between the fair market value of the shares price (determined as if the Purchase Right were exercised on the entry date). Any additional gain recognized by the participant on the disposition of the shares is a capital gain. If the fair market value of the shares on the date of disposition is less than the purchase price, there is no ordinary income, and the loss recognized is a capital loss.

A capital gain or loss will be long-term if the participant holds the shares for more than 12 months and short-term if the participant holds the shares for 12 months or less. Currently, long-term capital gains are generally subject to a maximum tax rate of 15%.

If the participant disposes of the shares in a disqualifying disposition, the Company should be entitled to a deduction equal to the amount of ordinary income recognized by the participant as a result of the disposition, except to the extent such deduction is limited by applicable provisions of the Code or the regulations thereunder. In all other cases, no deduction is allowed the Company.

### New Plan Benefits

Because benefits under the 1998 Plan will depend on employees elections to participate and to purchase shares under the 1998 Plan at various future dates, it is not possible to determine the benefits that will be received by executive officers and other employees. Non-employee directors are not eligible to participate in the 1998 Plan.

#### Vote Required and Board of Directors Recommendation

The affirmative vote of the holders of a majority of the shares present or represented by proxy and entitled to vote at the Annual Meeting is required for adoption of this proposal. Abstentions and broker non-votes will be counted as present for purposes of determining the presence of a quorum. However, abstentions and broker non-votes will have no effect on the outcome of this proposal.

The Board of Directors believes that the opportunity to purchase shares under the 1998 Plan is an important factor in motivating and maintaining the morale of the Company s valuable employees. The Board believes equity-based reward programs such as the 1998 Plan are valuable tools to retain the Company s valued employees and to closely align their interests with those of our stockholders. Consequently, the Board believes that it is in the best interests of our stockholders to approve the adoption of the 1998 Plan.

#### THEREFORE, THE BOARD UNANIMOUSLY RECOMMENDS A VOTE FOR APPROVAL OF THE AMENDMENT

TO THE 1998 PLAN.

#### **PROPOSAL 4**

Amendments to the Invitrogen Corporation 2004 Equity Incentive Plan

#### **Proposal Summary**

At the annual meeting, the stockholders will be asked to approve amendments to the 2004 Plan. The Compensation and Organizational Development Committee of the Board of Directors (the Compensation Committee) adopted the amendments to the 2004 Plan on February 27, 2008, subject to its approval by stockholders.

Currently, the remaining shares available for future awards under the 2004 Plan are insufficient to meet our long-term incentive needs. The proposed amendments to the 2004 Plan therefore provide for an additional 1,500,000 shares, for a total of 2,742,648 shares available for awards to be made under the 2004 Plan. The Company estimates that these additional shares will be sufficient to make awards under the 2004 Plan for the next one to two years.

The Company believes that its future success depends heavily on its ability to attract, motivate and retain the highest caliber employees. Equity is a key component of Invitrogen's total compensation package and closely aligns key employees interests with those of the Company's stockholders. Given competitive market norms for compensation in the life sciences industry, the Company would be unable to compete effectively for critical technical and professional talent absent sufficient equity incentives without significantly increasing Invitrogen's cash compensation costs.

The board of directors unanimously recommends voting for this proposal.

A complete copy of the revised plan is attached as Appendix B. You are urged to read this entire proposal and the complete plan document. We believe these amendments are necessary to keep and attract key employees critical to Invitrogen s success, and thus are in the best interests of Invitrogen s stockholders. We have explained our rationale for supporting this proposal in the section Why We Believe You Should Vote For this Proposal below.

#### Why We Believe You Should Vote For this Proposal

1. The proposal supports our continued emphasis on performance-based compensation.

Long-term incentive grants to our executive officers have focused on incentivizing and rewarding performance through stockholder value creation. Invitrogen has continued to grant stock options in recent years while many other companies have moved to granting time-vested restricted stock units. Stock options are inherently performance-based as they only have value when the stock price increases after the date of grant, unlike time-vested restricted shares. Further displaying our commitment to performance-based compensation, in 2007 we granted our CEO (Mr. Lucier) an award of performance shares tied to stock price achievement targets. This grant, which represents Mr. Lucier s only long-term incentive compensation for 2007, 2008, and 2009, requires Mr. Lucier to remain employed at Invitrogen for the full 3-year period and only provides full payouts at a share price of \$105 (a 63% premium to the \$64.27 share price on the date of grant. The threshold stock price below which no shares will be awarded under Mr. Lucier s 2007 grant was \$80.00, reflecting a 24% increase in value over the grant date. In 2007, 20% of the long-term incentive grants made to Mr. Lucier s direct reports also included performance share awards with stock price hurdles.

Through our continued use of stock options, as well as through our share-price-based performance share program, Invitrogen has demonstrated our commitment to linking executive pay to company performance and stockholder value creation.

2. The Long-Term Incentive Plan has broad participation and is instrumental in recruiting and retaining talent. Long-term incentive awards represent a significant portion of the compensation package for many Invitrogen employees, not just the top executives and Board members. For the period 2005 2007, 76% of the equity grants made by the company were made to individuals other than Named Executive Officers. Currently, long-term incentive awards can be made under the following programs/practices:

Sales Professionals Recognize leading Regional Managers, consistently outstanding Account Managers and Technical Sales Specialists

High Potential Employees Motivate and retain future leaders

CEO Discretionary Pool Immediate, positive reward/recognition to key employees

Acquisitions Retain key talent and build allegiance to Invitrogen

Our ability to compete for key employee talent against other life sciences industry companies is dependent upon our ability to grant equity-based pay. Like other employees in the life sciences industry, our key employees consider equity-based awards to be an important part of their overall annual compensation. Long-term incentive awards are commonplace among companies with which we compete for senior management, scientific/technical talent and other key employees. As a result, we expect to continue providing equity awards as a strategic component of total compensation to attract and retain critical employees.

3. We have reduced our equity utilization in recent years.

Based upon analyses performed by the Compensation Committees independent compensation consultant, Invitrogen s annual share grants under the long-term incentive program (annual burn rate ) have somewhat exceeded our Peer Group norms (discussed in the Compensation Discussion

and Analysis). The annual shares granted by Invitrogen under the long-term incentive program (annual burn rate ) have been actively managed down over the last 3 years. In 2004, our gross burn rate (shares and options granted divided by common shares outstanding at year end) was 4.6%. We reduced this figure to 3.2% by 2006, and maintained this rate of 3.2% in 2007 (excluding the performance-share award granted to Mr. Lucier in 2007). In the future, we intend to continue to reduce our burn rate to align with industry norms.

4. Due to significant stock option award cancellations, net dilution to stockholders has been substantially less than our burn rate. Due to our continued use of stock options, which are more likely to be cancelled than full-value awards, the net dilution incurred by Invitrogen s stockholders has been significantly less than our gross run rate. On average, for each of the past 3 years, we have cancelled awards primarily due to planned turnover equal to 1.24% of the common shares outstanding at the company, which significantly exceeds the cancellation rate of our Peer Group. The median Peer Group cancellation rate is .44% per year. Therefore, assessments of Invitrogen s equity utilization based on full-value awards may overstate the dilutive impact of our compensation program. Our Compensation and Organizational Development Committee and Board of Directors will actively seek to balance meeting employee hiring, retention and compensation goals and avoiding excessive stockholder dilution.

Our Compensation and Organizational Development Committee and Board of Directors will actively seek to balance meeting employee hiring, retention and compensation goals and avoiding excessive stockholder dilution.

#### 5. Invitrogen s Overhang is largely due to employees holding vested options.

Equity Incentive overhang is defined as the sum of (a) shares available under existing plans, (b) awards granted under existing plans but not yet exercised and (c) the number of shares being added under the proposed amendment as a percent of the Company s fully diluted shares (including those requested under this proposal). As of December 31, 2007, Invitrogen s overhang (without the shares requested under this proposal) was calculated at 14%. The addition of the 1,500,000 equity incentive shares requested increases the overhang to 15.9%.

A significant portion of Invitrogen's senior leadership team has been hired by the Company during the past 4 years. In accordance with the Company's executive compensation philosophy, these executives were granted stock options upon joining the company. Given the Company's strong financial performance since their hiring, many of these executives have had the opportunity to exercise stock options that are vested and are well in the money, but they have not done so because of their belief in our Company's future. For the same reason, many other employees have held their vested stock options rather than exercising them. One result of individuals holding their options is that Invitrogen has a large overhang of stock options. As of December 31, 2007, executives and employees held approximately 2.6 million vested but unexercised options with a value approaching \$95 million based on an average strike price of \$58.85 per share. We believe these figures reflect employee confidence in the future of the Company and continued alignment of interests between stockholders and employees.

6. The investor-supported Stock Buy-back Program has increased the dilutive impact of equity grants. Since 2006, Invitrogen has repurchased approximately 9.5 million shares of common stock under an authorized program. These repurchases raise the total potential dilution and burn rate by reducing the number of common shares outstanding (without the repurchase program, Invitrogen s dilution would be 14.7%). Nevertheless, the Board believes the repurchase program has been beneficial to stockholders and intends to continue its authorized buy-back program as long as it is in the best interests of stockholders.

7. Equity Long-Term Incentives are instrumental in driving Company performance.

The Company strongly believes that granting equity based awards more broadly encourages employees to think and behave like owners of the Company, rewarding all grant recipients when value is created for stockholders. New hire, annual performance and acquisition grants have been instrumental in driving Invitrogen s performance and share price as well as in retaining our key employees. We believe the continued use of equity compensation is necessary to recruit and retain key employees and focus critical employees on maximizing stockholder value while paying competitively and minimizing cash compensation costs.

#### Highlights of Key 2004 Plan Provisions

The 2004 Plan includes provisions designed to serve stockholders interests and promote effective corporate governance, including the following:

No Evergreen Provision. Assuming approval of the share reserve requested in this proposal, the 2004 Plan fixes the number of shares available for future grants to 15,000,000 shares and does not provide for any increase based on an increase in the number of outstanding shares of common stock.

No Discounted Stock Options. The 2004 Plan prohibits the granting of stock options, which must be exercised within ten years of the date of grant, at an exercise price that is less than the fair market value of the common stock on the date the stock option is granted.

Higher Share Counting for Full-Value Awards. While any shares subject to stock options or stock appreciation rights count against the share reserve under the 2004 Plan as one share of stock for each stock option and/or stock appreciation right, any shares of stock subject to any other award type under the 2004 Plan (e.g., restricted stock, restricted stock units, performance shares and performance units) shall count against the share reserve as 1.6 shares for each share subject to such full-value award.

Limitations on Size of Awards. The 2004 Plan limits the number of shares that may be granted to any participant in any calendar year (with an additional amount permitted in the first year of employment with the Company as summarized below) subject to: 1,000,000 stock options and stock appreciation rights awards;

100,000 restricted shares and restricted stock units; and,

600,000 performance-based awards.

Minimum Vesting Periods. All awards are subject to a minimum one-year vesting schedule.

No Repricings. Neither the 2004 Plan nor any award agreement may be amended to provide for the repricing of any stock option and/or stock appreciation right.

No Liberal Share Reserve Features. The 2004 Plan does not contain any liberal share counting features. Specifically, the only shares that ever return to the share reserve and thus become available for new awards under the 2004 Plan are those awards that are forfeited or repurchased and/or are settled in cash.

### Summary of the 2004 Plan

The following summary of the 2004 Plan is qualified in its entirety by the specific language of the 2004 Plan, a copy of which is attached to this Proxy Statement as Appendix B and which may also be accessed from the SEC s website at *http://www.sec.gov.* In addition, a copy of the 2004 Plan may be obtained upon written request to the Company a copy of which is available to any stockholder upon request.

*General.* The purpose of the 2004 Plan is to advance the interests of the Company by providing an incentive program that will enable the Company to attract and retain employees, consultants and directors upon whose judgment, interest and efforts the Company s success is dependent and to provide them with an equity interest in the success of the Company in order to motivate superior performance. These incentives are provided through the grant of stock options (including indexed options), stock appreciation rights, restricted stock purchase rights, restricted stock bonuses, restricted stock units, performance shares, performance units and deferred stock units.

*Authorized Shares.* When the 2004 Plan was initially approved by the stockholders, all of our predecessor stock plans (the Prior Plans) were terminated and a total of five million seven hundred thousand (5,700,000) shares of our Common Stock were reserved for the granting of new awards under the 2004 Plan. However, the actual 2004 Plan share reserve also included all of the options and other awards the Company has granted which are still outstanding as of April 29, 2004 (collectively the Prior Awards) to the extent any such Prior Award expires, lapses or otherwise terminates for any reason without having been exercised or settled in full, or if any shares subject to forfeiture or repurchase are forfeited or repurchased by the Company. Consequently, the maximum number of shares which may be granted under the 2004 Plan, prior to adoption of this amendment to increase the share reserve by 1,500,000 shares, assuming all of the Prior Awards actually do expire, terminate or otherwise lapse unexercised and are returned to the 2004 Plan share reserve, was equal to thirteen million five hundred and sixteen thousand (13,516,000). If stockholders approve this amendment, the maximum number of share shares of Common Stock which may granted under the 2004 Plan shall be increased to 15,000,000 (however, only 2,742,648 shares of Common Stock shall actually be available for grant of new awards). Upon any stock dividend, stock split, reverse stock split, recapitalization or similar change in our capital structure, appropriate adjustments will be made to the shares subject to the 2004 Plan, to the award grant limitations and to all outstanding awards.

*No Liberal Share Reserve Features.* If any outstanding award, including any Prior Award, for any reason expires or is terminated or canceled without having been exercised or settled in full, or if shares of Common Stock acquired pursuant to an award subject to forfeiture or repurchase, including any Prior Awards, are forfeited or repurchased by the Company, the shares of Common Stock allocable to the terminated portion of such award, including any Prior Awards, or such forfeited or repurchased shares of Common Stock shall again be available for grant under the 2004 Plan. Shares of Common Stock shall not be deemed to have been granted pursuant to the 2004 Plan with respect to any portion of an award that is settled in cash. Notwithstanding the foregoing, the following shares of Common Stock shall not be available for reissuance under the 2004

Plan: (i) shares of Common Stock that are withheld and/or surrendered in satisfaction of tax withholding obligations; (ii) shares of Common Stock which are surrendered, through tender to the Company or attestation of ownership, in payment of any applicable exercise price; and (iii) shares of Common Stock subject to the grant of a stock appreciation right which are not required to be issued upon the exercise of such stock appreciation right.

*Share Accounting.* Any shares of our Common Stock which are granted under the 2004 Plan in the form of stock options or stock appreciation rights shall be counted against the 2004 Plan share reserve as one share of our Common Stock for every one share subject to the stock option or stock appreciation right. However, any shares of our Common Stock, which are granted under the 2004 Plan in an award form other than stock options or stock appreciation rights, shall be counted against the 2004 Plan share reserve as 1.6 shares of our Common Stock for every one share subject to such award.

*Administration.* The 2004 Plan will be administered by the Compensation and Organizational Development Committee of the Board of Directors duly appointed to administer the 2004 Plan, or, in the absence of such committee, by the Board of Directors. In the case of awards intended to qualify for the performance-based compensation exemption under Section 162(m) of the Code, administration must be by a compensation committee composed solely of two or more outside directors within the meaning of Section 162(m). (For purposes of this summary, the term Committee will refer to either such duly appointed committee or the Board of Directors.) Subject to the provisions of the 2004 Plan, the Committee determines in its discretion the persons to whom and the times at which awards are granted, the types and sizes of such awards, and all of their terms and conditions. The Committee may, subject to certain limitations on the exercise its discretion required by Section 162(m), amend, cancel, renew, or grant a new award in substitution for, any award, waive any restrictions or conditions applicable to any award, and accelerate, continue, extend or defer the vesting of any award. However, the 2004 Plan forbids, without stockholder approval, the repricing of any outstanding stock option and/or stock appreciation right. The 2004 Plan provides, subject to certain limitations, for indemnification by the Company of any director, officer or employee against all reasonable expenses, including attorneys fees, incurred in connection with any legal action arising from such person s action or failure to act in administering the 2004 Plan. The Committee will interpret the 2004 Plan and awards granted thereunder, and all determinations of the Committee will be final and binding on all persons having an interest in the 2004 Plan or any

*Eligibility.* Awards may be granted to employees, directors and consultants of the Company or any present or future parent or subsidiary corporations of the Company. Incentive stock options may be granted only to employees who, as of the time of grant, are employees of the Company or any parent or subsidiary corporation of the Company. As of February 1, 2008, the Company had approximately 4,283 employees, including 10 executive officers and 9 directors who would be eligible under the 2004 Plan. We also have several consultants who would be eligible under the 2004 Plan.

award.

*Stock Options.* Each option granted under the 2004 Plan must be evidenced by a written agreement between the Company and the optionee specifying the number of shares subject to the option and the other terms and conditions of the option, consistent with the requirements of the 2004 Plan. The exercise price of each option may not be less than the fair market value of a share of Common Stock on the date of grant. However, any incentive stock option granted to a person who at the time of grant owns stock possessing more than 10% of the total combined voting power of all classes of stock of the Company or any parent or subsidiary corporation of the Company (a Ten Percent Stockholder) must have an exercise price equal to at least 110% of the fair market value of a share of Common Stock on the date of grant. The exercise price of each indexed stock option, and the terms and adjustments which may be made to such an option, will be determined by the Committee in its sole discretion at the time of grant. On February 29, 2008, the closing price of the Company s Common Stock on the Nasdaq Global Select Market was \$84.49 per share. Subject to appropriate adjustment in the event of any change in the capital structure of the Company, no employee may be granted in any fiscal year of the Company options which in the aggregate are for more than one million (1,000,000) shares, provided however, that the Company may make an additional one-time grant to any newly-hired employee of a stock option for the purchase of up to an additional five hundred thousand (500,000) shares.

The 2004 Plan provides that the option exercise price may be paid in cash, by check, or in cash equivalent, by the assignment of the proceeds of a sale with respect to some or all of the shares being acquired upon the exercise of the option, to the extent legally permitted, by tender of shares of Common Stock owned by the optionee having a fair market value not less than the exercise price by such other lawful consideration as approved by the Committee, or by any combination of these. However, if the optionee tenders shares of Common Stock in payment of the exercise price, those shares are not added to the share reserve. Also, the Committee may restrict the forms of payment permitted in connection with any option grant. No option may be exercised unless the optionee has made adequate provision for federal, state, local and foreign taxes, if any, relating to the exercise of the option, including, if permitted or required by the Company, through the optionee s surrender of a portion of the option shares to the Company.

Options will become vested and exercisable at such times or upon such events and subject to such terms, conditions, performance criteria or restrictions as specified by the Committee. The maximum term of any option granted under the 2004 Plan is ten years, provided that an incentive stock option granted to a Ten Percent Stockholder must have a term not exceeding five years. The Committee will specify in each written option agreement, and solely in its discretion, the period of post-termination exercise applicable to each option.

Stock options are nontransferable by the optionee other than by will or by the laws of descent and distribution, and are exercisable during the optionee s lifetime only by the optionee. However, a nonstatutory stock option may be assigned or transferred to the extent permitted by the Committee in its sole discretion.

*Stock Appreciation Rights.* Each stock appreciation right granted under the 2004 Plan must be evidenced by a written agreement between the Company and the participant specifying the number of shares subject to the award and the other terms and conditions of the award, consistent with the requirements of the 2004 Plan.

A stock appreciation right gives a participant the right to receive the appreciation in the fair market value of Company Common Stock between the date of grant of the award and the date of its exercise. The Company may pay the appreciation either in cash or in shares of Common Stock. The Committee may grant stock appreciation rights under the 2004 Plan in tandem with a related stock option or as a freestanding award. A tandem stock appreciation right is exercisable only at the time and to the same extent that the related option is exercisable, and its exercise causes the related option to be canceled. Freestanding stock appreciation rights vest and become exercisable at the times and on the terms established by the Committee. A stock appreciation right may not be granted at less than the fair market value of a share of Company Common Stock on the date of grant. The maximum term of any stock appreciation right granted under the 2004 Plan is ten years. Subject to appropriate adjustment in the event of any change in the capital structure of the Company, no employee may be granted in any fiscal year of the Company stock appreciation rights which in the aggregate are for more than one million (1,000,000) shares, provided however, that the Company may make an additional one-time grant to any newly-hired employee of a stock appreciation right for the purchase of up to an additional five hundred thousand (500,000) shares.

Stock appreciation rights are generally nontransferable by the participant other than by will or by the laws of descent and distribution, and are generally exercisable during the participant s lifetime only by the participant.

*Restricted Stock Awards*. The Committee may grant restricted stock awards under the 2004 Plan either in the form of a restricted stock purchase right, giving a participant an immediate right to purchase Common Stock, or in the form of a restricted stock bonus, for which the participant furnishes consideration in the form of services to the Company. The Committee determines the purchase price payable under restricted stock purchase awards, which may be less than the then current fair market value of our Common Stock. Restricted stock awards may be subject to vesting conditions based on such service or performance criteria as the Committee specifies, and the shares acquired may not be transferred by the participant until vested. Unless otherwise provided by the Committee, a participant will forfeit any shares of restricted stock as to which the restrictions have not lapsed prior to the participant s termination of service. Participants holding restricted stock will have the right to vote the shares and to receive any dividends paid, except that dividends or other distributions paid in shares will be subject to the same restrictions as the original award. Subject to appropriate adjustment in the event of any change in the capital structure of the Company, no employee may be granted in any fiscal year of the Company more than one hundred thousand (100,000) shares of restricted stock on which the restrictions are based on performance criteria, provided however, that the Company may make an additional one-time grant to any newly-hired employee of a restricted stock award of up to an additional fifty thousand (50,000) shares.

*Restricted Stock Units.* Under the 2004 Plan, the Committee may grant restricted stock units that represent a right to receive shares of Common Stock at a future date determined in accordance with the participant s award agreement. No monetary payment is required for receipt of restricted stock units or the shares issued in settlement of the award, the consideration for which is furnished in the form of the participant s services to the Company. The Committee may grant restricted stock unit awards subject to the attainment of performance goals similar to those described below in connection with performance shares and performance units, or may make the awards subject to vesting conditions similar to those applicable to restricted stock awards. Participants have no voting rights or rights to receive cash dividends with respect to restricted stock unit awards until shares of Common Stock are issued in settlement of such awards. However, the Committee may grant restricted stock units that entitle their holders to receive dividend equivalents, which are rights to receive additional restricted stock units for a number of shares whose value is equal to any cash dividends we pay. Subject to appropriate adjustment in the event of any change in the capital structure of the Company, no employee may be granted in any fiscal year of the Company more than one hundred thousand (100,000) restricted stock units on which the restrictions are based on performance criteria, provided however, that the Company may make an additional one-time grant to any newly-hired employee of restricted stock units of up to an additional fifty thousand (50,000) shares.

# Edgar Filing: INVITROGEN CORP - Form DEFR14A

*Performance Awards*. The Committee may grant performance awards subject to such conditions and the attainment of such performance goals over such periods as the Committee determines in writing and sets forth in a written agreement between the Company and the participant. These awards may be designated as performance shares or performance units. Performance shares and performance units are unfunded bookkeeping entries generally having initial values, respectively, equal to the fair market value determined on the grant date of a share of Common Stock and \$100 per unit. Performance awards will specify a predetermined amount of performance shares or performance units that may be earned by the participant to the extent that one or more predetermined performance goals are attained within a predetermined performance period. To the extent earned, performance awards may be settled in cash, shares of Common Stock (including shares of restricted stock) or any combination thereof. Subject to appropriate adjustment in the event of any change in the capital structure of the Company, for each fiscal year of the Company contained in the applicable performance period, no employee may be granted performance shares that could result in the employee receiving more than six hundred thousand (600,000) shares of Common Stock or performance units that could result in the employee receiving more than four million dollars (\$4,000,000). A participant may receive only one performance award with respect to any performance period.

Prior to the beginning of the applicable performance period or such later date as permitted under Section 162(m) of the Code, the Committee will establish one or more performance goals applicable to the award. Performance goals will be based on the attainment of specified target levels with respect to one or more measures of business or financial performance of the Company and each parent and subsidiary corporation consolidated therewith for financial reporting purposes, or such division or business unit of the Company as may be selected by the Committee. The Committee, in its discretion, may base performance goals on one or more of the following such measures, which may be used individually or in tandem with other measures: revenue, gross margin, operating margin, operating income, pre-tax profit, earnings before interest, taxes, depreciation and/or amortization, net income, cash flow, expenses, stock price, earnings per share, return on stockholder equity, return on capital, return on net assets, economic value added, number of customers, market share, same store sales, return on investment, profit after tax and guest and/or customer satisfaction. The target levels with respect to these performance measures may be expressed on an absolute basis or relative to a standard specified by the Committee. The degree of attainment of performance measures will, according to criteria established by the Committee, be computed before the effect of changes in accounting standards, restructuring charges and similar extraordinary items occurring after the establishment of the performance goals applicable to a performance award.

Following completion of the applicable performance period, the Committee will certify in writing the extent to which the applicable performance goals have been attained and the resulting value to be paid to the participant. The Committee retains the discretion to eliminate or reduce, but not increase, the amount that would otherwise be payable to the participant on the basis of the performance goals attained. However, no such reduction may increase the amount paid to any other participant. In its discretion, the Committee may provide for the payment to a participant awarded performance shares of dividend equivalents with respect to cash dividends paid on the Company s Common Stock. Performance award payments may be made in lump sum or in installments. If any payment is to be made on a deferred basis, the Committee may provide for the payment of dividend equivalents or interest during the deferral period.

Unless otherwise provided by the Committee, if a participant s service terminates due to the participant s death, disability or retirement prior to completion of the applicable performance period, the final award value will be determined at the end of the performance period on the basis of the performance goals attained during the entire performance period but will be prorated for the number of months of the participant s service during the performance period. If a participant s service terminates prior to completion of the applicable performance period for any other reason, the 2004 Plan provides that, unless otherwise determined by the Committee, the performance award will be forfeited. No performance award may be sold or transferred other than by will or the laws of descent and distribution prior to the end of the applicable performance period.

*Deferred Stock Awards.* The 2004 Plan provides that certain participant s who are executives or members of a select group of highly compensated employees may elect to receive, in lieu of payment in cash or stock of all or any portion of such participant s cash and/or stock compensation, an award of deferred stock units. A participant electing to receive deferred stock units will be granted automatically, on the effective date of such deferral election, an award (a Deferred Stock Unit Award) for a number of stock units equal to the amount of the deferred compensation divided by an amount equal to the fair market value of a share of our Common Stock as quoted by the national or regional securities exchange or market system on which the Common Stock is listed on the date of grant. A stock unit is an unfunded bookkeeping entry representing a right to receive one share of our Common Stock in accordance with the terms and conditions of the Deferred Stock Unit Award. Participant s are not required to pay any additional cash consideration in connection with the settlement of a Deferred Stock Unit Award. A participant s compensation not paid in the form of a Deferred Stock Unit Award will be paid in cash in accordance with the Company s normal payment procedures.

# Edgar Filing: INVITROGEN CORP - Form DEFR14A

Each Deferred Stock Unit Award will be evidenced by a written agreement between the Company and the participant specifying the number of stock units subject to the award and the other terms and conditions of the Deferred Stock Unit Award, consistent with the requirements of the 2004 Plan. Deferred Stock Unit Awards are fully vested upon grant and will be settled by distribution to the participant of a number of whole shares of Common Stock equal to the number of stock units subject to the award terms and the participant s election. A holder of stock unit has no voting rights or other rights as a stockholder until shares of Common Stock are issued to the participant in settlement of the stock unit. However, participants holding stock units will be entitled to receive dividend equivalents with respect to any payment of cash dividends on an equivalent number of shares of Common Stock. Such dividend equivalents will be credited in the form of additional whole and fractional stock units determined by the fair market value of a share of Common Stock on the dividend payment date. Prior to settlement, no Deferred Stock Unit Award may be assigned or transferred other than by will or the laws of descent and distribution.

*Change in Control.* The 2004 Plan defines a Change in Control of the Company as any of the following events upon which the stockholders of the Company immediately before the event do not retain immediately after the event, in substantially the same proportions as their ownership of shares of the Company s voting stock immediately before the event, direct or indirect beneficial ownership of a majority of the total combined voting power of the voting securities of the Company, its successor or the corporation to which the assets of the Company were transferred: (i) a sale or exchange by the stockholders in a single or series of related transactions of more than 50% of the Company s voting stock; (ii) a merger or consolidation in which the Company is a party; (iii) the sale, exchange or transfer of all or substantially all of the assets of the Company; or (iv) a liquidation or dissolution of the Company. If a Change in Control occurs, the surviving, continuing, successor or purchasing corporation or parent corporation thereof may either assume all outstanding awards or substitute new awards having an equivalent value.

In the event of a Change in Control and the outstanding stock options and stock appreciation rights are not assumed or replaced, then all unexercisable, unvested or unpaid portions of such outstanding awards will become immediately exercisable, vested and payable in full immediately prior to the date of the Change in Control.

In the event of a Change in Control, the lapsing of all vesting conditions and restrictions on any shares subject to any restricted stock award, restricted stock unit and performance award held by a participant whose service with the Company has not terminated prior to the Change in Control may accelerate in accordance with the terms of the award.

Any award not assumed, replaced or exercised prior to the Change in Control will terminate. The 2004 Plan authorizes the Committee, in its discretion, to provide for different treatment of any award, as may be specified in such award s written agreement, which may provide for acceleration of the vesting or settlement of any award, or provide for longer periods of exercisability, upon a Change in Control.

*Termination or Amendment.* The 2004 Plan will continue in effect until the first to occur of (i) its termination by the Committee, (ii) the date on which all shares available for issuance under the 2004 Plan have been issued and all restrictions on such shares under the terms of the 2004 Plan and the agreements evidencing awards granted under the 2004 Plan have lapsed, or (iii) the tenth anniversary of the 2004 Plan s effective date. The Committee may terminate or amend the 2004 Plan at any time, provided that no amendment may be made without stockholder approval if the Committee deems such approval necessary for compliance with any applicable tax or securities law or other regulatory requirements, including the requirements of any stock exchange or market system on which the Common Stock of the Company is then listed. No termination or amendment may affect any outstanding award unless expressly provided by the Committee, and, in any event, may not adversely affect an outstanding award without the consent of the participant unless necessary to comply with any applicable law, regulation or rule.

#### Summary of U.S. Federal Income Tax Consequences

The following summary is intended only as a general guide to the U.S. federal income tax consequences of participation in the 2004 Plan and does not attempt to describe all possible federal or other tax consequences of such participation or tax consequences based on particular circumstances.

*Incentive Stock Options*. An optionee recognizes no taxable income for regular income tax purposes as a result of the grant or exercise of an incentive stock option qualifying under Section 422 of the Code. Optionees who neither dispose of their shares within two years following the date the option was granted nor within one year following the exercise of the option will normally recognize a capital gain or loss equal to the difference, if any, between the sale price and the purchase price of the shares. If an optionee satisfies such holding periods upon a sale of the shares, the Company will not be entitled to any deduction for federal income tax purposes. If an optionee disposes of shares within two years after the date of grant or within one year after the date of exercise (a disqualifying disposition), the difference between the fair market value of the

shares on the determination date (see discussion under Nonstatutory Stock Options below) and the option exercise price (not to exceed the gain realized on the sale if the disposition is a transaction with respect to which a loss, if sustained, would be recognized) will be taxed as ordinary income at the time of disposition. Any gain in excess of that amount will be a capital gain. If a loss is recognized, there will be no ordinary income, and such loss will be a capital loss. Any ordinary income recognized by the optionee upon the disqualifying disposition of the shares generally should be deductible by the Company for federal income tax purposes, except to the extent such deduction is limited by applicable provisions of the Code.

The difference between the option exercise price and the fair market value of the shares on the determination date of an incentive stock option (see discussion under Nonstatutory Stock Options below) is treated as an adjustment in computing the optionee s alternative minimum taxable income and may be subject to an alternative minimum tax which is paid if such tax exceeds the regular tax for the year. Special rules may apply with respect to certain subsequent sales of the shares in a disqualifying disposition, certain basis adjustments for purposes of computing the alternative minimum taxable income on a subsequent sale of the shares and certain tax credits which may arise with respect to optionees subject to the alternative minimum tax.

Nonstatutory Stock Options and Indexed Stock Options. Options not designated or qualifying as incentive stock options, or as an indexed stock option, will be nonstatutory stock options having no special tax status. An optionee generally recognizes no taxable income as the result of the grant of such an option. Upon exercise of a nonstatutory stock option, the optionee normally recognizes ordinary income in the amount of the difference between the option exercise price and the fair market value of the shares on the determination date (as defined below). If the optionee is an employee, such ordinary income generally is subject to withholding of income and employment taxes. The determination date is the date on which the option is exercised unless the shares are subject to a substantial risk of forfeiture (as in the case where an optionee is permitted to exercise an unvested option and receive unvested shares which, until they vest, are subject to the Company s right to repurchase them at the original exercise price upon the optionee s termination of service) and are not transferable, in which case the determination date is the earlier of (i) the date on which the shares become transferable or (ii) the date on which the shares are no longer subject to a substantial risk of forfeiture. If the determination date is after the exercise date, the optionee may elect, pursuant to Section 83(b) of the Code, to have the exercise date be the determination date by filing an election with the Internal Revenue Service no later than 30 days after the date the option is exercised. Upon the sale of stock acquired by the exercise of a nonstatutory stock option, any gain or loss, based on the difference between the sale price and the fair market value on the determination date, will be taxed as capital gain or loss. No tax deduction is available to the Company with respect to the grant of a nonstatutory stock option or the sale of the stock acquired pursuant to such grant. The Company generally should be entitled to a deduction equal to the amount of ordinary income recognized by the optionee as a result of the exercise of a nonstatutory stock option, except to the extent such deduction is limited by applicable provisions of the Code.

*Stock Appreciation Rights.* No taxable income is reportable when a stock appreciation right is granted to a participant. Upon exercise, the participant will recognize ordinary income in an amount equal to the amount of cash received and the fair market value of any shares of our Common Stock received. Any additional gain or loss recognized upon any later disposition of the shares would be capital gain or loss.

*Restricted Stock Awards.* A participant acquiring restricted stock generally will recognize ordinary income equal to the fair market value of the shares on the determination date (as defined above under Nonstatutory Stock Options ). If the participant is an employee, such ordinary income generally is subject to withholding of income and employment taxes. If the determination date is after the date on which the participant acquires the shares, the participant may elect, pursuant to Section 83(b) of the Code, to have the date of acquisition be the determination date by filing an election with the Internal Revenue Service no later than 30 days after the date the shares are acquired. Upon the sale of shares acquired pursuant to a restricted stock award, any gain or loss, based on the difference between the sale price and the fair market value on the determination date, will be taxed as capital gain or loss. The Company generally should be entitled to a deduction equal to the amount of ordinary income recognized by the participant on the determination date, except to the extent such deduction is limited by applicable provisions of the Code.

*Performance and Restricted Stock Units Awards.* A participant generally will recognize no income upon the grant of a performance share, performance units or restricted stock units award. Upon the settlement of such awards, participants normally will recognize ordinary income in the year of receipt in an amount equal to the cash received and the fair market value of any nonrestricted shares received. If the participant is an employee, such ordinary income generally is subject to withholding of income and employment taxes. If the participant receives shares of restricted stock, the participant generally will be taxed in the same manner as described above (see discussion under Restricted Stock ). Upon the sale of any shares received, any gain or loss, based on the difference between the sale price and the fair market value on the determination date (as defined above under Nonstatutory Stock Options ), will be taxed as capital gain or loss. The Company generally

should be entitled to a deduction equal to the amount of ordinary income recognized by the participant on the determination date, except to the extent such deduction is limited by applicable provisions of the Code.

*Deferred Stock Unit Awards.* A participant generally will recognize no income upon the grant of a Deferred Stock Unit Award. Upon the settlement of such an award, the participant normally will recognize ordinary income in the year of settlement in an amount equal to the fair market value of any unrestricted shares of our Common Stock received. Upon the sale of any shares received, any gain or loss, based on the difference between the sale price and the fair market value on the determination date, will be taxed as capital gain or loss. The Company generally should be entitled to a deduction equal to the amount of ordinary income recognized by the participant on the determination date, except to the extent such deduction is limited by applicable provisions of the Internal Revenue Code.

## New Plan Benefits

No awards will be granted under the 2004 Plan with respect to shares covered by this amendment prior to the approval of the amendment by the stockholders of the Company. Awards under the 2004 Plan will be granted at the discretion of the Committee, and, accordingly, are not yet determinable. In addition, benefits under the 2004 Plan, will depend on a number of factors, including the fair market value of the Company s Common Stock on future dates, actual Company performance against performance goals established with respect to performance awards and decisions made by the participants. Consequently, it is not possible to determine the benefits that might be received by participants under the 2004 Plan.

The affirmative vote of a majority of the votes cast at the meeting, at which a quorum is present, either in person or by proxy, is required to approve the adoption of the proposed amendment to the 2004 Plan. If you hold your shares in your own name and abstain from voting on this matter, your abstention will have no effect on the vote. If you hold your shares through a broker and you do not instruct the broker on how to vote on this proposal, your broker will not have authority to vote your shares. Abstentions and broker non-votes will each be counted as present for purposes of determining the presence of a quorum but will not have any effect on the outcome of the proposal.

The Board believes that the proposed adoption of the amendment to the 2004 Plan is in the best interests of the Company and its stockholders for the reasons stated above.

## THEREFORE, THE BOARD UNANIMOUSLY RECOMMENDS

## A VOTE FOR APPROVAL OF THE ADOPTION OF THE AMENDMENT TO THE 2004 PLAN.

47

## ADDITIONAL INFORMATION

*Advance Notice Procedures.* Our bylaws require that, for business to be properly brought by a stockholder before an annual meeting, notice must be delivered by the stockholder and received at the offices of the Company not less than 120 days prior to the anniversary of the date of the prior year s proxy statement, except if we did not hold an annual meeting the previous year, or if the date of this year s Annual Meeting has been changed by more than 30 days from the date of the previous year s meeting, then the deadline is a reasonable time before we begin to print and mail our proxy materials.

*Stockholders Sharing the Same Last Name and Address.* In accordance with notices that we sent to certain stockholders, we are sending only one copy of the Company s Annual Report and Proxy Statement to stockholders who share the same last name and address, unless they have notified the Company that they want to continue receiving multiple copies. This practice, known as householding, is designed to reduce duplicate mailings and save significant printing and postage costs as well as natural resources.

If you received a householded mailing this year and you would like to have additional copies of the Company s Annual Report and/or Proxy Statement mailed to you, or you would like to opt out of this practice for future mailings, please submit your request to Investor Relations via e-mail at ir@invitrogen.com, by fax to (760) 603-7229 or by mail to Investor Relations, Invitrogen Corporation, 5791 Van Allen Way, Carlsbad, CA 92008, or call at (760) 603-7200. We will promptly send additional copies of the Annual Report and/or Proxy Statement upon receipt of such request. You may also contact the Company if you received multiple copies of the Annual Meeting materials and would prefer to receive a single copy in the future.

Householding for bank and brokerage accounts is limited to accounts within the same bank or brokerage firm. For example, if you and your spouse share the same last name and address, and you and your spouse each have two accounts containing Invitrogen stock at two different brokerage firms, your household will receive two copies of the Invitrogen Annual Meeting materials one from each brokerage firm.

*Stockholder Communications with Board of Directors.* Any stockholder who wishes to communicate with the Board of Directors may do so by writing to the Company s Secretary at the following address: 5791 Van Allen Way, Carlsbad, CA 92008.

*Stockholder Proposals for the Next Annual Meeting.* All stockholder proposals that are intended to be presented at the 2009 Annual Meeting of Stockholders of the Company must be received by the Company at our principal executive offices at 5791 Van Allen Way., Carlsbad, California 92008, ATTN: Corporate Secretary, no later than November 15, 2008, for inclusion in the Board of Directors Proxy Statement and form of proxy relating to the meeting. Any stockholder who intends to present a proposal at the Company s 2009 Annual Meeting of Stockholders without requesting the Company to include such proposal in the Company s Proxy Statement must notify the Company no later than January 29, 2009, of his, her or its intention to present the proposal. Otherwise, the Company may exercise discretionary voting with respect to such stockholder proposal pursuant to authority conferred on the Company by proxies to be solicited by the Board of Directors of the Company and delivered to the Company in connection with the meeting.

48

## TRANSACTION OF OTHER BUSINESS

At the date of this Proxy Statement, the only business the Board of Directors intends to present or knows that others will present at the Annual Meeting is as set forth above. If any other matter or matters are properly brought before the meeting, or any adjournment thereof, it is the intention of the persons named in the accompanying form of proxy to vote the proxy on such matters in accordance with their best judgment.

By Order of the Board of Directors

John A. Cottingham Senior Vice President, General Counsel & Secretary

March 14, 2008

Carlsbad, California

49

## Appendix A

#### INVITROGEN CORPORATION

## **1998 EMPLOYEE STOCK PURCHASE PLAN**

#### 1. <u>Establishment, Purpose and Term of Plan.</u>

1.1 Establishment. Invitrogen Corporation hereby establishes the 1998 Employee Stock Purchase Plan (the *Plan*), effective as of the effective date of the initial registration by the Company of its Stock under Section 12 of the Securities Exchange Act of 1934, as amended (the *Effective Date*).

1.2 **Purpose.** The purpose of the Plan is to advance the interests of the Company and its stockholders by providing an incentive to attract, retain and reward Eligible Employees of the Participating Company Group and by motivating such persons to contribute to the growth and profitability of the Participating Company Group. The Plan provides such Eligible Employees with an opportunity to acquire a proprietary interest in the Company through the purchase of Stock. The Company intends that the Plan qualify as an employee stock purchase plan under Section 423 of the Code (including any amendments or replacements of such section), and the Plan shall be so construed.

1.3 **Term of Plan.** The Plan shall continue in effect until the earlier of its termination by the Board or the date on which all of the shares of Stock available for issuance under the Plan have been issued.

#### 2. <u>Definitions</u> <u>and</u> <u>Construction</u>.

2.1 **Definitions.** Any term not expressly defined in the Plan but defined for purposes of Section 423 of the Code shall have the same definition herein. Whenever used herein, the following terms shall have their respective meanings set forth below:

(a) *Board* means the Board of Directors of the Company. If one or more Committees have been appointed by the Board to administer the Plan, Board also means such Committee(s).

(b) Code means the Internal Revenue Code of 1986, as amended, and any applicable regulations promulgated thereunder.

(c) *Committee* means the Compensation Committee, or another committee of the Board duly appointed to administer the Plan and having such powers as shall be specified by the Board. Unless the powers of the Committee have been specifically limited, the Committee shall have all of the powers of the Board granted herein, including, without limitation, the power to amend or terminate the Plan at any time, subject to the terms of the Plan and any applicable limitations imposed by law.

(d) Company means Invitrogen Corporation., a Delaware corporation, or any successor corporation thereto.

(e) *Compensation* means, with respect to any Offering Period, base wages or salary, commissions, overtime, bonuses, annual awards, other incentive payments, shift premiums, and all other compensation paid in cash during such Offering Period before deduction for any contributions to any plan maintained by a Participating Company and described in Section 401(k) or Section 125 of the Code. Compensation shall not include reimbursements of expenses, allowances, long-term disability, workers compensation or any amount deemed received without the actual transfer of cash or any amounts directly or indirectly paid pursuant to the Plan or any other stock purchase or stock option plan, or any compensation other than base wages or salary.

(f) Eligible Employee means an Employee who meets the requirements set forth in Section 5 for eligibility to participate in the Plan.

(g) *Employee* means any person who is treated as a regular employee for purposes of Section 423 of the Code of a Participating Company and whose customary employment is for not more than five months in any calendar year. A Participant shall be deemed to have ceased to be an Employee either upon an actual termination of employment or upon the corporation employing the Participant ceasing to be a Participating Company. For purposes of the Plan, an individual shall not be deemed to have ceased to be an Employee while such individual is on any military leave, sick leave, or other bona fide leave of absence approved by the Company of ninety (90) days or less. In the event an individual s leave of absence exceeds ninety (90) days, the individual shall be deemed to have ceased to be an Employee on the ninety-first (91st) day of such leave unless the individual s right to reemployment with the Participating Company Group is guaranteed either by statute or by contract. The Company shall determine in good faith and in the exercise of its discretion whether an individual has become or has ceased to be an Employee and the effective date of such individual s employment or termination of employment, as the case may be. For purposes of an individual s participation in or other rights, if any, under the Plan as of the time of the Company s determination, all such determinations by the Company shall be final, binding and conclusive, notwithstanding that the Company or any governmental agency subsequently makes a contrary determination.

(h) *Fair Market Value* means, as of any date, if there is then a public market for the Stock, the closing price of a share of Stock (or the mean of the closing bid and asked prices if the Stock is so quoted instead) as quoted on the Nasdaq National Market, the Nasdaq Small-Cap Market or such other national or regional securities exchange or market system constituting the primary market for the Stock, as reported in <u>The Wall</u> <u>Street Journal</u> or such other source as the Company deems reliable. If the relevant date does not fall on a day on which the Stock has traded on such securities exchange or market system, the date on which the Fair Market Value shall be established shall be the last day on which the Stock was so traded prior to the relevant date, or such other appropriate day as shall be determined by the Board, in its sole discretion. If there is then no public market for the Stock, the Fair Market Value on any relevant date shall be as determined by the Board. Notwithstanding the foregoing, the Fair Market Value per share of Stock on the Effective Date shall be deemed to be the public offering

price set forth in the final prospectus filed with the Securities and Exchange Commission in connection with the public offering of the Stock on the Effective Date.

(i) *Offering* means an offering of Stock as provided in Section 6.

(j) Offering Date means, for any Offering, the first day of the Offering Period with respect to such Offering.

(k) *Offering Period* means a period established in accordance with Section 6.1.

(1) *Parent Corporation* means any present or future parent corporation of the Company, as defined in Section 424(e) of the Code.

(m) *Participant* means an Eligible Employee who has become a participant in an Offering Period in accordance with Section 7 and remains a participant in accordance with the Plan.

(n) *Participating Company* means the Company or any Parent Corporation or Subsidiary Corporation designated by the Board as a corporation the Employees of which may, if Eligible Employees, participate in the Plan. The Board shall have the sole and absolute discretion to determine from time to time which Parent Corporations or Subsidiary Corporations shall be Participating Companies. Invitrogen BV, the Company s Netherlands subsidiary shall be a Participating Company unless and until the Board decides otherwise.

(o) *Participating Company Group* means, at any point in time, the Company and all other corporations collectively which are then Participating Companies.

(p) *Purchase Date* means, for any Purchase Period, the last day of such period.

(q) *Purchase Period* means a period, if any, established in accordance with Section 6.2.

(r) Purchase Price means the price at which a share of Stock may be purchased under the Plan, as determined in accordance with Section 9.

(s) *Purchase Right* means an option granted to a Participant pursuant to the Plan to purchase such shares of Stock as provided in Section 8, which the Participant may or may not exercise during the Offering Period in which such option is outstanding. Such option arises from the right of a Participant to withdraw any accumulated payroll deductions of the Participant not previously applied to the purchase of Stock under the Plan and to terminate participation in the Plan at any time during an Offering Period.

(t) Stock means the common stock of the Company, as adjusted from time to time in accordance with Section 4.2.

(u) *Subscription Agreement* means a written agreement in such form as specified by the Company, stating an Employee's election to participate in the Plan and authorizing payroll deductions under the Plan from the Employee's Compensation.

(v) *Subscription Date* means the last business day prior to the Offering Date of an Offering Period or such earlier date as the Company shall establish.

(w) Subsidiary Corporation means any present or future subsidiary corporation of the Company, as defined in Section 424(f) of the Code.

2.2 **Construction.** Captions and titles contained herein are for convenience only and shall not affect the meaning or interpretation of any provision of the Plan. Except when otherwise indicated by the context, the singular shall include the plural and the plural shall include the singular. Use of the term or is not intended to be exclusive, unless the context clearly requires otherwise.

#### 3. <u>Administration.</u>

3.1 **Administration by the Board.** The Plan shall be administered by the Board. All questions of interpretation of the Plan, of any form of agreement or other document employed by the Company in the administration of the Plan, or of any Purchase Right shall be determined by the Board and shall be final and binding upon all persons having an interest in the Plan or the Purchase Right. Subject to the provisions of the Plan, the Board shall determine all of the relevant terms and conditions of Purchase Rights granted pursuant to the Plan; provided, however, that all Participants granted Purchase Rights pursuant to the Plan shall have the same rights and privileges within the meaning of Section 423(b)(5) of the Code. All expenses incurred in connection with the administration of the Plan shall be paid by the Company.

3.2 **Authority of Officers.** Any officer of the Company shall have the authority to act on behalf of the Company with respect to any matter, right, obligation, determination or election that is the responsibility of or that is allocated to the Company herein, provided that the officer has apparent authority with respect to such matter, right, obligation, determination or election.

3.3 **Policies and Procedures Established by the Company.** The Company may, from time to time, consistent with the Plan and the requirements of Section 423 of the Code, establish, change or terminate such rules, guidelines, policies, procedures, limitations, or adjustments as deemed advisable by the Company, in its sole discretion, for the proper administration of the Plan, including, without limitation, (a) a minimum payroll deduction amount required for participation in an Offering, (b) a limitation on the frequency or number of changes permitted in the rate of payroll deduction during an Offering, (c) an exchange ratio applicable to amounts withheld in a currency other than United States dollars, (d) a payroll deduction greater than or less than the amount designated by a Participant in order to adjust for the Company s delay or mistake in processing a Subscription Agreement or in otherwise effecting a Participant s election under the Plan or as advisable to comply with the requirements

of Section 423 of the Code, and (e) determination of the date and manner by which the Fair Market Value of a share of Stock is determined for purposes of administration of the Plan.

#### 4. <u>SHARES SUBJECT</u> TO PLAN.

4.1 **Maximum Number of Shares Issuable.** Subject to adjustment as provided in Section 4.2, the maximum aggregate number of shares of Stock that may be issued under the Plan shall be one million eight hundred fifty thousand (1,850,000) and shall consist of authorized but unissued or reacquired shares of Stock, or any combination thereof. If an outstanding Purchase Right for any reason expires or is terminated or canceled, the shares of Stock allocable to the unexercised portion of such Purchase Right shall again be available for issuance under the Plan.

4.2 Adjustments for Changes in Capital Structure. In the event of any stock dividend, stock split, reverse stock split, recapitalization, combination, reclassification or similar change in the capital structure of the Company, or in the event of any merger (including a merger effected for the purpose of changing the Company s domicile), sale of assets or other reorganization in which the Company is a party, appropriate adjustments shall be made in the number and class of shares subject to the Plan and each Purchase Right and in the Purchase Price. If a majority of the shares which are of the same class as the shares that are subject to outstanding Purchase Rights are exchanged for, converted into, or otherwise become (whether or not pursuant to an Ownership Change Event) shares of another corporation (the New Shares), the Board may unilaterally amend the outstanding Purchase Rights to provide that such Purchase Rights are exercisable for New Shares. In the event of any such amendment, the number of shares subject to, and the Purchase Price of, the outstanding Purchase Rights shall be adjusted in a fair and equitable manner, as determined by the Board, in its sole discretion. Notwithstanding the foregoing, any fractional share resulting from an adjustment pursuant to this Section 4.2 shall be rounded down to the nearest whole number, and in no event may the Purchase Price be decreased to an amount less than the par value, if any, of the stock subject to the Purchase Right. The adjustments determined by the Board pursuant to this Section 4.2 shall be final, binding and conclusive.

#### 5. <u>Eligibility.</u>

5.1 **Employees Eligible to Participate.** Each Employee of a Participating Company is eligible to participate in the Plan and shall be deemed an Eligible Employee as of the first Offering Date following the commencement of his/her service with a Participating Company.

5.2 **Exclusion of Certain Stockholders.** Notwithstanding any provision of the Plan to the contrary, no Employee shall be granted a Purchase Right under the Plan if, immediately after such grant, such Employee would own or hold options to purchase stock of the Company or of any Parent Corporation or Subsidiary Corporation possessing five percent (5%) or more of the total combined voting power or value of all classes of stock of such corporation, as determined in accordance with Section 423(b)(3) of the Code. For purposes of this Section 5.2, the attribution rules of Section 424(d) of the Code shall apply in determining the stock ownership of such Employee.

## 6. <u>Offerings</u>.

6.1 **Offering Periods.** Except as otherwise set forth below, the Plan shall be implemented by sequential Offerings of approximately twenty-four (24) months duration (an **Offering Period**). The first Offering Period shall commence on the Effective Date and end on January 31, 2001. Subsequent Offering Periods shall commence on the first day of February, May, August, and November of each year and end on the last day of the 24th month of such Offering Period. Notwithstanding the foregoing, the Board may establish a different duration for one or more future Offering Periods or different commencing or ending dates for such Offering Periods; provided, however, that no Offering Period may have a duration exceeding twenty-seven (27) months. If the first or last day of an Offering Period is not a day on which the national or regional securities exchange or market system constituting the primary market for the Stock is open for trading, the Company shall specify the trading day that will be deemed the first or last day, as the case may be, of the Offering Period.

6.2 **Purchase Periods.** Generally, each Offering Period will consist of eight (8) quarterly Purchase Periods which begin on the first day of February, May, August and November of each year and end on the last day of the third month of each such Purchase Period (i.e. the following April, July, October and January). The Purchase Period commencing on the Effective Date shall end on the last day of April, 1999. The Board may establish different Purchase Periods which may consist of two (2) or more consecutive Purchase Periods having such duration as the Board shall specify. The last day of each Purchase Period shall be a Purchase Date. If the first or last day of a Purchase Period is not a day on which the national or regional securities exchange or market system constituting the primary market for the Stock is open for trading, the Company shall specify the trading day that will be deemed the first or last day, as the case may be, of the Purchase Period.

#### 7. <u>PARTICIPATION IN THE PLAN</u>.

7.1 **Initial Participation.** An Eligible Employee may become a Participant in an Offering Period by delivering a properly completed Subscription Agreement to the office designated by the Company not later than the close of business for such office on the Subscription Date established by the Company for such Offering Period. An Eligible Employee who does not deliver a properly completed Subscription Agreement to the Company s designated office on or before the Subscription Date for an Offering Period shall not participate in the Plan for that Offering Period or for any subsequent Offering Period unless such Eligible Employee subsequently delivers a properly completed Subscription Agreement to the appropriate office of the Company on or before the Subscription Date for such subsequent Offering Period. An Employee who becomes an Eligible Employee after the Offering Date of an Offering Period shall not be eligible to participate in such Offering Period but may participate in any subsequent Offering Period provided such Employee is still an Eligible Employee as of the Offering Date of such subsequent Offering Period.

7.2 **Continued Participation.** A Participant shall automatically participate in the next Offering Period commencing immediately after the final Purchase Date of each Offering Period in which the Participant participates provided that such Participant remains an Eligible Employee on the Offering Date of the new Offering Period and has not either (a) withdrawn from the Plan pursuant to Section 12.1 or (b) terminated employment as provided in Section 13. A Participant who may automatically participate in a subsequent Offering Period, as provided in this Section, is not required to deliver any additional Subscription Agreement for the subsequent Offering Period in accordance with the procedures set forth in Section 7.1 if the Participant desires to change any of the elections contained in the Participant s then effective Subscription Agreement.

7.3 **One Offering Period per Participant.** A Participant may participate in only one Offering Period at any given time. A Participant s delivery to the Company of a Subscription Agreement for any given Offering Period shall constitute Participant s withdrawal from any concurrent Offering Period and termination of any Purchase Right granted pursuant thereto.

## 8. <u>**RIGHT TO PURCHASE SHARES.</u>**</u>

8.1 **Grant of Purchase Right.** Except as set forth below, on the Offering Date of each Offering Period, each Participant in such Offering Period shall be granted automatically a Purchase Right consisting of an option to purchase the lesser of (a) that number of whole shares of Stock determined by dividing Fifty Thousand Dollars (\$50,000) by the Fair Market Value of a share of Stock on such Offering Date, reduced by the aggregate purchase price of any Stock purchased during any concurrent Offering Period(s) or (b) five thousand (5,000) shares of Stock, reduced by the number of shares of Stock purchased during any concurrent Offering Period(s). No Purchase Right shall be granted on an Offering Date to any person who is not, on such Offering Date, an Eligible Employee.

8.2 **Pro Rata Adjustment of Purchase Right.** Notwithstanding the provisions of Section 8.1, if the Board establishes an Offering Period of any duration other than twenty four months, then (a) the dollar amount in Section 8.1 shall be determined by multiplying \$2,083.33 by the number of months (rounded to the nearest whole month) in the Offering Period and rounding to the nearest whole dollar, and (b) the share amount in Section 8.1 shall be determined by multiplying 208.33 shares by the number of months (rounded to the nearest whole month) in the Offering Period and rounding to the nearest whole month) in the Offering Period and rounding to the nearest whole month) in the Offering Period and rounding to the nearest whole share.

8.3 **Calendar Year Purchase Limitation.** Notwithstanding any provision of the Plan to the contrary, no Participant shall be granted a Purchase Right which permits his or her right to purchase shares of Stock under the Plan to accrue at a rate which, when aggregated with such Participant s rights to purchase shares under all other employee stock purchase plans of a Participating Company intended to meet the requirements of Section 423 of the Code, exceeds Twenty-Five Thousand Dollars (\$25,000) in Fair Market Value (or such other limit, if any, as may be imposed by the Code) for each calendar year in which such Purchase Right is outstanding at any time. For purposes of the preceding sentence, the Fair Market Value of shares purchased

during a given Offering Period shall be determined as of the Offering Date for such Offering Period. The limitation described in this Section 8.3 shall be applied in conformance with applicable regulations under Section 423(b)(8) of the Code.

## 9. <u>PURCHASE PRICE.</u>

The Purchase Price at which each share of Stock may be acquired in an Offering Period upon the exercise of all or any portion of a Purchase Right shall be established by the Board; provided, however, that the Purchase Price shall not be less than eighty-five percent (85%) of the lesser of (a) the Fair Market Value of a share of Stock on the Offering Date of the Offering Period or (b) the Fair Market Value of a share of Stock on the Offering Period or (b) the Fair Market Value of a share of Stock on the Offering Period, the Purchase Price for that Offering Period shall be eighty-five percent (85%) of the lesser of (a) the Fair Market Value of a share of Stock on the Offering Date of the Offering Period, the Purchase Price for that Offering Period, or (b) the Fair Market Value of a share of Stock on the Offering Period, or (b) the Fair Market Value of a share of Stock on the Purchase Date.

#### 10. ACCUMULATION OF PURCHASE PRICE THROUGH PAYROLL DEDUCTION.

Shares of Stock acquired pursuant to the exercise of all or any portion of a Purchase Right may be paid for only by means of payroll deductions from the Participant s Compensation accumulated during the Offering Period for which such Purchase Right was granted, subject to the following:

10.1 **Amount of Payroll Deductions**. Except as otherwise provided herein, the amount to be deducted under the Plan from a Participant s Compensation on each payday during an Offering Period shall be determined by the Participant s Subscription Agreement. The Subscription Agreement shall set forth the percentage of the Participant s Compensation to be deducted on each payday during an Offering Period in whole percentages of not less than one percent (1%) (except as a result of an election pursuant to Section 10.3 to stop payroll deductions made effective following the first payday during an Offering) or more than fifteen percent (15%). Notwithstanding the foregoing, the Board may change the limits on payroll deductions effective as of any future Offering Date.

10.2 **Commencement of Payroll Deductions.** Payroll deductions shall commence on the first payday following the Offering Date and shall continue to the end of the Offering Period unless sooner altered or terminated as provided herein.

10.3 **Election to Change or Stop Payroll Deductions.** During an Offering Period, a Participant may elect to increase or decrease the rate of or to stop deductions from his or her Compensation by delivering to the Company s designated office an amended Subscription Agreement authorizing such change on or before the Change Notice Date. The **Change Notice Date** shall be the date prior to the beginning of the Purchase Period for which such election is to be effective as established by the Company from time to time and announced to the Participants. A Participant who elects to decrease the rate of his or her payroll deductions to zero percent (0%) shall nevertheless remain a Participant in the current Offering Period unless such Participant withdraws from the Plan as provided in Section 12.1.

10.4 Administrative Suspension of Payroll Deductions. The Company may, in its sole discretion, suspend a Participant s payroll deductions under the Plan as the Company deems advisable to avoid accumulating payroll deductions in excess of the amount that could reasonably be anticipated to purchase the maximum number of shares of Stock permitted during a calendar year under the limit set forth in Section 8.3. Payroll deductions shall be resumed at the rate specified in the Participant s then effective Subscription Agreement at the beginning of the next Purchase Period the Purchase Date of which falls in the following calendar year.

10.5 **Participant Accounts.** Individual bookkeeping accounts shall be maintained for each Participant. All payroll deductions from a Participant s Compensation shall be credited to such Participant s Plan account and shall be deposited with the general funds of the Company. All payroll deductions received or held by the Company may be used by the Company for any corporate purpose.

10.6 No Interest Paid. Interest shall not be paid on sums deducted from a Participant s Compensation pursuant to the Plan.

10.7 **Voluntary Withdrawal from Plan Account.** A Participant may withdraw all or any portion of the payroll deductions credited to his or her Plan account and not previously applied toward the purchase of Stock by delivering to the Company's designated office a written notice on a form provided by the Company for such purpose. A Participant who withdraws the entire remaining balance credited to his or her Plan account shall be deemed to have withdrawn from the Plan in accordance with Section 12.1. Amounts withdrawn shall be returned to the Participant as soon as practicable after the withdrawal and may not be applied to the purchase of shares in any Offering under the Plan. The Company may from time to time establish or change limitations on the frequency of withdrawals permitted under this Section, establish a minimum dollar amount that must be retained in the Participant's Plan account, or terminate the withdrawal right provided by this Section.

## 11. <u>Purchase of Shares.</u>

11.1 Exercise of Purchase Right. On each Purchase Date of an Offering Period, each Participant who has not withdrawn from the Plan and whose participation in the Offering has not terminated before such Purchase Date shall automatically acquire pursuant to the exercise of the Participant s Purchase Right the number of whole shares of Stock determined by dividing (a) the total amount of the Participant s payroll deductions accumulated in the Participant s Plan account during the Offering Period and not previously applied toward the purchase of Stock by (b) the Purchase Price. However, in no event shall the number of shares purchased by the Participant during an Offering Period exceed the number of shares subject to the Participant s Purchase Right. No shares of Stock shall be purchase Date on behalf of a Participant whose participation in the Offering or the Plan has terminated before such Purchase Date.

11.2 **Pro Rata Allocation of Shares.** In the event that the number of shares of Stock which might be purchased by all Participants in the Plan on a Purchase Date exceeds the number of shares of Stock available in the Plan as provided in Section 4.1, the Company shall

make a pro rata allocation of the remaining shares in as uniform a manner as shall be practicable and as the Company shall determine to be equitable. Any fractional share resulting from such pro rata allocation to any Participant shall be disregarded.

11.3 **Delivery of Certificates.** As soon as practicable after each Purchase Date, the Company shall arrange the delivery to each Participant, as appropriate, of a certificate representing the shares acquired by the Participant on such Purchase Date; provided that the Company may deliver such shares to a broker that holds such shares in street name for the benefit of the Participant. Shares to be delivered to a Participant under the Plan shall be registered in the name of the Participant, or, if requested by the Participant, in the name of the Participant and his or her spouse, or, if applicable, in the names of the heirs of the Participant.

11.4 **Return of Cash Balance.** Any cash balance remaining in a Participant's Plan account following any Purchase Date shall be refunded to the Participant as soon as practicable after such Purchase Date. However, if the cash to be returned to a Participant pursuant to the preceding sentence is an amount less than the amount that would have been necessary to purchase an additional whole share of Stock on such Purchase Date, the Company may retain such amount in the Participant's Plan account to be applied toward the purchase of shares of Stock in the subsequent Purchase Period or Offering Period, as the case may be.

11.5 **Tax Withholding.** At the time a Participant s Purchase Right is exercised, in whole or in part, or at the time a Participant disposes of some or all of the shares of Stock he or she acquires under the Plan, the Participant shall make adequate provision for the foreign, federal, state and local tax withholding obligations of the Participating Company Group, if any, which arise upon exercise of the Purchase Right or upon such disposition of shares, respectively. The Participating Company Group may, but shall not be obligated to, withhold from the Participant s compensation the amount necessary to meet such withholding obligations.

11.6 **Expiration of Purchase Right.** Any portion of a Participant s Purchase Right remaining unexercised after the end of the Offering Period to which the Purchase Right relates shall expire immediately upon the end of the Offering Period.

11.7 **Reports to Participants.** Each Participant who has exercised all or part of his or her Purchase Right shall receive, as soon as practicable after the Purchase Date, a report of such Participant s Plan account setting forth the total payroll deductions accumulated prior to such exercise, the number of shares of Stock purchased, the Purchase Price for such shares, the date of purchase and the cash balance, if any, remaining immediately after such purchase that is to be refunded or retained in the Participant s Plan account pursuant to Section 11.4. The report required by this Section may be delivered in such form and by such means, including by electronic transmission, as the Company may determine.

## 12. WITHDRAWAL FROM THE PLAN.

12.1 **Voluntary Withdrawal from the Plan.** A Participant may withdraw from the Plan by signing and delivering to the Company s designated office a written notice of withdrawal on a form provided by the Company for such purpose. Such withdrawal may be

elected at any time prior to the end of an Offering Period. A Participant who voluntarily withdraws from the Plan is prohibited from resuming participation in the Plan in the same Offering from which he or she withdrew, but may participate in any subsequent Offering by again satisfying the requirements of Sections 5 and 7.1. The Company may impose, from time to time, a requirement that the notice of withdrawal from the Plan be on file with the Company s designated office for a reasonable period prior to the effectiveness of the Participant s withdrawal.

12.2 **Return of Payroll Deductions.** Upon a Participant s voluntary withdrawal from the Plan pursuant to Sections 12.1 or 10.7 or an automatic withdrawal pursuant to Section 12.3, the Participant s accumulated payroll deductions which have not been applied toward the purchase of shares of Stock shall be refunded to the Participant as soon as practicable after the withdrawal, without the payment of any interest, and the Participant s interest in the Plan shall terminate. Such accumulated payroll deductions to be refunded in accordance with this Section may not be applied to any other Offering under the Plan.

12.3 **Automatic Withdrawal from an Offering.** If the Fair Market Value of a share of Stock on a Purchase Date (other than the final Purchase Date of an Offering Period) is less that the Fair Market Value of a share of Stock on the Offering Date for such Offering Period, then every Participant shall automatically be (a) withdrawn from such Offering Period after the acquisition of shares of Stock on the Purchase Date and (b) enrolled in the new Offering Period effective on its Offering Date. A Participant may elect not to be automatically withdrawn from an Offering Period pursuant to this Section 12.2 by delivering to the Company s designated office not later than the close of business on the Offering Date of the new Offering Period a written notice indicating such decision.

## 13. <u>TERMINATION OF EMPLOYMENT OR ELIGIBILITY.</u>

Upon a Participant s ceasing, prior to a Purchase Date, to be an Employee of the Participating Company Group for any reason, including retirement, disability or death, or the failure of a Participant to remain an Eligible Employee, the Participant s participation in the Plan shall terminate immediately. In such event, the payroll deductions credited to the Participant s Plan account since the last Purchase Date shall, as soon as practicable, be returned to the Participant or, in the case of the Participant s death, to the Participant s legal representative, and all of the Participant s rights under the Plan shall terminate. Interest shall not be paid on sums returned pursuant to this Section 13. A Participant whose participation has been so terminated may again become eligible to participate in the Plan by again satisfying the requirements of Sections 5 and 7.1.

#### 14. <u>CHANGE IN CONTROL.</u> 14.1 **Definitions**.

(a) An *Ownership Change Event* shall be deemed to have occurred if any of the following occurs with respect to the Company: (i) the direct or indirect sale or exchange in a single or series of related transactions by the stockholders of the Company of more

than fifty percent (50%) of the voting stock of the Company; (ii) a merger or consolidation in which the Company is a party; (iii) the sale, exchange, or transfer of all or substantially all of the assets of the Company; or (iv) a liquidation or dissolution of the Company.

(b) A *Change in Control* shall mean an Ownership Change Event or a series of related Ownership Change Events (collectively, the **Transaction**) wherein the stockholders of the Company immediately before the Transaction do not retain immediately after the Transaction, in substantially the same proportions as their ownership of shares of the Company s voting stock immediately before the Transaction, direct or indirect beneficial ownership of more than fifty percent (50%) of the total combined voting power of the outstanding voting stock of the Company or the corporation or corporations to which the assets of the Company were transferred (the **Transferee Corporation(s)**), as the case may be. For purposes of the preceding sentence, indirect beneficial ownership shall include, without limitation, an interest resulting from ownership of the voting stock of one or more corporations which, as a result of the Transaction, own the Company or the Transferee Corporation(s), as the case may be, either directly or through one or more subsidiary corporations. The Board shall have the right to determine whether multiple sales or exchanges of the voting stock of the Company or multiple Ownership Change Events are related, and its determination shall be final, binding and conclusive.

14.2 Effect of Change in Control on Purchase Rights. In the event of a Change in Control, the surviving, continuing, successor, or purchasing corporation or parent corporation thereof, as the case may be (the Acquiring Corporation), may assume the Company's rights and obligations under the Plan. If the Acquiring Corporation elects not to assume the Company's rights and obligations under outstanding Purchase Rights, the Purchase Date of the then current Offering Period (or Purchase Period) shall be accelerated to a date before the date of the Change in Control specified by the Board, but the number of shares of Stock subject to outstanding Purchase Rights shall not be adjusted. All Purchase Rights which are neither assumed by the Acquiring Corporation in connection with the Change in Control nor exercised as of the date of the Change in Control shall terminate and cease to be outstanding effective as of the date of the Change in Control.

## 15. <u>Nontransferability of Purchase Rights.</u>

A Purchase Right may not be transferred in any manner otherwise than by will or the laws of descent and distribution and shall be exercisable during the lifetime of the Participant only by the Participant.

## 16. <u>COMPLIANCE WITH SECURITIES LAW.</u>

The issuance of shares under the Plan shall be subject to compliance with all applicable requirements of federal, state and foreign law with respect to such securities. A Purchase Right may not be exercised if the issuance of shares upon such exercise would constitute a violation of any applicable federal, state or foreign securities laws or other law or regulations or the requirements of any securities exchange or market system upon which the Stock may then be listed. In addition, no Purchase Right may be exercised unless (a) a

registration statement under the Securities Act of 1933, as amended, shall at the time of exercise of the Purchase Right be in effect with respect to the shares issuable upon exercise of the Purchase Right, or (b) in the opinion of legal counsel to the Company, the shares issuable upon exercise of the Purchase Right may be issued in accordance with the terms of an applicable exemption from the registration requirements of said Act. The inability of the Company to obtain from any regulatory body having jurisdiction the authority, if any, deemed by the Company s legal counsel to be necessary to the lawful issuance and sale of any shares under the Plan shall relieve the Company of any liability in respect of the failure to issue or sell such shares as to which such requisite authority shall not have been obtained. As a condition to the exercise of a Purchase Right, the Company may require the Participant to satisfy any qualifications that may be necessary or appropriate, to evidence compliance with any applicable law or regulation, and to make any representation or warranty with respect thereto as may be requested by the Company.

#### 17. <u>Rights as a Stockholder and Employee.</u>

A Participant shall have no rights as a stockholder by virtue of the Participant s participation in the Plan until the date of the issuance of a certificate for the shares purchased pursuant to the exercise of the Participant s Purchase Right (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company). No adjustment shall be made for dividends, distributions or other rights for which the record date is prior to the date such certificate is issued, except as provided in Section 4.2. Nothing herein shall confer upon a Participant any right to continue in the employ of the Participating Company Group or interfere in any way with any right of the Participating Company Group to terminate the Participant s employment at any time.

#### 18. <u>Legends.</u>

The Company may at any time place legends or other identifying symbols referencing any applicable federal, state or foreign securities law restrictions or any provision convenient in the administration of the Plan on some or all of the certificates representing shares of Stock issued under the Plan. The Participant shall, at the request of the Company, promptly present to the Company any and all certificates representing shares acquired pursuant to a Purchase Right in the possession of the Participant in order to carry out the provisions of this Section.

## 19. NOTIFICATION OF SALE OF SHARES.

The Company may require the Participant to give the Company prompt notice of any disposition of shares acquired by exercise of a Purchase Right within two years from the date of granting such Purchase Right or one year from the date of exercise of such Purchase Right. The Company may require that until such time as a Participant disposes of shares acquired upon exercise of a Purchase Right, the Participant shall hold all such shares in the Participant share (or, if elected by the Participant, in the name of the Participant and his or her spouse but not in the name of any nominee) until the lapse of the time periods with respect to such Purchase Right referred to in the preceding sentence. The Company may direct that the certificates evidencing shares acquired by exercise of a Purchase Right refer to such requirement to give prompt notice of disposition.

#### 20. <u>Notices.</u>

All notices or other communications by a Participant to the Company under or in connection with the Plan shall be deemed to have been duly given when received in the form specified by the Company at the location, or by the person, designated by the Company for the receipt thereof.

#### 21. <u>INDEMNIFICATION.</u>

In addition to such other rights of indemnification as they may have as members of the Board or officers or employees of the Participating Company Group, members of the Board and any officers or employees of the Participating Company Group to whom authority to act for the Board or the Company is delegated shall be indemnified by the Company against all reasonable expenses, including attorneys fees, actually and necessarily incurred in connection with the defense of any action, suit or proceeding, or in connection with any appeal therein, to which they or any of them may be a party by reason of any action taken or failure to act under or in connection with the Plan, or any right granted hereunder, and against all amounts paid by them in settlement thereof (provided such settlement is approved by independent legal counsel selected by the Company) or paid by them in satisfaction of a judgment in any such action, suit or proceeding, except in relation to matters as to which it shall be adjudged in such action, suit or proceeding that such person is liable for gross negligence, bad faith or intentional misconduct in duties; provided, however, that within sixty (60) days after the institution of such action, suit or proceeding, such person shall offer to the Company, in writing, the opportunity at its own expense to handle and defend the same.

## 22. <u>Amendment or Termination of the Plan.</u>

The Board may at any time amend or terminate the Plan, except that (a) such termination shall not affect Purchase Rights previously granted under the Plan, except as permitted under the Plan, provided however that the Board may terminate the Plan (and any Offerings and future Purchase Rights) on any Purchase Date if the Board determines that such termination is in the best interests of the Company and its stockholders; and (b) no amendment may adversely affect a Purchase Right previously granted under the Plan (except to the extent permitted by the Plan or as may be necessary to qualify the Plan as an employee stock purchase plan pursuant to Section 423 of the Code or to obtain qualification or registration of the shares of Stock under applicable federal, state or foreign securities laws). In addition, an amendment to the Plan must be approved by the stockholders of the Company within twelve (12) months of the adoption of such amendment if such amendment would authorize the sale of more shares than are authorized for issuance under the Plan or would change the definition of the corporations that may be designated by the Board as Participating Companies. In the event that the Board approves an amendment to increase the number of shares authorized for issuance under the Plan (the *Additional Shares*), the Board, in its sole discretion, may specify that such Additional Shares may only be issued pursuant to Purchase Rights granted after the date on which the

stockholders of the Company approve such amendment, and such designation by the Board shall not be deemed to have adversely affected any Purchase Right granted prior to the date on which the stockholders approve the amendment.

IN WITNESS WHEREOF, the undersigned Secretary of the Company certifies that the foregoing Invitrogen Corporation 1998 Employee Stock Purchase Plan was duly adopted by the Board of Directors of the Company on April 21, 2006.

John A. Cottingham

## PLAN HISTORY

November 20, 1998	Board adopts the Plan, with an initial reserve of 250,000 shares.
January 15, 1999	Stockholders approve Plan, with an initial reserve of 250,000 shares.
February 28, 2000	Board amends the Plan to increase the share reserve by 100,000 to a total of 350,000 shares.
April 27, 2000	Stockholders approve the share reserve increase to a total of 350,000 shares.
August 8, 2000	Board amends the Plan to increase the share reserve by 200,000 to a total of 550,000 shares.
September 14, 2000	Stockholders approve the share reserve increase to a total of 550,000 shares.
February 27, 2002	Board amends the Plan to increase the share reserve by 300,000 to a total of 850,000 shares.
May 23, 2002	Stockholders approve the share reserve increase to a total of 850,000 shares.
February 20, 2006	Board amends the Plan to increase the share reserve by 1,000,000 to a total of 1,850,000 shares.
April 21, 2006	Stockholders approve the share reserve increase to a total of 1,850,000 shares.

Appendix B

## INVITROGEN CORPORATION

## 2004 EQUITY INCENTIVE PLAN

## TABLE OF CONTENTS

1.	Establishment, Purpose and Term of Plan 1.1 Establishment 1.2 Purpose 1.3 Term of Plan	Page B-1 B-1 B-1 B-1 B-1
2.	Definitions and Construction 2.1 Definitions 2.2 Construction	B-1 B-1 B-4
3.	Administration3.1Administration by the Committee3.2Authority of Officers3.3Administration with Respect to Insiders3.4Committee Complying with Section 162(m)3.5Powers of the Committee3.6No Repricing3.7Indemnification	B-4 B-4 B-5 B-5 B-5 B-6 B-6 B-6
4.	<ul> <li>Shares Subject to Plan</li> <li>4.1 Maximum Number of Shares Issuable</li> <li>4.2 Adjustments for Changes in Capital Structure</li> </ul>	B-6 B-6 B-6
5.	<ul> <li>Eligibility and Award Limitations</li> <li>5.1 Persons Eligible for Awards</li> <li>5.2 Participation</li> <li>5.3 Incentive Stock Option Limitations</li> <li>5.4 Section 162(m) Award Limits</li> </ul>	B-7 B-7 B-7 B-7 B-7 B-7
6.	<ul> <li>Terms and Conditions of Options</li> <li>6.1 Exercise Price</li> <li>6.2 Exercisability and Term of Options</li> <li>6.3 Payment of Exercise Price</li> <li>6.4 Effect of Termination of Service</li> <li>6.5 Transferability of Options</li> </ul>	B-8 B-8 B-8 B-8 B-9 B-9 B-9
7.	<ul> <li>Terms and Conditions of Stock Appreciation Rights</li> <li>7.1 Types of SARs Authorized</li> <li>7.2 Exercise Price</li> <li>7.3 Exercisability and Term of SARs</li> <li>7.4 Exercise of SARs</li> <li>7.5 Deemed Exercise of SARs</li> <li>7.6 Effect of Termination of Service</li> <li>7.7 Nontransferability of SARs</li> </ul>	B-9 B-9 B-9 B-9 B-10 B-10 B-10 B-10
8.	<ul> <li>Terms and Conditions of Restricted Stock Awards</li> <li>8.1 Types of Restricted Stock Awards Authorized</li> <li>8.2 Purchase Price</li> <li>8.3 Purchase Period</li> <li>8.4 Payment of Purchase Price</li> <li>8.5 Vesting and Restrictions on Transfer</li> <li>8.6 Voting Rights; Dividends and Distributions</li> <li>8.7 Effect of Termination of Service</li> <li>8.8 Nontransferability of Restricted Stock Award Rights</li> <li>8.9 Minimum Vesting Period</li> </ul>	B-10 B-10 B-10 B-10 B-10 B-11 B-11 B-11

0	Transford Conditions of Defermines Annuals	Page
9.	Terms and Conditions of Performance Awards 9.1 Types of Performance Awards Authorized	B-11 B-11
	9.2 Initial Value of Performance Shares and Performance Units	B-11
	9.3 Establishment of Performance Period, Performance Goals and Performance Award Formula	B-11
	9.4 Measurement of Performance Goals	B-12
	9.5 Settlement of Performance Awards	B-13
	9.6 Voting Rights; Dividend Equivalent Rights and Distributions	B-13
	9.7 Effect of Termination of Service	B-14
	9.8 Nontransferability of Performance Awards	B-14
10.	Terms and Conditions of Restricted Stock Unit Awards	B-14
	10.1 Grant of Restricted Stock Unit Awards	B-14
	10.2 Purchase Price	B-14
	10.3 Vesting 10.4 Vesting Pictus Dividend Equivalent Pictus and Distributions	B-14 B-14
	<ul><li>10.4 Voting Rights, Dividend Equivalent Rights and Distributions</li><li>10.5 Effect of Termination of Service</li></ul>	B-14 B-15
	10.6 Settlement of Restricted Stock Unit Awards	B-15 B-15
	10.7 Nontransferability of Restricted Stock Unit Awards	B-15 B-15
11.	Deferred Stock Units	B-15
	11.1 Establishment of Deferred Stock Unit Program	B-15
	11.2 Terms and Conditions of Deferred Stock Units	B-15
12.	Standard Forms of Award Agreement	B-16
	12.1 Award Agreements	B-16
	12.2 Authority to Vary Terms	B-16
13.	8	B-16
	13.1 Definitions	B-16
	13.2 Effect of Change in Control on Options and SARs	B-17
	13.3 Effect of Change in Control on Restricted Stock Awards	B-17
	<ul> <li>13.4 Effect of Change in Control on Performance Awards</li> <li>13.5 Effect of Change in Control on Protected Stock Unit Awards</li> </ul>	B-17 B-17
	<ul><li>13.5 Effect of Change in Control on Restricted Stock Unit Awards</li><li>13.6 Effect of Change in Control on Deferred Stock Units</li></ul>	B-17 B-18
14.	-	B-18
	Compliance with Securities Law	
15.	Tax Withholding	B-18
	<ul><li>15.1 Tax Withholding in General</li><li>15.2 Withholding in Shares</li></ul>	B-18 B-18
16		
16.	Amendment or Termination of Plan	B-18
17.		B-18
	17.1 Repurchase Rights	B-18
	17.2 Provision of Information	B-19
	17.3 Rights as Employee, Consultant or Director	B-19
	<ul><li>17.4 Rights as a Stockholder</li><li>17.5 Fractional Shares</li></ul>	B-19 B-19
	17.5 Fractional Shares 17.6 Severability	B-19 B-19
	17.6 Severability 17.7 Beneficiary Designation	В-19 В-19
	17.7 Designation 17.8 Unfunded Obligation	B-19 B-19
	17.9 Section 409A of the Code	B-19

## INVITROGEN CORPORATION

#### 2004 EQUITY INCENTIVE PLAN

#### 1. <u>Establishment, Purpose and Term of Plan</u>.

1.1 Establishment. The Invitrogen Corporation 2004 Equity Incentive Plan (the *Plan*) originally established effective as of April 29, 2004, the date of its initial approval by the stockholders of the Company (the *Effective Date*), is hereby amended and restated effective as of April 30, 2008.

1.2 **Purpose**. The purpose of the Plan is to advance the interests of the Participating Company Group and its stockholders by providing an incentive to attract, retain and reward persons performing services for the Participating Company Group and by motivating such persons to contribute to the growth and profitability of the Participating Company Group. The Plan seeks to achieve this purpose by providing for Awards in the form of Options, Indexed Options, Stock Appreciation Rights, Restricted Stock Purchase Rights, Restricted Stock Bonuses, Performance Shares, Performance Units, Restricted Stock Units and Deferred Stock Units. After the Effective Date, the Company shall terminate, and no longer issue any awards from under, the Company s 1997 Stock Option Plan, 2000 Nonstatutory Stock Option Plan, 2001 Stock Incentive Plan (formerly the Molecular Probes, Inc. 2001 Stock Incentive Plan) or the 2002 Stock Incentive Plan (formerly the Molecular Probes, Inc. 2002 Stock Incentive Plan).

1.3 **Term of Plan**. The Plan shall continue in effect until the earlier of its termination by the Board or the date on which all of the shares of Stock available for issuance under the Plan have been issued and all restrictions on such shares under the terms of the Plan and the agreements evidencing Awards granted under the Plan have lapsed. However, all Incentive Stock Options shall be granted, if at all, within ten (10) years from the Effective Date.

## 2. <u>Definitions and Construction</u>.

2.1 **Definitions**. Whenever used herein, the following terms shall have their respective meanings set forth below:

(a) *Affiliate* means (i) an entity, other than a Parent Corporation, that directly, or indirectly through one or more intermediary entities, controls the Company or (ii) an entity, other than a Subsidiary Corporation, that is controlled by the Company directly, or indirectly through one or more intermediary entities. For this purpose, the term control (including the term controlled by ) means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of the relevant entity, whether through the ownership of voting securities, by contract or otherwise; or shall have such other meaning assigned such term for the purposes of registration on Form S-8 under the Securities Act.

(b) *Award* means any Option, Indexed Option, SAR, Restricted Stock Purchase Right, Restricted Stock Bonus, Performance Share, Performance Unit, Restricted Stock Unit or Deferred Stock Unit granted under the Plan.

(c) Award Agreement means a written agreement between the Company and a Participant setting forth the terms, conditions and restrictions of the Award granted to the Participant. An Award Agreement may be an Option Agreement, an Indexed Option Agreement, a SAR Agreement, a Restricted Stock Purchase Agreement, a Restricted Stock Bonus Agreement, a Performance Share Agreement, a Performance Unit Agreement, Restricted Stock Unit Agreement, or a Deferred Stock Unit Agreement.

(d) *Board* means the Board of Directors of the Company.

(e) *Code* means the Internal Revenue Code of 1986, as amended, and any applicable regulations promulgated thereunder.

(f) *Committee* means the Compensation and Organization Committee or other committee of the Board duly appointed to administer the Plan and having such powers as shall be specified by the Board. If no committee of the Board has been appointed to administer the Plan, the Board shall exercise all of the powers of the Committee granted herein, and, in any event, the Board may in its discretion exercise any or all of such powers.

(g) *Company* means Invitrogen Corporation, a Delaware corporation, or any successor corporation thereto.

(h) *Consultant* means a person engaged to provide consulting or advisory services (other than as an Employee or a member of the Board) to a Participating Company, provided that the identity of such person, the nature of such services or the entity to which such services are provided would not preclude the Company from offering or selling securities to such person pursuant to the Plan in reliance on a Form S-8 Registration

Statement under the Securities Act.

B-1

(i) *Deferred Stock Unit* means a bookkeeping entry representing a right granted to a Participant pursuant to Section 11 of the Plan to receive a share of Stock on a date determined in accordance with the provisions of Section 11 and the Participant s Award Agreement.

(j) *Director* means a member of the Board.

(k) *Disability* means the permanent and total disability of the Participant, within the meaning of Section 22(e)(3) of the Code.

(1) *Dividend Equivalent* means a credit, made at the discretion of the Committee or as otherwise provided by the Plan, to the account of a Participant in an amount equal to the cash dividends paid on one share of Stock for each share of Stock represented by an Award held by such Participant.

(m) *Employee* means any person treated as an employee (including an Officer or a Director who is also treated as an employee) in the records of a Participating Company and, with respect to any Incentive Stock Option granted to such person, who is an employee for purposes of Section 422 of the Code; provided, however, that neither service as a Director nor payment of a Director s fee shall be sufficient to constitute employment for purposes of the Plan. The Company shall determine in good faith and in the exercise of its discretion whether an individual has become or has ceased to be an Employee and the effective date of such individual s employment or termination of employment, as the case may be. For purposes of an individual s rights, if any, under the Plan as of the time of the Company s determination, all such determinations by the Company shall be final, binding and conclusive, notwithstanding that the Company or any court of law or governmental agency subsequently makes a contrary determination.

(n) *Exchange Act* means the Securities Exchange Act of 1934, as amended.

(o) *Fair Market Value* means, as of any date, the value of a share of Stock or other property as determined by the Committee, in its discretion, or by the Company, in its discretion, if such determination is expressly allocated to the Company herein, subject to the following:

(i) If, on such date, the Stock is listed on a national or regional securities exchange or market system, the Fair Market Value of a share of Stock shall be the closing price of a share of Stock (or the mean of the closing bid and asked prices of a share of Stock if the Stock is so quoted instead) as quoted on the Nasdaq Global Select Market, the Nasdaq Global Market, the Nasdaq Capital Market or such other national or regional securities exchange or market system constituting the primary market for the Stock, as reported in *The Wall Street Journal* or such other source as the Company deems reliable. If the relevant date does not fall on a day on which the Stock has traded on such securities exchange or market system, the date on which the Fair Market Value shall be established shall be the last day on which the Stock was so traded prior to the relevant date, or such other appropriate day as shall be determined by the Committee, in its discretion.

(ii) If, on such date, the Stock is not listed on a national or regional securities exchange or market system, the Fair Market Value of a share of Stock shall be as determined by the Committee in good faith without regard to any restriction other than a restriction which, by its terms, will never lapse.

(p) *Incentive Stock Option* means an Option intended to be (as set forth in the Award Agreement) and which qualifies as an incentive stock option within the meaning of Section 422(b) of the Code.

(q) *Indexed Option* means an Option with an exercise price which either increases by a fixed percentage over time or changes by reference to a published index, as determined by the Committee and set forth in the Option Agreement.

(r) Insider means an Officer, a Director or any other person whose transactions in Stock are subject to Section 16 of the Exchange Act.

(s) *Nonstatutory Stock Option* means an Option not intended to be (as set forth in the Award Agreement) an incentive stock option within the meaning of Section 422(b) of the Code.

(t) *Officer* means any person designated by the Board as an officer of the Company.

(u) *Option* means the right to purchase Stock at a stated price for a specified period of time granted to a Participant pursuant to Section 6 of the Plan. An Option may be either an Incentive Stock Option, a Nonstatutory Stock Option or an Indexed Option.

(v) Parent Corporation means any present or future parent corporation of the Company, as defined in Section 424(e) of the Code.

(w) *Participant* means any eligible person who has been granted one or more Awards.

(x) Participating Company means the Company or any Parent Corporation, Subsidiary Corporation or Affiliate.

(y) Participating Company Group means, at any point in time, all entities collectively which are then Participating Companies.

(z) Performance Award means an Award of Performance Shares or Performance Units.

(aa) *Performance Award Formula* means, for any Performance Award, a formula or table established by the Committee pursuant to Section 9.3 of the Plan which provides the basis for computing the value of a Performance Award at one or more threshold levels of attainment of the applicable Performance Goal(s) measured as of the end of the applicable Performance Period.

(bb) *Performance Goal* means a performance goal established by the Committee pursuant to Section 9.3 of the Plan.

(cc) *Performance Period* means a period established by the Committee pursuant to Section 9.3 of the Plan at the end of which one or more Performance Goals are to be measured.

(dd) *Performance Share* means a bookkeeping entry representing a right granted to a Participant pursuant to Section 9 of the Plan to receive a payment equal to the value of a Performance Share, as determined by the Committee, based on performance.

(ee) *Performance Unit* means a bookkeeping entry representing a right granted to a Participant pursuant to Section 9 of the Plan to receive a payment equal to the value of a Performance Unit, as determined by the Committee, based upon performance.

(ff) *Prior Plan Options* means any option or other award granted by the Company which is subject to vesting or repurchase by the Company, including specifically, all such options and awards granted pursuant to the Company s 1997 Stock Option Plan, 2000 Nonstatutory Stock Option Plan, 2001 Stock Incentive Plan (formerly the Molecular Probes, Inc. 2001 Stock Incentive Plan) or the 2002 Stock Incentive Plan (formerly the Molecular Probes, Inc. 2001 Stock Incentive Plan) or the Effective Date.

(gg) Restricted Stock Award means an Award of a Restricted Stock Bonus or a Restricted Stock Purchase Right.

(hh) Restricted Stock Bonus means Stock granted to a Participant pursuant to Section 8 of the Plan.

(ii) Restricted Stock Purchase Right means a right to purchase Stock granted to a Participant pursuant to Section 8 of the Plan.

(jj) *Restricted Stock Unit* or *Stock Unit* means a bookkeeping entry representing a right granted to a Participant pursuant to Section 10 of the Plan to receive a share of Stock on a date determined in accordance with the provisions of Section 10 and the Participant s Award Agreement.

(kk) *Restriction Period* means the period established in accordance with Section 8.5 of the Plan during which shares subject to a Restricted Stock Award are subject to Vesting Conditions.

(ll) **Retirement** means a Participant s termination of Service, if as of the date of such termination, the Participant has reached the age of fifty-eight (58) and has completed eight (8) years of continuous Service to the Participating Company Group. A Participant who terminates Service with the Participating Company Group and resumes Service more than six (6) months after his or her original termination date, will not have his or her Service with the Participant Company Group prior to his or her original termination date count for purposes of determining Retirement. Notwithstanding the foregoing, the Board shall have the discretion to determine on a case by case basis whether such prior Service with the Participant Company Group may be counted for purposes of Retirement. The Board will notify any rehired Participant if the Board has determined such prior Service will count towards Retirement, and in the absence of such notification from the Board, such Service shall not be counted for purposes of Retirement.

(mm) Rule 16b-3 means Rule 16b-3 under the Exchange Act, as amended from time to time, or any successor rule or regulation.

(nn) **SAR** or **Stock Appreciation Right** means a bookkeeping entry representing, for each share of Stock subject to such SAR, a right granted to a Participant pursuant to Section 7 of the Plan to receive payment of an amount equal to the excess, if any, of the Fair Market Value of a share of Stock on the date of exercise of the SAR over the exercise price.

(oo) Section 162(m) means Section 162(m) of the Code.

(pp) Securities Act means the Securities Act of 1933, as amended.

(qq) *Service* means a Participant s employment or service with the Participating Company Group, whether in the capacity of an Employee, a Director or a Consultant. A Participant s Service shall not be deemed to have terminated merely because of a change in the capacity in which the Participant renders such Service or a change in the Participating Company for which the Participant renders such Service, provided that there is no interruption or termination of the Participant s Service. Furthermore, a Participant s Service shall not be deemed to have terminated if the Participant takes any military leave, sick leave, or other bona fide leave of absence approved by the Company. However, if any such leave taken by a Participant exceeds ninety (90) days, then on the one hundred eighty-first (181st) day following the commencement of such leave any Incentive Stock Option held by the Participant s right to return to Service with the Participanting Company Group is guaranteed by statute or contract. Notwithstanding the foregoing, unless otherwise designated by the Company or required by law, a leave of absence shall not be treated as Service for purposes of determining vesting under the Participant s Award Agreement. A Participant s Service shall be deemed to have terminated either upon an actual termination of Service or upon the entity for which the Participant performs Service ceasing to be a Participating Company. Subject to the foregoing, the Company, in its discretion, shall determine whether the Participant s Service has terminated and the effective date of such termination.

(rr) *Stock* means the common stock of the Company, as adjusted from time to time in accordance with Section 4.2 of the Plan.

(ss) Subsidiary Corporation means any present or future subsidiary corporation of the Company, as defined in Section 424(f) of the Code.

(tt) *Ten Percent Owner* means a Participant who, at the time an Incentive Stock Option is granted to the Participant, owns stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of a Participating Company (other than an Affiliate) within the meaning of Section 422(b)(6) of the Code.

(uu) *Vesting Conditions* mean those conditions established in accordance with Section 8.5 or Section 10.3 of the Plan prior to the satisfaction of which shares subject to a Restricted Stock Award or Restricted Stock Unit Award, respectively, remain subject to forfeiture or a repurchase option in favor of the Company upon the Participant s termination of Service. For example, any Award made pursuant to this Plan shall have a minimum vesting period of one year.

2.2 **Construction**. Captions and titles contained herein are for convenience only and shall not affect the meaning or interpretation of any provision of the Plan. Except when otherwise indicated by the context, the singular shall include the plural and the plural shall include the singular. Use of the term or is not intended to be exclusive, unless the context clearly requires otherwise.

#### 3. <u>Administration</u>.

3.1 Administration by the Committee. The Plan shall be administered by the Committee. All questions of interpretation of the Plan or of any Award shall be determined by the Committee, and such determinations shall be final and binding upon all persons having an interest in the Plan or such Award.

3.2 **Authority of Officers**. Any Officer shall have the authority to act on behalf of the Company with respect to any matter, right, obligation, determination or election which is the responsibility of or which is allocated to the Company herein, provided the Officer has apparent authority with respect to such matter, right, obligation, determination or election. The Board may, in its discretion, delegate to a committee comprised of one or more Officers the authority to grant one or more Awards, without further approval of the Board or the Committee, to any Employee, other than a person who, at the time of

such grant, is an Insider; provided, however, that (a) such Awards shall not be granted for shares in excess of the maximum aggregate number of shares of Stock authorized for issuance pursuant to Section 4.1, (b) the exercise price per share of each Option shall be not less than the Fair Market Value per share of the Stock on the effective date of grant (or, if the Stock has not traded on such date, on the last day preceding the effective date of grant on which the Stock was traded), and (iii) each such Award shall be subject to the terms and conditions of the appropriate standard form of Award Agreement approved by the Board or the Committee and shall conform to the provisions of the Plan and such other guidelines as shall be established from time to time by the Board or the Committee.

3.3 Administration with Respect to Insiders. With respect to participation by Insiders in the Plan, at any time that any class of equity security of the Company is registered pursuant to Section 12 of the Exchange Act, the Plan shall be administered in compliance with the requirements, if any, of Rule 16b-3.

3.4 **Committee Complying with Section 162(m)**. If the Company is a publicly held corporation within the meaning of Section 162(m), the Board may establish a Committee of outside directors within the meaning of Section 162(m) to approve the grant of any Award which might reasonably be anticipated to result in the payment of employee remuneration that would otherwise exceed the limit on employee remuneration deductible for income tax purposes pursuant to Section 162(m).

3.5 **Powers of the Committee**. In addition to any other powers set forth in the Plan and subject to the provisions of the Plan, the Committee shall have the full and final power and authority, in its discretion:

(a) to determine the persons to whom, and the time or times at which, Awards shall be granted and the number of shares of Stock or units to be subject to each Award;

(b) to determine the type of Award granted and to designate Options as Incentive Stock Options, Nonstatutory Stock Options or Indexed Options;

(c) to determine the Fair Market Value of shares of Stock or other property;

(d) to determine the terms, conditions and restrictions applicable to each Award (which need not be identical) and any shares acquired pursuant thereto, including, without limitation, (i) the exercise or purchase price of shares purchased pursuant to any Award, (ii) the method of payment for shares purchased pursuant to any Award, (iii) the method for satisfaction of any tax withholding obligation arising in connection with Award, including by the withholding or delivery of shares of Stock, (iv) the timing, terms and conditions of the exercisability or vesting of any Award or any shares acquired pursuant thereto, (v) the Performance Award Formula and Performance Goals applicable to any Award and the extent to which such Performance Goals have been attained, (vi) the time of the expiration of any Award, (vii) the effect of the Participant s termination of Service on any of the foregoing, and (viii) all other terms, conditions and restrictions applicable to any Award or shares acquired pursuant thereto not inconsistent with the terms of the Plan;

(e) to determine whether an Award of Restricted Stock Units, SARs, Performance Shares or Performance Units will be settled in shares of Stock, cash, or in any combination thereof;

(f) to approve one or more forms of Award Agreement;

(g) to amend, modify, extend, cancel or renew any Award or to waive any restrictions or conditions applicable to any Award or any shares acquired pursuant thereto;

(h) to accelerate, continue, extend or defer the exercisability or vesting of any Award or any shares acquired pursuant thereto, including with respect to the period following a Participant s termination of Service;

(i) to prescribe, amend or rescind rules, guidelines and policies relating to the Plan, or to adopt sub-plans or supplements to, or alternative versions of, the Plan, including, without limitation, as the Committee deems necessary or desirable to comply with the laws or regulations of or to accommodate the tax policy, accounting principles or custom of, foreign jurisdictions whose citizens may be granted Awards;

(j) to authorize, in conjunction with any applicable Company deferred compensation plan, that the receipt of cash or Stock subject to any Award (other than an Option) under this Plan, may be deferred under the terms and conditions of such Company deferred compensation plan and in compliance with the applicable requirements of Section 409A of the Code; and

# Edgar Filing: INVITROGEN CORP - Form DEFR14A

B-5

(k) to correct any defect, supply any omission or reconcile any inconsistency in the Plan or any Award Agreement and to make all other determinations and take such other actions with respect to the Plan or any Award as the Committee may deem advisable to the extent not inconsistent with the provisions of the Plan or applicable law.

3.6 **No Repricing**. Without the affirmative vote of holders of a majority of the shares of Stock cast in person or by proxy at a meeting of the stockholders of the Company at which a quorum representing a majority of all outstanding shares of Stock is present or represented by proxy, the Board shall not approve a program providing for either (a) the cancellation of outstanding Options and/or SARs and the grant in substitution therefore of any new Awards, including specifically any new Options and/or SARs having a lower exercise price or (b) the amendment of outstanding Options and/or SARs to reduce the exercise price thereof. This paragraph shall not be construed to apply to issuing or assuming a stock option in a transaction to which section 424(a) applies, within the meaning of Section 424 of the Code.

3.7 **Indemnification**. In addition to such other rights of indemnification as they may have as members of the Board or the Committee or as officers or employees of the Participating Company Group, members of the Board or the Committee and any officers or employees of the Participating Company Group to whom authority to act for the Board, the Committee or the Company is delegated shall be indemnified by the Company against all reasonable expenses, including attorneys fees, actually and necessarily incurred in connection with the defense of any action, suit or proceeding, or in connection with any appeal therein, to which they or any of them may be a party by reason of any action taken or failure to act under or in connection with the Plan, or any right granted hereunder, and against all amounts paid by them in settlement thereof (provided such settlement is approved by independent legal counsel selected by the Company) or paid by them in satisfaction of a judgment in any such action, suit or proceeding, except in relation to matters as to which it shall be adjudged in such action, suit or proceeding that such person is liable for gross negligence, bad faith or intentional misconduct in duties; provided, however, that within sixty (60) days after the institution of such action, suit or proceeding, such person shall offer to the Company, in writing, the opportunity at its own expense to handle and defend the same.

### 4. <u>SHARES SUBJECT TO PLAN.</u> 4.1 Maximum Number of Shares Issuable.

(a) *Plan Share Reserve*. Subject to adjustment as provided in Section 4.2, the maximum aggregate number of shares of Stock that may be granted under the Plan shall be Fifteen Million and Sixteen Thousand Shares (15,016,000), reduced at any time by the number of shares subject to the Prior Plan Options (which as of the Effective Date equaled approximately Seven Million Eight Hundred and Sixteen Thousand Shares (7,816,000)). Such shares shall consist of authorized but unissued or reacquired shares of Stock or any combination thereof. If any outstanding Award, including any Prior Plan Options, for any reason expires or is terminated or canceled without having been exercised or settled in full, or if shares of Stock acquired pursuant to an Award subject to forfeiture or repurchase, including any Prior Plan Options, are forfeited or repurchased by the Company, the shares of Stock allocable to the terminated portion of such Award, including any Prior Plan Options, or such forfeited or repurchased shares of Stock shall again be available for grant under the Plan. Shares of Stock shall not be deemed to have been granted pursuant to the Plan with respect to any portion of an Award that is settled in cash. Notwithstanding the provisions of this Section 4.1(a), the following shares of Stock shall not be available for reissuance under the Plan: (i) shares of Stock that are withheld and/or surrendered in satisfaction of tax withholding obligations pursuant to Section 15.2; (ii) shares of Stock which are surrendered, through tender to the Company or attestation of ownership, in payment of any applicable Option exercise price; and (iii) shares of Stock subject to the grant of a SAR which are not required to be issued upon the exercise of such SAR.

(b) *Share Accounting*. Any shares of Stock subject to Options or SARs shall be counted against the Plan Share Reserve set forth in Section 4.1(a) as one share of Stock for every one share subject thereto. However, any shares of Stock subject to Awards other than Options or SARs shall be counted against the Plan Share Reserve set forth in Section 4.1(a) as 1.6 shares of Stock for every one share subject thereto.

4.2 Adjustments for Changes in Capital Structure. Subject to any required action by the stockholders of the Company, in the event of any change in the Stock effected without receipt of consideration by the Company, whether through merger, consolidation, reorganization, reincorporation, recapitalization, reclassification, stock dividend, stock split, reverse stock split, split-up, split-off, spin-off, combination of shares, exchange of shares, or similar change in the capital structure of the Company, or in the event of payment of a dividend or distribution to the stockholders of the Company in a form other than Stock (excepting normal cash dividends) that has a material effect on the Fair Market Value of shares of Stock, appropriate adjustments shall be made in the number and class of shares subject to the Plan and to any outstanding Awards, and in the exercise or purchase price per share under any outstanding Award in order to prevent dilution or enlargement of Participants rights under the Plan. For purposes of the foregoing, conversion of any convertible securities of the Company shall not be treated as effected without receipt of consideration by the Company. Any fractional share resulting from an

adjustment pursuant to this Section 4.2 shall be rounded down to the nearest whole number, and in no event may the exercise or purchase price under any Award be decreased to an amount less than the par value, if any, of the stock subject to such Award. The adjustments determined by the Committee pursuant to this Section 4.2 shall be final, binding and conclusive.

### 5. <u>Eligibility</u> and Award Limitations.

5.1 **Persons Eligible for Awards**. Awards may be granted only to Employees, Consultants and Directors. For purposes of the foregoing sentence, Employees, Consultants and Directors shall include prospective Employees, prospective Consultants and prospective Directors to whom Awards are granted in connection with written offers of an employment or other service relationship with the Participating Company Group; provided, however, that no Stock subject to any such Award shall vest, become exercisable or be issued prior to the date on which such person commences Service.

5.2 **Participation**. Awards are granted solely at the discretion of the Committee. Eligible persons may be granted more than one (1) Award. However, eligibility in accordance with this Section shall not entitle any person to be granted an Award, or, having been granted an Award, to be granted an additional Award.

## 5.3 Incentive Stock Option Limitations.

(a) *Persons Eligible*. An Incentive Stock Option may be granted only to a person who, on the effective date of grant, is an Employee of the Company, a Parent Corporation or a Subsidiary Corporation (each being an *ISO-Qualifying Corporation*). Any person who is not an Employee of an ISO-Qualifying Corporation on the effective date of the grant of an Option to such person may be granted only a Nonstatutory Stock Option. An Incentive Stock Option granted to a prospective Employee upon the condition that such person become an Employee of an ISO-Qualifying Corporation shall be deemed granted effective on the date such person commences Service with an ISO-Qualifying Corporation, with an exercise price determined as of such date in accordance with Section 6.1.

(b) *Fair Market Value Limitation*. To the extent that options designated as Incentive Stock Options (granted under all stock option plans of the Participating Company Group, including the Plan) become exercisable by a Participant for the first time during any calendar year for stock having a Fair Market Value greater than One Hundred Thousand dollars (\$100,000), the portion of such options which exceeds such amount shall be treated as Nonstatutory Stock Options. For purposes of this Section, options designated as Incentive Stock Options shall be taken into account in the order in which they were granted, and the Fair Market Value of stock shall be determined as of the time the option with respect to such stock is granted. If the Code is amended to provide for a different limitation from that set forth in this Section, such different limitation shall be deemed incorporated herein effective as of the date and with respect to such Options as required or permitted by such amendment to the Code. If an Option is treated as an Incentive Stock Option in part and as a Nonstatutory Stock Option in part by reason of the limitation set forth in this Section, the Participant may designate which portion of such Option the Participant is exercising. In the absence of such designation, the Participant shall be deemed to have exercised the Incentive Stock Option portion of the Option first. Upon exercise, shares issued pursuant to each such portion shall be separately identified.

5.4 Section 162(m) Award Limits. The following limits shall apply to the grant of any Award if, at the time of grant, the Company is a publicly held corporation within the meaning of Section 162(m).

(i) **Options and SARs**. Subject to adjustment as provided in Section 4.2, no Employee shall be granted within any fiscal year of the Company one or more Options or Freestanding SARs which in the aggregate are for more than One Million (1,000,000) shares of Stock, provided, however, that the Company may make an additional one-time grant to any newly-hired Employee of an Option and/or SAR for the purchase of up to an additional Five Hundred Thousand (500,000) shares of Stock. An Option which is canceled (or a Freestanding SAR as to which the exercise price is reduced to reflect a reduction in the Fair Market Value of the Stock) in the same fiscal year of the Company in which it was granted shall continue to be counted against such limit for such fiscal year.

(ii) **Restricted Stock Awards and Restricted Stock Units**. Subject to adjustment as provided in Section 4.2, no Employee shall be granted within any fiscal year of the Company one or more Restricted Stock Awards or Restricted Stock Units, subject to Vesting Conditions based on the attainment of Performance Goals, for more than One Hundred Thousand (100,000) shares of Stock, provided, however, that the Company may make an additional one-time grant to any newly-hired Employee of a Restricted Stock Award or Restricted Stock Units of up to an additional Fifty Thousand (50,000) shares of Stock.

(iii) **Performance Awards**. Subject to adjustment as provided in Section 4.2, no Employee shall be granted (A) Performance Shares which could result in such Employee receiving more than Six Hundred Thousand

(600,000) shares of Stock for each full fiscal year of the Company contained in the Performance Period for such Award, or (B) Performance Units which could result in such Employee receiving more than Four Million dollars (\$4,000,000) for each full fiscal year of the Company contained in the Performance Period for such Award. No Participant may be granted more than one Performance Award for the same Performance Period.

### 6. <u>TERMS AND CONDITIONS OF OPTIONS</u>.

Options shall be evidenced by Award Agreements specifying the number of shares of Stock covered thereby, in such form as the Committee shall from time to time establish. No Option or purported Option shall be a valid and binding obligation of the Company unless evidenced by a fully executed Award Agreement. Award Agreements evidencing Options may incorporate all or any of the terms of the Plan by reference and shall comply with and be subject to the following terms and conditions:

6.1 **Exercise Price**. The exercise price for each Option shall be established in the discretion of the Committee; provided, however, that (a) the exercise price per share shall be not less than the Fair Market Value of a share of Stock on the effective date of grant of the Option, (b) no Incentive Stock Option granted to a Ten Percent Owner shall have an exercise price per share less than one hundred ten percent (110%) of the Fair Market Value of a share of Stock on the effective date of grant of the Contrary in this Section 6.1, in the case of an Indexed Option, the Committee shall determine the exercise price of such Indexed Option and the terms and conditions that affect, if any, any adjustments to the exercise price of such Indexed Option. Notwithstanding the foregoing, an Option may be granted with an exercise price lower than the minimum exercise price set forth above if such Option is granted pursuant to an assumption or substitution for another option in a manner qualifying under the provisions of Section 424(a) of the Code.

6.2 Exercisability and Term of Options. Options shall be exercisable at such time or times, or upon such event or events, and subject to such terms, conditions, performance criteria and restrictions as shall be determined by the Committee and set forth in the Award Agreement evidencing such Option; provided, however, that (a) no Option shall be exercisable after the expiration of ten (10) years after the effective date of grant of such Option, (b) no Incentive Stock Option granted to a Ten Percent Owner shall be exercisable after the expiration of five (5) years after the effective date of grant of such Option, and (c) no Option granted to a prospective Employee, prospective Consultant or prospective Director may become exercisable prior to the date on which such person commences Service. Subject to the foregoing, unless otherwise specified by the Committee in the grant of an Option, any Option granted hereunder shall terminate ten (10) years after the effective date of grant of the Option, unless earlier terminated in accordance with its provisions.

### 6.3 Payment of Exercise Price.

(a) *Forms of Consideration Authorized*. Except as otherwise provided below, payment of the exercise price for the number of shares of Stock being purchased pursuant to any Option shall be made (i) in cash, by check or in cash equivalent, (ii) by tender to the Company, or attestation to the ownership, of shares of Stock owned by the Participant having a Fair Market Value not less than the exercise price, (iii) by delivery of a properly executed notice of exercise together with irrevocable instructions to a broker providing for the assignment to the Company of the proceeds of a sale or loan with respect to some or all of the shares being acquired upon the exercise of the Option (including, without limitation, through an exercise complying with the provisions of Regulation T as promulgated from time to time by the Board of Governors of the Federal Reserve System) (a *Cashless Exercise*), (iv) by such other consideration as may be approved by the Committee from time to time to the extent permitted by applicable law, or (v) by any combination thereof. The Committee may at any time or from time to time grant Options which do not permit all of the foregoing forms of consideration to be used in payment of the exercise price or which otherwise restrict one or more forms of consideration.

### (b) Limitations on Forms of Consideration.

(i) **Tender of Stock**. Notwithstanding the foregoing, an Option may not be exercised by tender to the Company, or attestation to the ownership, of shares of Stock to the extent such tender or attestation would constitute a violation of the provisions of any law, regulation or agreement restricting the redemption of the Company s stock. Unless otherwise provided by the Committee, an Option may not be exercised by tender to the Company, or attestation to the ownership, of shares of Stock unless such shares either have been owned by the Participant for more than six (6) months (and not used for another Option exercise by attestation during such period) or were not acquired, directly or indirectly, from the Company.

(ii) **Cashless Exercise**. The Company reserves, at any and all times, the right, in the Company s sole and absolute discretion, to establish, decline to approve or terminate any program or procedures for the exercise of

Options by means of a Cashless Exercise, including with respect to one or more Participants specified by the Company notwithstanding that such program or procedures may be available to other Participants.

6.4 **Effect of Termination of Service**. An Option shall be exercisable after a Participant s termination of Service to such extent and during such period as determined by the Committee, in its discretion, and set forth in the Award Agreement evidencing such Option.

6.5 **Transferability of Options**. During the lifetime of the Participant, an Option shall be exercisable only by the Participant or the Participant s guardian or legal representative. Prior to the issuance of shares of Stock upon the exercise of an Option, the Option shall not be subject in any manner to anticipation, alienation, sale, exchange, transfer, assignment, pledge, encumbrance, or garnishment by creditors of the Participant or the Participant s beneficiary, except transfer by will or by the laws of descent and distribution. Notwithstanding the foregoing, to the extent permitted by the Committee, in its discretion, and set forth in the Award Agreement evidencing such Option, a Nonstatutory Stock Option shall be assignable or transferable subject to the applicable limitations, if any, described in the General Instructions to Form S-8 Registration Statement under the Securities Act. Notwithstanding any of the foregoing, the Board may permit further transferability of any Option, on a general or specific basis, and may impose conditions and limitations on any permitted transferability.

### 7. <u>TERMS AND CONDITIONS OF STOCK APPRECIATION RIGHTS.</u>

Stock Appreciation Rights shall be evidenced by Award Agreements specifying the number of shares of Stock subject to the Award, in such form as the Committee shall from time to time establish. No SAR or purported SAR shall be a valid and binding obligation of the Company unless evidenced by a fully executed Award Agreement. Award Agreements evidencing SARs may incorporate all or any of the terms of the Plan by reference and shall comply with and be subject to the following terms and conditions:

7.1 **Types of SARs Authorized**. SARs may be granted in tandem with all or any portion of a related Option (a *Tandem SAR*) or may be granted independently of any Option (a *Freestanding SAR*). A Tandem SAR may be granted either concurrently with the grant of the related Option or at any time thereafter prior to the complete exercise, termination, expiration or cancellation of such related Option.

7.2 **Exercise Price**. The exercise price for each SAR shall be established in the discretion of the Committee; provided, however, that (a) the exercise price per share subject to a Tandem SAR shall be the exercise price per share under the related Option and (b) the exercise price per share subject to a Freestanding SAR shall be not less than the Fair Market Value of a share of Stock on the effective date of grant of the SAR.

### 7.3 Exercisability and Term of SARs.

(a) *Tandem SARs*. Tandem SARs shall be exercisable only at the time and to the extent, and only to the extent, that the related Option is exercisable, subject to such provisions as the Committee may specify where the Tandem SAR is granted with respect to less than the full number of shares of Stock subject to the related Option. The Committee may, in its discretion, provide in any Award Agreement evidencing a Tandem SAR that such SAR may not be exercised without the advance approval of the Company and, if such approval is not given, then the Option shall nevertheless remain exercisable in accordance with its terms. A Tandem SAR shall terminate and cease to be exercisable no later than the date on which the related Option expires or is terminated or canceled. Upon the exercise of a Tandem SAR with respect to some or all of the shares subject to such SAR, the related Option shall be canceled automatically as to the number of shares subject to such Option, the related Tandem SAR as to some or all of the shares subject to such Option, the related Tandem SAR shall be canceled automatically as to the number of shares subject to such Option, the related Tandem SAR shall be canceled automatically as to the related Option was exercised.

(b) *Freestanding SARs*. Freestanding SARs shall be exercisable at such time or times, or upon such event or events, and subject to such terms, conditions, performance criteria and restrictions as shall be determined by the Committee and set forth in the Award Agreement evidencing such SAR; provided, however, that no Freestanding SAR shall be exercisable after the expiration of ten (10) years after the effective date of grant of such SAR.

7.4 **Exercise of SARs**. Upon the exercise (or deemed exercise pursuant to Section 7.5) of a SAR, the Participant (or the Participant s legal representative or other person who acquired the right to exercise the SAR by reason of the Participant s death) shall be entitled to receive payment of an amount for each share with respect to which the SAR is exercised equal to the excess, if any, of the Fair Market Value of a share of Stock on the date of exercise of the SAR over the exercise price. Payment of such amount shall be made in cash, shares of Stock, or any combination thereof as determined by the Committee. Unless otherwise provided in the Award Agreement evidencing such SAR, payment shall be made in a lump sum as soon as practicable following the date of exercise of the SAR. The Award Agreement evidencing any SAR may provide for deferred

payment in a lump sum or in installments. When payment is to be made in shares of Stock, the number of shares to be issued shall be determined on the basis of the Fair Market Value of a share of Stock on the date of exercise of the SAR. For purposes of Section 7, a SAR shall be deemed exercised on the date on which the Company receives notice of exercise from the Participant.

7.5 **Deemed Exercise of SARs**. If, on the date on which a SAR would otherwise terminate or expire, the SAR by its terms remains exercisable immediately prior to such termination or expiration and, if so exercised, would result in a payment to the holder of such SAR, then any portion of such SAR which has not previously been exercised shall automatically be deemed to be exercised as of such date with respect to such portion.

7.6 Effect of Termination of Service. Subject to earlier termination of the SAR as otherwise provided herein a SAR shall be exercisable after a Participant s termination of Service to such extent and during such period as determined by the Committee, in its discretion, and set forth in the Award Agreement evidencing such SAR and thereafter shall terminate.

7.7 **Nontransferability of SARs**. During the lifetime of the Participant, a SAR shall be exercisable only by the Participant or the Participant s guardian or legal representative. Prior to the exercise of a SAR, the SAR shall not be subject in any manner to anticipation, alienation, sale, exchange, transfer, assignment, pledge, encumbrance, or garnishment by creditors of the Participant or the Participant s beneficiary, except transfer by will or by the laws of descent and distribution. Notwithstanding any of the foregoing, the Board may permit further transferability of any SAR, on a general or specific basis, and may impose conditions and limitations on any permitted transferability.

### 8. <u>TERMS AND CONDITIONS OF RESTRICTED STOCK AWARDS</u>.

Restricted Stock Awards shall be evidenced by Award Agreements specifying whether the Award is a Restricted Stock Bonus or a Restricted Stock Purchase Right and the number of shares of Stock subject to the Award, in such form as the Committee shall from time to time establish. No Restricted Stock Award or purported Restricted Stock Award shall be a valid and binding obligation of the Company unless evidenced by a fully executed Award Agreement. Award Agreements evidencing Restricted Stock Awards may incorporate all or any of the terms of the Plan by reference and shall comply with and be subject to the following terms and conditions:

8.1 **Types of Restricted Stock Awards Authorized**. Restricted Stock Awards may be in the form of either a Restricted Stock Bonus or a Restricted Stock Purchase Right. Restricted Stock Awards may be granted upon such conditions as the Committee shall determine, including, without limitation, upon the attainment of one or more Performance Goals described in Section 9.4. If either the grant of a Restricted Stock Award or the lapsing of the Restriction Period is to be contingent upon the attainment of one or more Performance Goals, the Committee shall follow procedures substantially equivalent to those set forth in Sections 9.3 through 9.5(a).

8.2 **Purchase Price**. The purchase price for shares of Stock issuable under each Restricted Stock Purchase Right shall be established by the Committee in its discretion. No monetary payment (other than applicable tax withholding) shall be required as a condition of receiving shares of Stock pursuant to a Restricted Stock Bonus, the consideration for which shall be services actually rendered to a Participating Company or for its benefit. Notwithstanding the foregoing, the Participant shall furnish consideration in the form of cash or past services rendered to a Participating Company or for its benefit having a value not less than the par value of the shares of Stock subject to such Restricted Stock Award.

8.3 **Purchase Period**. A Restricted Stock Purchase Right shall be exercisable within a period established by the Committee, which shall in no event exceed thirty (30) days from the effective date of the grant of the Restricted Stock Purchase Right; provided, however, that no Restricted Stock Purchase Right granted to a prospective Employee, prospective Consultant or prospective Director may become exercisable prior to the date on which such person commences Service.

8.4 **Payment of Purchase Price**. Except as otherwise provided below, payment of the purchase price for the number of shares of Stock being purchased pursuant to any Restricted Stock Purchase Right shall be made (a) in cash, by check, or in cash equivalent, (b) by such other consideration as may be approved by the Committee from time to time to the extent permitted by applicable law, or (iii) by any combination thereof. The Committee may at any time or from time to time grant Restricted Stock Purchase Rights which do not permit all of the foregoing forms of consideration to be used in payment of the purchase price or which otherwise restrict one or more forms of consideration. Restricted Stock Bonuses shall be issued in consideration for past services actually rendered to a Participating Company or for its benefit.

8.5 Vesting and Restrictions on Transfer. Shares issued pursuant to any Restricted Stock Award may or may not be made subject to Vesting Conditions based upon the satisfaction of such Service requirements, conditions, restrictions or performance criteria, including, without limitation, Performance Goals as described in Section 9.4, as shall be established by the Committee and set forth in the Award Agreement evidencing such Award. During any Restriction Period in which shares

acquired pursuant to a Restricted Stock Award remain subject to Vesting Conditions, such shares may not be sold, exchanged, transferred, pledged, assigned or otherwise disposed of other than pursuant to an Ownership Change Event, as defined in Section 13.1, or as provided in Section 8.8. Upon request by the Company, each Participant shall execute any agreement evidencing such transfer restrictions prior to the receipt of shares of Stock hereunder and shall promptly present to the Company any and all certificates representing shares of Stock acquired hereunder for the placement on such certificates of appropriate legends evidencing any such transfer restrictions. All Awards shall be subject to a minimum vesting period of one year as a Vesting Condition.

8.6 Voting Rights; Dividends and Distributions. Except as provided in this Section, Section 8.5 and any Award Agreement, during the Restriction Period applicable to shares subject to a Restricted Stock Award, the Participant shall have all of the rights of a stockholder of the Company holding shares of Stock, including the right to vote such shares and to receive all dividends and other distributions paid with respect to such shares. However, in the event of a dividend or distribution paid in shares of Stock or any other adjustment made upon a change in the capital structure of the Company as described in Section 4.2, then any and all new, substituted or additional securities or other property (other than normal cash dividends) to which the Participant is entitled by reason of the Participant s Restricted Stock Award shall be immediately subject to the same Vesting Conditions as the shares subject to the Restricted Stock Award with respect to which such dividends or distributions were paid or adjustments were made.

8.7 Effect of Termination of Service. Unless otherwise provided by the Committee in the grant of a Restricted Stock Award and set forth in the Award Agreement, if a Participant s Service terminates for any reason, whether voluntary or involuntary (including the Participant s death or Disability), then (a) the Company shall have the option to repurchase for the purchase price paid by the Participant any shares acquired by the Participant to a Restricted Stock Purchase Right which remain subject to Vesting Conditions as of the date of the Participant s termination of Service and (b) the Participant shall forfeit to the Company any shares acquired by the Participant pursuant to a Restricted Stock Bonus which remain subject to Vesting Conditions as of the date of the Participant s termination of Service. The Company shall have the right to assign at any time any repurchase right it may have, whether or not such right is then exercisable, to one or more persons as may be selected by the Company.

8.8 **Nontransferability of Restricted Stock Award Rights**. Prior to the issuance of shares of Stock pursuant to a Restricted Stock Award, rights to acquire such shares shall not be subject in any manner to anticipation, alienation, sale, exchange, transfer, assignment, pledge, encumbrance or garnishment by creditors of the Participant or the Participant s beneficiary, except transfer by will or the laws of descent and distribution. All rights with respect to a Restricted Stock Award granted to a Participant hereunder shall be exercisable during his or her lifetime only by such Participant or the Participant s guardian or legal representative.

8.9 Minimum Vesting Period. Notwithstanding anything to the contrary in this Plan, Shares issued pursuant to any Award shall be subject to a minimum vesting period of one year as a Vesting Condition.

## 9. <u>TERMS AND CONDITIONS OF PERFORMANCE AWARDS</u>.

Performance Awards shall be evidenced by Award Agreements in such form as the Committee shall from time to time establish. No Performance Award or purported Performance Award shall be a valid and binding obligation of the Company unless evidenced by a fully executed Award Agreement. Award Agreements evidencing Performance Awards may incorporate all or any of the terms of the Plan by reference and shall comply with and be subject to the following terms and conditions:

9.1 **Types of Performance Awards Authorized**. Performance Awards may be in the form of either Performance Shares or Performance Units. Each Award Agreement evidencing a Performance Award shall specify the number of Performance Shares or Performance Units subject thereto, the Performance Award Formula, the Performance Goal(s) and Performance Period applicable to the Award, and the other terms, conditions and restrictions of the Award.

9.2 **Initial Value of Performance Shares and Performance Units**. Unless otherwise provided by the Committee in granting a Performance Award, each Performance Share shall have an initial value equal to the Fair Market Value of one (1) share of Stock, subject to adjustment as provided in Section 4.2, on the effective date of grant of the Performance Share, and each Performance Unit shall have an initial value of one hundred dollars (\$100). The final value payable to the Participant in settlement of a Performance Award determined on the basis of the applicable Performance Award Formula will depend on the extent to which Performance Goals established by the Committee are attained within the applicable Performance Period established by the Committee.

9.3 Establishment of Performance Period, Performance Goals and Performance Award Formula. In granting each Performance Award, the Committee shall establish in writing the applicable Performance Period, Performance Award

Formula and one or more Performance Goals which, when measured at the end of the Performance Period, shall determine on the basis of the Performance Award Formula the final value of the Performance Award to be paid to the Participant. Unless otherwise permitted in compliance with the requirements under Section 162(m) with respect to performance-based compensation, the Committee shall establish the Performance Goal(s) and Performance Award Formula applicable to each Performance Award no later than the earlier of (a) the date ninety (90) days after the commencement of the applicable Performance Period or (b) the date on which 25% of the Performance Period has elapsed, and, in any event, at a time when the outcome of the Performance Goals remains substantially uncertain. Once established, the Performance Goals and Performance Award Formula shall not be changed during the Performance Period. The Company shall notify each Participant granted a Performance Award of the terms of such Award, including the Performance Period, Performance Goal(s) and Performance Award Formula.

9.4 **Measurement of Performance Goals**. Performance Goals shall be established by the Committee on the basis of targets to be attained (*Performance Targets*) with respect to one or more measures of business or financial performance (each, a *Performance Measure*), subject to the following:

(a) *Performance Measures*. Performance Measures shall have the same meanings as used in the Company's financial statements, or, if such terms are not used in the Company's financial statements, they shall have the meaning applied pursuant to generally accepted accounting principles, or as used generally in the Company's industry. Performance Measures shall be calculated with respect to the Company and each Subsidiary Corporation consolidated therewith for financial reporting purposes or such division or other business unit as may be selected by the Committee. For purposes of the Plan, the Performance Measures applicable to a Performance Award shall be calculated in accordance with generally accepted accounting principles, but prior to the accrual or payment of any Performance Award for the same Performance Period and excluding the effect (whether positive or negative) of any change in accounting standards or any extraordinary, unusual or nonrecurring item, as determined by the Committee, occurring after the establishment of the Performance Goals applicable to the Performance Award. Performance Measures may be one or more of the following, or a combination of the any of the following, as determined by the Committee:

(i) revenue;

- (ii) gross margin;
- (iii) operating margin;
- (iv) operating income;

(v) pre-tax profit;

(vi) earnings before interest, taxes and depreciation;

(vii) net income;

(viii) cash flow;

(ix) expenses;

- (x) the market price of the Stock;
- (xi) earnings per share;
- (xii) return on stockholder equity;
- (xiii) return on capital;
- (xiv) return on net assets;
- (xv) economic value added;
- (xvi) number of customers;
- (xvii) market share;

(xviii) return on investment

(xix) profit after tax; and

(xx) customer satisfaction.

(b) *Performance Targets*. Performance Targets may include a minimum, maximum, target level and intermediate levels of performance, with the final value of a Performance Award determined under the applicable Performance Award Formula by the level attained during the applicable Performance Period. A Performance Target may be stated as an absolute value or as a value determined relative to a standard selected by the Committee.

### 9.5 Settlement of Performance Awards.

(a) *Determination of Final Value*. As soon as practicable following the completion of the Performance Period applicable to a Performance Award, the Committee shall certify in writing the extent to which the applicable Performance Goals have been attained and the resulting final value of the Award earned by the Participant and to be paid upon its settlement in accordance with the applicable Performance Award Formula.

(b) *Discretionary Adjustment of Award Formula*. In its discretion, the Committee may, either at the time it grants a Performance Award or at any time thereafter, provide for the positive or negative adjustment of the Performance Award Formula applicable to a Performance Award granted to any Participant who is not a covered employee within the meaning of Section 162(m) (a *Covered Employee*) to reflect such Participant s individual performance in his or her position with the Company or such other factors as the Committee may determine. If permitted under a Covered Employee s Award Agreement, the Committee shall have the discretion, on the basis of such criteria as may be established by the Committee, to reduce some or all of the value of the Performance Award that would otherwise be paid to the Covered Employee upon its settlement notwithstanding the attainment of any Performance Goal and the resulting value of the Performance Award determined in accordance with the Performance Award Formula. No such reduction may result in an increase in the amount payable upon settlement of another Participant s Performance Award.

(c) *Effect of Leaves of Absence*. Unless otherwise required by law, payment of the final value, if any, of a Performance Award held by a Participant who has taken in excess of thirty (30) days in leaves of absence during a Performance Period shall be prorated on the basis of the number of days of the Participant s Service during the Performance Period during which the Participant was not on a leave of absence.

(d) *Notice to Participants*. As soon as practicable following the Committee s determination and certification in accordance with Sections 9.5(a) and (b), the Company shall notify each Participant of the determination of the Committee.

(e) *Payment in Settlement of Performance Awards*. As soon as practicable following the Committee's determination and certification in accordance with Sections 9.5(a) and (b), payment shall be made to each eligible Participant (or such Participant's legal representative or other person who acquired the right to receive such payment by reason of the Participant's death) of the final value of the Participant's Performance Award. Payment of such amount shall be made in cash, shares of Stock, or a combination thereof as determined by the Committee. Unless otherwise provided in the Award Agreement evidencing a Performance Award, payment shall be made in a lump sum. An Award Agreement may provide for deferred payment in a lump sum or in installments. If any payment is to be made on a deferred basis, the Committee may, but shall not be obligated to, provide for the payment during the deferral period of Dividend Equivalents or interest.

(f) *Provisions Applicable to Payment in Shares*. If payment is to be made in shares of Stock, the number of such shares shall be determined by dividing the final value of the Performance Award by the value of a share of Stock determined by the method specified in the Award Agreement. Such methods may include, without limitation, the closing market price on a specified date (such as the settlement date) or an average of market prices over a series of trading days. Shares of Stock issued in payment of any Performance Award may be fully vested and freely transferable shares or may be shares of Stock subject to Vesting Conditions as provided in Section 8.5. Any shares subject to Vesting Conditions shall be evidenced by an appropriate Award Agreement and shall be subject to the provisions of Sections 8.5 through 8.8 above.

9.6 Voting Rights; Dividend Equivalent Rights and Distributions. Participants shall have no voting rights with respect to shares of Stock represented by Performance Share Awards until the date of the issuance of such shares, if any (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company). However, the Committee, in its discretion, may provide in the Award Agreement evidencing any Performance Share Award that the Participant shall be entitled to receive Dividend Equivalents with respect to the payment of cash dividends on Stock having a record date prior to the date on which the Performance Shares are settled or forfeited. Such Dividend Equivalents, if

any, shall be credited to the Participant in the form of additional whole Performance Shares as of the date of payment of such cash dividends on Stock. The number of additional Performance Shares (rounded to the nearest whole number) to be so credited shall be determined by dividing (a) the amount of cash dividends paid on such date with respect to the number of shares of Stock represented by the Performance Shares previously credited to the Participant by (b) the Fair Market Value per share of Stock on such date. Dividend Equivalents may be paid currently or may be accumulated and paid to the extent that Performance Shares become nonforfeitable, as determined by the Committee. Settlement of Dividend Equivalents may be made in cash, shares of Stock, or a combination thereof as determined by the Committee, and may be paid on the same basis as settlement of the related Performance Share as provided in Section 9.5. Dividend Equivalents shall not be paid with respect to Performance Units. In the event of a dividend or distribution paid in shares of Stock or any other adjustment made upon a change in the capital structure of the Company as described in Section 4.2, appropriate adjustments shall be made in the Participant s Performance Share Award so that it represents the right to receive upon settlement any and all new, substituted or additional securities or other property (other than normal cash dividends) to which the Participant would entitled by reason of the shares of Stock issuable upon settlement of the Performance Share Award, and all such new, substituted or additional securities or other property shall be immediately subject to the same Performance Goals as are applicable to the Award.

9.7 Effect of Termination of Service. The effect of a Participant s termination of Service on the Performance Award shall be determined by the Committee, in its discretion, and set forth in the Award Agreement evidencing such Performance Award.

9.8 **Nontransferability of Performance Awards**. Prior to settlement in accordance with the provisions of the Plan, no Performance Award shall be subject in any manner to anticipation, alienation, sale, exchange, transfer, assignment, pledge, encumbrance, or garnishment by creditors of the Participant or the Participant s beneficiary, except transfer by will or by the laws of descent and distribution. All rights with respect to a Performance Award granted to a Participant hereunder shall be exercisable during his or her lifetime only by such Participant or the Participant s guardian or legal representative.

### 10. <u>Terms and Conditions of Restricted Stock Unit Awards</u>.

Restricted Stock Unit Awards shall be evidenced by Award Agreements specifying the number of Restricted Stock Units subject to the Award, in such form as the Committee shall from time to time establish. No Restricted Stock Unit Award or purported Restricted Stock Unit Award shall be a valid and binding obligation of the Company unless evidenced by a fully executed Award Agreement. Award Agreements evidencing Restricted Stock Units may incorporate all or any of the terms of the Plan by reference and shall comply with and be subject to the following terms and conditions:

10.1 **Grant of Restricted Stock Unit Awards**. Restricted Stock Unit Awards may be granted upon such conditions as the Committee shall determine, including, without limitation, upon the attainment of one or more Performance Goals described in Section 9.4. If either the grant of a Restricted Stock Unit Award or the Vesting Conditions with respect to such Award is to be contingent upon the attainment of one or more Performance Goals, the Committee shall follow procedures substantially equivalent to those set forth in Sections 9.3 through 9.5(a).

10.2 **Purchase Price**. No monetary payment (other than applicable tax withholding, if any) shall be required as a condition of receiving a Restricted Stock Unit Award, the consideration for which shall be services actually rendered to a Participating Company or for its benefit.

10.3 **Vesting**. Restricted Stock Units may or may not be made subject to Vesting Conditions based upon the satisfaction of such Service requirements, conditions, restrictions or performance criteria, including, without limitation, Performance Goals as described in Section 9.4, as shall be established by the Committee and set forth in the Award Agreement evidencing such Award.

10.4 **Voting Rights, Dividend Equivalent Rights and Distributions**. Participants shall have no voting rights with respect to shares of Stock represented by Restricted Stock Units until the date of the issuance of such shares (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company). However, the Committee, in its discretion, may provide in the Award Agreement evidencing any Restricted Stock Unit Award that the Participant shall be entitled to receive Dividend Equivalents with respect to the payment of cash dividends on Stock having a record date prior to date on which Restricted Stock Units held by such Participant are settled. Such Dividend Equivalents, if any, shall be paid by crediting the Participant with additional whole Restricted Stock Units as of the date of payment of such cash dividends on Stock. The number of additional Restricted Stock Units (rounded to the nearest whole number) to be so credited shall be determined by dividing (a) the amount of cash dividends paid on such date with respect to the number of shares of Stock represented by the Restricted Stock Units previously credited to the Participant by (b) the Fair Market Value per share of Stock on such date. Such additional Restricted Stock Units shall be subject to the same terms and conditions and shall be settled in the same manner and at the same time (or as soon thereafter as practicable) as the Restricted Stock Units

originally subject to the Restricted Stock Unit Award. In the event of a dividend or distribution paid in shares of Stock or any other adjustment made upon a change in the capital structure of the Company as described in Section 4.2, appropriate adjustments shall be made in the Participant s Restricted Stock Unit Award so that it represents the right to receive upon settlement any and all new, substituted or additional securities or other property (other than normal cash dividends) to which the Participant would entitled by reason of the shares of Stock issuable upon settlement of the Award, and all such new, substituted or additional securities or other property shall be immediately subject to the same Vesting Conditions as are applicable to the Award.

10.5 **Effect of Termination of Service**. Unless otherwise provided by the Committee in the grant of a Restricted Stock Unit Award and set forth in the Award Agreement, if a Participant s Service terminates for any reason, whether voluntary or involuntary (including the Participant s death or Disability), then the Participant shall forfeit to the Company any Restricted Stock Units pursuant to the Award which remain subject to Vesting Conditions as of the date of the Participant s termination of Service.

10.6 **Settlement of Restricted Stock Unit Awards**. The Company shall issue to a Participant on the date on which Restricted Stock Units subject to the Participant s Restricted Stock Unit Award vest or on such other date determined by the Committee, in its discretion, and set forth in the Award Agreement one (1) share of Stock (and/or any other new, substituted or additional securities or other property pursuant to an adjustment described in Section 10.4) for each Restricted Stock Unit then becoming vested or otherwise to be settled on such date, subject to the withholding of applicable taxes. Notwithstanding the foregoing, if permitted by the Committee and set forth in the Award Agreement, the Participant may elect in accordance with terms specified in the Award Agreement to defer receipt of all or any portion of the shares of Stock or other property otherwise issuable to the Participant pursuant to this Section.

10.7 **Nontransferability of Restricted Stock Unit Awards**. Prior to the issuance of shares of Stock in settlement of a Restricted Stock Unit Award, the Award shall not be subject in any manner to anticipation, alienation, sale, exchange, transfer, assignment, pledge, encumbrance, or garnishment by creditors of the Participant or the Participant s beneficiary, except transfer by will or by the laws of descent and distribution. All rights with respect to a Restricted Stock Unit Award granted to a Participant hereunder shall be exercisable during his or her lifetime only by such Participant or the Participant s guardian or legal representative.

## 11. DEFERRED STOCK UNITS.

11.1 **Establishment of Deferred Stock Unit Program**. The Committee, in its discretion and upon such terms and conditions as it may determine, may establish one or more programs pursuant to the Plan under which Participants designated by the Committee who are Insiders or otherwise among a select group of highly compensated Employees may irrevocably elect, prior to a date specified by the Committee, to reduce such Participant s compensation otherwise payable in cash (subject to any minimum or maximum reductions imposed by the Committee) and to be granted automatically at such time or times as specified by the Committee one or more Awards of Deferred Stock Units with respect to such numbers of shares of Stock as determined in accordance with the rules of the program established by the Committee and having such other terms and conditions as established by the Committee.

11.2 **Terms and Conditions of Deferred Stock Units**. Deferred Stock Units granted pursuant to this Section 11 shall be evidenced by Award Agreements in such form as the Committee shall from time to time establish. No such Deferred Stock Unit or purported Deferred Stock Unit shall be a valid and binding obligation of the Company unless evidenced by a fully executed Award Agreement. Award Agreements evidencing Deferred Stock Units may incorporate all or any of the terms of the Plan by reference and shall comply with and be subject to the following terms and conditions:

(a) Vesting Conditions. Deferred Stock Units shall not be subject to any vesting conditions.

## (b) Terms and Conditions of Deferred Stock Units.

(i) **Voting Rights; Dividend Equivalent Rights and Distributions**. Participants shall have no voting rights with respect to shares of Stock represented by Deferred Stock Units until the date of the issuance of such shares (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company). However, a Participant shall be entitled to receive Dividend Equivalents with respect to the payment of cash dividends on Stock having a record date prior to date on which Deferred Stock Units held by such Participant are settled. Such Dividend Equivalents shall be paid by crediting the Participant with additional whole and/or fractional Deferred Stock Units as of the date of payment of such cash dividends on Stock. The method of determining the number of additional Deferred Stock Units to be so credited shall be specified by the Committee and set forth in the Award Agreement. Such additional Deferred Stock Units shall be subject to the same terms and conditions and shall be settled in the same manner and at the same time (or as soon thereafter as

practicable) as the Deferred Stock Units originally subject to the Deferred Stock Unit Award. In the event of a dividend or distribution paid in shares of Stock or any other adjustment made upon a change in the capital structure of the Company as described in Section 4.2, appropriate adjustments shall be made in the Participant s Deferred Stock Unit Award so that it represent the right to receive upon settlement any and all new, substituted or additional securities or other property (other than normal cash dividends) to which the Participant would entitled by reason of the shares of Stock issuable upon settlement of the Award.

(ii) **Settlement of Deferred Stock Unit Awards**. A Participant electing to receive an Award of Deferred Stock Units pursuant to this Section 11, shall specify at the time of such election a settlement date with respect to such Award. The Company shall issue to the Participant as soon as practicable following the earlier of the settlement date elected by the Participant or the date of termination of the Participant s Service, a number of whole shares of Stock equal to the number of whole Deferred Stock Units subject to the Deferred Stock Unit Award. Such shares of Stock shall be fully vested, and the Participant shall not be required to pay any additional consideration (other than applicable tax withholding) to acquire such shares. Any fractional Deferred Stock Unit subject to the Deferred Stock Unit Award shall be settled by the Company by payment in cash of an amount equal to the Fair Market Value as of the payment date of such fractional share.

(iii) **Nontransferability of Deferred Stock Unit Awards**. Prior to their settlement in accordance with the provision of the Plan, no Deferred Stock Unit Award shall be subject in any manner to anticipation, alienation, sale, exchange, transfer, assignment, pledge, encumbrance, or garnishment by creditors of the Participant or the Participant s beneficiary, except transfer by will or by the laws of descent and distribution. All rights with respect to a Deferred Stock Unit Award granted to a Participant hereunder shall be exercisable during his or her lifetime only by such Participant or the Participant s guardian or legal representative.

### 12. <u>Standard Forms of Award Agreement</u>.

12.1 Award Agreements. Each Award shall comply with and be subject to the terms and conditions set forth in the appropriate form of Award Agreement approved by the Committee and as amended from time to time. Any Award Agreement may consist of an appropriate form of Notice of Grant and a form of Agreement incorporated therein by reference, or such other form or forms as the Committee may approve from time to time.

12.2 **Authority to Vary Terms**. The Committee shall have the authority from time to time to vary the terms of any standard form of Award Agreement either in connection with the grant or amendment of an individual Award or in connection with the authorization of a new standard form or forms; provided, however, that the terms and conditions of any such new, revised or amended standard form or forms of Award Agreement are not inconsistent with the terms of the Plan.

#### 13. <u>CHANGE IN CONTROL</u>. 13.1 **Definitions**.

(a) An *Ownership Change Event* shall be deemed to have occurred if any of the following occurs with respect to the Company: (i) the direct or indirect sale or exchange in a single or series of related transactions by the stockholders of the Company of more than fifty percent (50%) of the voting stock of the Company; (ii) a merger or consolidation in which the Company is a party; (iii) the sale, exchange, or transfer of all or substantially all of the assets of the Company (other than a sale, exchange or transfer to one or more subsidiaries of the Company); or (iv) a liquidation or dissolution of the Company. Notwithstanding the foregoing, for any Awards granted after , 2007, an Ownership Change Event shall not include any event described in Section 13.1(a)(iv).

(b) A *Change in Control* shall mean an Ownership Change Event or series of related Ownership Change Events (collectively, a *Transaction*) in which the stockholders of the Company immediately before the Transaction do not retain immediately after the Transaction, in substantially the same proportions as their ownership of shares of the Company s voting stock immediately before the Transaction, direct or indirect beneficial ownership of more than fifty percent (50%) of the total combined voting power of the outstanding voting securities of the Company or, in the case of an Ownership Change Event described in Section 13.1(a)(iii), the entity to which the assets of the Company were transferred (the

*Transferee*), as the case may be. For purposes of the preceding sentence, indirect beneficial ownership shall include, without limitation, an interest resulting from ownership of the voting securities of one or more corporations or other business entities which own the Company or the Transferee, as the case may be, either directly or through one or more subsidiary corporations or other business entities. The Committee shall have the right to determine whether multiple sales or exchanges of the voting securities of the Company or multiple Ownership Change Events are related, and its determination shall be final, binding and conclusive.

### 13.2 Effect of Change in Control on Options and SARs.

(a) *Accelerated Vesting*. Notwithstanding any other provision of the Plan to the contrary, the Committee, in its sole discretion, may provide in any Award Agreement or, in the event of a Change in Control, may take such actions as it deems appropriate to provide for the acceleration of the exercisability and vesting in connection with such Change in Control of any or all outstanding Options and SARs and shares acquired upon the exercise of such Options and SARs upon such conditions and to such extent as the Committee shall determine.

(b) Assumption or Substitution. In the event of a Change in Control, the surviving, continuing, successor, or purchasing corporation or other business entity or parent corporation thereof, as the case may be (the Acquiring Corporation ), may, without the consent of the Participant, either assume the Company s rights and obligations under outstanding Options and SARs or substitute for outstanding Options and SARs substantially equivalent options and stock appreciation rights for the Acquiring Corporation s stock. In the event that the Acquiring Corporation elects not to assume or substitute for outstanding Options and SARs in connection with a Change in Control, or if the Acquiring Corporation is not a publicly held corporation within the meaning of Section 162(m), the exercisability and vesting of each such outstanding Option, SAR and any shares acquired upon the exercise thereof held by a Participant whose Service has not terminated prior to such date shall be accelerated, effective as of the date ten (10) days prior to the date of the Change in Control. The exercise or vesting of any Option, SAR and any shares acquired upon the exercise thereof that was permissible solely by reason of this Section 13.2 and the provisions of such applicable Award Agreement shall be conditioned upon the consummation of the Change in Control. Any Options and SARs which are neither assumed or substituted for by the Acquiring Corporation in connection with the Change in Control nor exercised as of the date of the Change in Control shall terminate and cease to be outstanding effective as of the date of the Change in Control. Notwithstanding the foregoing, shares acquired upon exercise of an Option prior to the Change in Control and any consideration received pursuant to the Change in Control with respect to such shares shall continue to be subject to all applicable provisions of the applicable Award Agreement evidencing such Option or SAR except as otherwise provided in such applicable Award Agreement. Furthermore, notwithstanding the foregoing, if the corporation the stock of which is subject to the outstanding Options and SARs immediately prior to an Ownership Change Event described in Section 13.1(a)(i) constituting a Change in Control is the surviving or continuing corporation and immediately after such Ownership Change Event less than fifty percent (50%) of the total combined voting power of its voting stock is held by another corporation or by other corporations that are members of an affiliated group within the meaning of Section 1504(a) of the Code without regard to the provisions of Section 1504(b) of the Code, the outstanding Options and SARs shall not terminate unless the Committee otherwise provides in its discretion.

(c) *Cash-Out*. The Committee may, in its sole discretion and without the consent of any Participant, determine that, upon the occurrence of a Change in Control, each or any Option or SAR outstanding immediately prior to the Change in Control shall be canceled in exchange for a payment with respect to each vested share of Stock subject to such canceled Option or SAR in (i) cash, (ii) stock of the Company or of a corporation or other business entity a party to the Change in Control, or (iii) other property which, in any such case, shall be in an amount having a Fair Market Value equal to the excess of the Fair Market Value of the consideration to be paid per share of Stock in the Change in Control over the exercise price per share under such Option or SAR (the *Spread*). In the event such determination is made by the Committee, the Spread (reduced by applicable withholding taxes, if any) shall be paid to Participants in respect of their canceled Options and SARs as soon as practicable following the date of the Change in Control.

13.3 **Effect of Change in Control on Restricted Stock Awards**. The Committee may, in its discretion, provide in any Award Agreement evidencing a Restricted Stock Award that, in the event of a Change in Control, the lapsing of the Restriction Period applicable to the shares subject to the Restricted Stock Award held by a Participant whose Service has not terminated prior to the Change in Control shall be accelerated effective immediately prior to the consummation of the Change in Control to such extent as specified in such Award Agreement. Any acceleration of the lapsing of the Restriction Period that was permissible solely by reason of this Section 13.3 and the provisions of such Award Agreement shall be conditioned upon the consummation of the Change in Control.

13.4 Effect of Change in Control on Performance Awards. The Committee may, in its discretion, provide in any Award Agreement evidencing a Performance Award that, in the event of a Change in Control, the Performance Award held by a Participant whose Service has not terminated prior to the Change in Control shall become payable effective as of the date of the Change in Control to such extent as specified in such Award Agreement.

13.5 Effect of Change in Control on Restricted Stock Unit Awards. The Committee may, in its discretion, provide in any Award Agreement evidencing a Restricted Stock Unit Award that, in the event of a Change in Control, the

Restricted Stock Unit Award held by a Participant whose Service has not terminated prior to such date shall be settled effective as of the date of the Change in Control to such extent as specified in such Award Agreement.

13.6 **Effect of Change in Control on Deferred Stock Units**. The Committee may, in its discretion, provide in any Award Agreement evidencing a Deferred Stock Unit Award that, in the event of a Change in Control, the Deferred Stock Units pursuant to such Award shall be settled effective as of the date of the Change in Control to such extent as specified in such Award Agreement.

#### 14. <u>COMPLIANCE WITH SECURITIES LAW.</u>

The grant of Awards and the issuance of shares of Stock pursuant to any Award shall be subject to compliance with all applicable requirements of federal, state and foreign law with respect to such securities and the requirements of any stock exchange or market system upon which the Stock may then be listed. In addition, no Award may be exercised or shares issued pursuant to an Award unless (a) a registration statement under the Securities Act shall at the time of such exercise or issuance be in effect with respect to the shares issuable pursuant to the Award or (b) in the opinion of legal counsel to the Company, the shares issuable pursuant to the Award may be issued in accordance with the terms of an applicable exemption from the registration requirements of the Securities Act. The inability of the Company to obtain from any regulatory body having jurisdiction the authority, if any, deemed by the Company s legal counsel to be necessary to the lawful issuance and sale of any shares hereunder shall relieve the Company of any liability in respect of the failure to issue or sell such shares as to which such requisite authority shall not have been obtained. As a condition to issuance of any Stock, the Company may require the Participant to satisfy any qualifications that may be necessary or appropriate, to evidence compliance with any applicable law or regulation and to make any representation or warranty with respect thereto as may be requested by the Company.

### 15. <u>Tax Withholding</u>.

15.1 **Tax Withholding in General**. The Company shall have the right to deduct from any and all payments made under the Plan, or to require the Participant, through payroll withholding, cash payment or otherwise, including by means of a Cashless Exercise of an Option, to make adequate provision for, the federal, state, local and foreign taxes, if any, required by law to be withheld by the Participating Company Group with respect to an Award or the shares acquired pursuant thereto. The Company shall have no obligation to deliver shares of Stock, to release shares of Stock from an escrow established pursuant to an Award Agreement, or to make any payment in cash under the Plan until the Participating Company Group s tax withholding obligations have been satisfied by the Participant.

15.2 **Withholding in Shares**. The Company shall have the right, but not the obligation, to deduct from the shares of Stock issuable to a Participant upon the exercise or settlement of an Award, or to accept from the Participant the tender of, a number of whole shares of Stock having a Fair Market Value, as determined by the Company, equal to all or any part of the tax withholding obligations of the Participating Company Group. The Fair Market Value of any shares of Stock withheld or tendered to satisfy any such tax withholding obligations shall not exceed the amount determined by the applicable minimum statutory withholding rates.

#### 16. <u>Amendment or Termination of Plan</u>.

The Committee may amend, suspend or terminate the Plan at any time. However, without the approval of the Company s stockholders, there shall be (a) no increase in the maximum aggregate number of shares of Stock that may be issued under the Plan (except by operation of the provisions of Section 4.2), (b) no change in the class of persons eligible to receive Incentive Stock Options, (c) no Option and/or SAR repricing as described in Section 3.6, (d) no amendment to permit the granting of Options (other than Indexed Options) with exercise prices less than Fair Market Value on the date of grant, and (e) no other amendment of the Plan that would require approval of the Company s stockholders under any applicable law, regulation or rule. No amendment, suspension or termination of the Plan shall affect any then outstanding Award unless expressly provided by the Committee. In any event, no amendment, suspension or termination of the Plan may adversely affect any then outstanding Award without the consent of the Participant unless necessary to comply with any applicable law, regulation or rule.

### 17. <u>Miscellaneous Provisions</u>.

17.1 **Repurchase Rights**. Shares issued under the Plan may be subject to one or more repurchase options, or other conditions and restrictions as determined by the Committee in its discretion at the time the Award is granted. The Company shall have the right to assign at any time any repurchase right it may have, whether or not such right is then exercisable, to one or more persons as may be selected by the Company. Upon request by the Company, each Participant shall execute any agreement evidencing such transfer restrictions prior to the receipt of shares of Stock hereunder and shall promptly present to

the Company any and all certificates representing shares of Stock acquired hereunder for the placement on such certificates of appropriate legends evidencing any such transfer restrictions.

17.2 **Provision of Information**. Each Participant shall be given access to information concerning the Company equivalent to that information generally made available to the Company s common stockholders.

17.3 **Rights as Employee, Consultant or Director**. No person, even though eligible pursuant to Section 5, shall have a right to be selected as a Participant, or, having been so selected, to be selected again as a Participant. Nothing in the Plan or any Award granted under the Plan shall confer on any Participant a right to remain an Employee, Consultant or Director or interfere with or limit in any way any right of a Participating Company to terminate the Participant s Service at any time. To the extent that an Employee of a Participating Company other than the Company receives an Award under the Plan, that Award shall in no event be understood or interpreted to mean that the Company is the Employee s employer or that the Employee has an employment relationship with the Company.

17.4 **Rights as a Stockholder**. A Participant shall have no rights as a stockholder with respect to any shares covered by an Award until the date of the issuance of such shares (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company). No adjustment shall be made for dividends, distributions or other rights for which the record date is prior to the date such shares are issued, except as provided in Section 4.2 or another provision of the Plan.

17.5 Fractional Shares. The Company shall not be required to issue fractional shares upon the exercise or settlement of any Award.

17.6 **Severability**. If any one or more of the provisions (or any part thereof) of this Plan or of any Award Agreement issued hereunder, shall be held to be invalid, illegal or unenforceable in any respect, such provision shall be modified so as to make it valid, legal and enforceable, and the validity, legality and enforceability of the remaining provisions (or any part thereof) of the Plan or of any Award Agreement shall not in any way be affected or impaired thereby. The Company may, without the consent of any Participant, and in a manner determined necessary solely in the discretion of the Company, amend the Plan and any outstanding Award Agreement as the Company deems necessary to ensure the Plan and all Awards remain valid, legal or enforceable in all respects.

17.7 **Beneficiary Designation**. Subject to local laws and procedures, each Participant may file with the Company a written designation of a beneficiary who is to receive any benefit under the Plan to which the Participant is entitled in the event of such Participant s death before he or she receives any or all of such benefit. Each designation will revoke all prior designations by the same Participant, shall be in a form prescribed by the Company, and will be effective only when filed by the Participant in writing with the Company during the Participant s lifetime. If a married Participant designates a beneficiary other than the Participant s spouse, the effectiveness of such designation may be subject to the consent of the Participant s spouse. If a Participant dies without an effective designation of a beneficiary who is living at the time of the Participant s death, the Company will pay any remaining unpaid benefits to the Participant s legal representative.

17.8 **Unfunded Obligation**. Participants shall have the status of general unsecured creditors of the Company. Any amounts payable to Participants pursuant to the Plan shall be unfunded and unsecured obligations for all purposes, including, without limitation, Title I of the Employee Retirement Income Security Act of 1974. No Participating Company shall be required to segregate any monies from its general funds, or to create any trusts, or establish any special accounts with respect to such obligations. The Company shall retain at all times beneficial ownership of any investments, including trust investments, which the Company may make to fulfill its payment obligations hereunder. Any investments or the creation or maintenance of any trust or any Participant account shall not create any vested or beneficial interest in any Participant or the Participant s creditors in any assets of any Participating Company. The Participants shall have no claim against any Participating Company for any changes in the value of any assets which may be invested or reinvested by the Company with respect to the Plan.

17.9 Section 409A of the Code. The provisions of the Plan are intended to comply with the requirements of Section 409A of the Code, and the Plan shall be so construed. Notwithstanding any other provision of the Plan to the contrary, the Committee may, in its sole and absolute discretion and without the consent of any Participant, amend the Plan to take effect retroactively or otherwise as it deems necessary or advisable for the purpose of conforming the Plan to the requirements of Section 409A of the Code.